quatrième série - tome 49

fascicule 6

novembre-décembre 2016

ANNALES SCIENTIFIQUES de L'ÉCOLE NORMALE SUPÉRIEURE

Jayadev S. ATHREYA & Alex ESKIN & Anton ZORICH Right-angled billiards and volumes of moduli spaces of quadratic differentials on \mathbb{CP}^1

SOCIÉTÉ MATHÉMATIQUE DE FRANCE

Annales Scientifiques de l'École Normale Supérieure

Publiées avec le concours du Centre National de la Recherche Scientifique

Responsable du comité de rédaction / Editor-in-chief

Antoine CHAMBERT-LOIR

Publication fondée en 1864 par Louis Pasteur	Co
Continuée de 1872 à 1882 par H. SAINTE-CLAIRE DEVILLE	N.
de 1883 à 1888 par H. DEBRAY	P. 1
de 1889 à 1900 par C. НЕВМІТЕ	E.
de 1901 à 1917 par G. DARBOUX	R.
de 1918 à 1941 par É. PICARD	A.
de 1942 à 1967 par P. MONTEL	

Comité de rédaction au 1 ^{er} janvier 2016	
N. Anantharaman	I. GALLAGHER
P. Bernard	B. KLEINER
E. Breuillard	E. Kowalski
R. Cerf	M. Mustață
A. Chambert-Loir	L. SALOFF-COSTE

Rédaction / Editor

Annales Scientifiques de l'École Normale Supérieure, 45, rue d'Ulm, 75230 Paris Cedex 05, France. Tél. : (33) 1 44 32 20 88. Fax : (33) 1 44 32 20 80. annales@ens.fr

Édition / Publication

Abonnements / Subscriptions

Société Mathématique de France Institut Henri Poincaré 11, rue Pierre et Marie Curie 75231 Paris Cedex 05 Tél. : (33) 01 44 27 67 99 Fax : (33) 01 40 46 90 96 Maison de la SMF Case 916 - Luminy 13288 Marseille Cedex 09 Fax : (33) 04 91 41 17 51 email : smf@smf.univ-mrs.fr

Tarifs

Europe : 519 €. Hors Europe : 548 €. Vente au numéro : 77 €.

© 2016 Société Mathématique de France, Paris

En application de la loi du l^{er} juillet 1992, il est interdit de reproduire, même partiellement, la présente publication sans l'autorisation de l'éditeur ou du Centre français d'exploitation du droit de copie (20, rue des Grands-Augustins, 75006 Paris). *All rights reserved. No part of this publication may be translated, reproduced, stored in a retrieval system or transmitted in any form or by any other means, electronic, mechanical, photocopying, recording or otherwise, without prior permission of the publisher.*

ISSN 0012-9593

Directeur de la publication : Stéphane Seuret Périodicité : 6 nºs / an

RIGHT-ANGLED BILLIARDS AND VOLUMES OF MODULI SPACES OF QUADRATIC DIFFERENTIALS ON CP¹

BY JAYADEV S. ATHREYA, ALEX ESKIN AND ANTON ZORICH with an appendix by Jon CHAIKA

ABSTRACT. – We use the relation between the volumes of the strata of meromorphic quadratic differentials with at most simple poles on \mathbb{CP}^1 and counting functions of the number of (bands of) simple closed geodesics in associated flat metrics with singularities to prove a very explicit formula for the volume of each such stratum conjectured by M. Kontsevich a decade ago.

Applying ergodic techniques to the Teichmüller geodesic flow we obtain quadratic asymptotics for the number of (bands of) closed trajectories and for the number of generalized diagonals in almost all right-angled billiards.

RÉSUMÉ. – Nous utilisons le lien entre les volumes des strates de différentielles méromorphes quadratiques avec des pôles simples sur \mathbb{CP}^1 et les fonctions de comptage du nombre de (cylindres de) géodésiques fermées simples pour la métrique plate associée afin de démontrer une formule très explicite pour le volume des strates, conjecturée par M. Kontsevich il y a une décennie.

En appliquant des techniques ergodiques au flot géodésique de Teichmüller nous obtenons une asymptotique quadratique pour le nombre de (bandes de) trajectoires fermées et le nombre de diagonales généralisées dans presque tout billard à angles « droits ».

1. Introduction

Motivated by the study of computing asymptotics for the number of generalized diagonals and for the number of closed billiard trajectories in right-angled polygons, we were naturally led to questions on Masur-Veech volumes of strata of moduli spaces of quadratic differentials on \mathbb{CP}^1 . Our main result, explicitly computing these volumes, resolves a conjecture of M. Kontsevich.

J.S.A. is partially supported by NSF grants DMS 1069153, NSF grants DMS 1107452, 1107263, 1107367 RNMS: GEometric structures And Representation varieties (the GEAR Network), and NSF CAREER grant DMS 1351853. A.E. is partially supported by NSF grants DMS 0244542 and DMS 0604251. A.Z. is partially supported by IUF and by ANR "GeoDyM"..

^{0012-9593/06/© 2016} Société Mathématique de France. Tous droits réservés ANNALES SCIENTIFIQUES DE L'ÉCOLE NORMALE SUPÉRIEURE

1.1. Volumes of moduli spaces of quadratic differentials

THEOREM 1.1 (Kontsevich Conjecture). – The volume of any stratum $Q_1(d_1, \ldots, d_k)$ of meromorphic quadratic differentials with at most simple poles on \mathbb{CP}^1 (i.e., $d_i \in \{-1; 0\} \cup \mathbb{N}$ for $i = 1, \ldots, k$, and $\sum_{i=1}^k d_i = -4$) is equal to

(1.1)
$$\operatorname{Vol} \mathcal{Q}_1(d_1, \dots, d_k) = 2\pi^2 \cdot \prod_{i=1}^k v(d_i)$$

(where all the zeroes and poles are "named").

Here, the function v is defined on integers n greater than or equal to -1 by

(1.2)
$$v(n) := \frac{n!!}{(n+1)!!} \cdot \pi^n \cdot \begin{cases} \pi & \text{when } n \text{ is odd} \\ 2 & \text{when } n \text{ is even} \end{cases}$$

for n = -1, 0, 1, 2, 3..., and the double factorial $n!! = n \cdot (n-2) \cdots$ is the product of all even (respectively odd) positive integers smaller than or equal to n. By convention we set

$$(-1)!! = 0!! = 1,$$

which implies that

$$v(-1) = 1$$
 and $v(0) = 2$.

This formula for the volume (up to some normalization factor) was conjectured by M. Kontsevich about ten years ago. It is much simpler than the formula for the volumes of the strata of Abelian differentials found by A. Eskin and A. Okounkov [25].

When this paper was written, there was not a single stratum of quadratic differentials for which the explicit volume was known, though an algorithm of computation was presented in [26]. In addition to this work, there is some very recent progress in evaluation of volumes of low-dimensional strata in genera different from 0. Rigorous formal methods used in [29] (in particular, implementation of the algorithm [26]) are confirmed by independent numerical experiments [12]. However, any known approach involves significant computer-assisted computations, and is limited to volumes of strata of sufficiently small dimension, while Theorem 1.1 provides a simple formula for *all* strata in genus 0.

Returning to our original motivation, we obtain as an important application of Theorem 1.1 asymptotics for the number of closed trajectories and for the number of generalized diagonals in right-angled polygons (see §1.3 below). This choice is particularly natural in the context of this paper since we have to solve an analogous problem for quadratic differentials and to compute the corresponding Siegel-Veech constants $c_{\mathcal{C}}$ for the strata of quadratic differentials in genus 0 anyway: it makes part of the proof of Theorem 1.1. This theorem also immediately provides asymptotics for certain Hurwitz numbers, see §1.2. Another example of applications is discussed in [14] where the values of volumes and the related Siegel-Veech constants are used to compute Lyapunov exponents of the Hodge bundle over hyperelliptic loci in the strata of quadratic differentials and to compute the diffusion rate for interesting families of generalized wind-tree billiards [13].

Strategy of the proof

We start by solving the counting problems for quadratic differentials. The Siegel-Veech constant c_{area} responsible for the exact quadratic asymptotics of the weighted number of bands of regular closed geodesics on almost any flat sphere in a given stratum $Q(d_1, \ldots, d_n)$ of meromorphic quadratic differentials with at most simple poles on \mathbb{CP}^1 was recently computed in [16],

(1.3)
$$c_{\text{area}}(\mathcal{Q}(d_1,\ldots,d_n)) = -\frac{1}{8\pi^2} \sum_{j=1}^n \frac{d_j(d_j+4)}{d_j+2}$$

Developing techniques elaborated in [21] for the strata of Abelian differentials and using the further results from [8] and [39] on the *principal boundary* of the strata of quadratic differentials we express the Siegel-Veech constant c_{area} in genus 0 in terms of the ratio of the volumes of appropriate strata,

(1.4)
$$c_{\text{area}}(\mathcal{Q}(d_1,\ldots,d_n)) = \frac{\text{explicit polynomial in volumes of simpler strata}}{\text{Vol}(\mathcal{Q}(d_1,\ldots,d_n))}$$

In this way we obtain a series of identities on the volumes of the strata of meromorphic quadratic differentials with at most simple poles in genus zero. The resulting identities recursively determine the volumes of all strata. The proof of Theorem 1.1, given in §5, consists in verifying that the expression (1.1) for the volume satisfies the combinatorial identities implied by (1.3) and (1.4). Part of this verification is performed in Appendix A.

REMARK 1.2 (Normalization conventions). – Note that the convention that all zeroes and poles are "named" affects the normalization: we compute the volumes of the corresponding covers over strata with "anonymous" singularities. For example, the stratum $Q(1, -1^5)$ of quadratic differentials with "anonymous" zeroes and poles is isomorphic to the stratum $\mathcal{H}(2)$ of holomorphic Abelian differentials; by convention the volume elements are chosen to be invariant under this isomorphism. However, by (1.1) we have

$$\operatorname{Vol} \mathcal{Q}_1(1, -1^5) = 2\pi^2 \cdot v(1) \cdot (v(-1))^5 = 2\pi^2 \cdot \frac{\pi^2}{2} \cdot 1^5 = 5! \cdot \frac{\pi^4}{120} = 5! \cdot \operatorname{Vol} \mathcal{H}_1(2),$$

which corresponds to 5! ways to give names to five simple poles.

Similarly,

$$\operatorname{Vol} \mathcal{Q}_1(2, -1^6) = 2\pi^2 \cdot v(2) \cdot (v(-1))^6 = 2\pi^2 \cdot \frac{4\pi^2}{3} \cdot 1^6 = \frac{6!}{2!} \cdot \frac{\pi^4}{135} = \frac{6!}{2!} \cdot \operatorname{Vol} \mathcal{H}_1(1, 1).$$

This time there is an extra factor $\frac{1}{2!}$ responsible for *forgetting the names* of the two zeroes of $\mathcal{H}(1,1)$.

1.2. Counting pillowcase covers

One of the ways to compute the volumes of the strata of Abelian or quadratic differentials (actually, the only one before the current paper) is to count *square-tiled surfaces* or *pillowcase covers*, see [25], [26], [27], [51]. In the current paper we proceed in the other direction: we first compute volumes of the strata by an alternative method, and then, as a corollary of volume computation, we get an explicit expression for the leading term of the function counting associated pillowcase covers, when the degree of the cover tends to infinity.

Actually, we consider two different types of branched covers over the standard "pillow". Covers of the first type ("pillowcase covers" as in [26]) are branched not only over the four corners of the pillow: all branching points of odd degree of a *pillowcase cover* in the sense of Eskin and Okounkov project to distinct points of \mathbb{CP}^1 different from the four corners. Namely, following [26] we define a *pillowcase cover* of degree 4d as a ramified cover

(1.5)
$$\pi: \mathscr{P} \to \mathscr{P}$$

over the pillowcase orbifold $\mathscr{P} = (\mathbb{C}/(\mathbb{Z} \oplus i\mathbb{Z}))/\pm$ (as in Figure 1) with ramification data given as follows. Let η be a partition and ν a partition of an even number into *odd* parts. Viewed as a map to the sphere, π has profile $(\nu, 2^{2d-|\nu|/2})$ over $0 \in \mathscr{P}$ and profile (2^{2d}) over the other three corners of \mathscr{P} . Additionally, π has profile $(\eta_i, 1^{4d-\eta_i})$ over $\ell(\eta)$ given points of \mathscr{P} and unramified elsewhere, where $\ell(\eta)$ is the number of parts in η . This ramification data determines the genus g of $\hat{\mathscr{P}}$ by

$$2 - 2g = \chi(\mathcal{P}) = \ell(\eta) + \ell(\nu) - |\eta| - |\nu|/2.$$

We consider only those ramification data for which $g = g(\hat{\mathscr{P}})$ in the above formula is equal to zero,

(1.6)
$$\ell(\eta) + \ell(\nu) - |\eta| - |\nu|/2 = 2.$$

Denote by $\operatorname{Cov}_{4d}^0(\eta,\nu)$ the number of inequivalent degree 4d connected covers $\pi: \hat{\mathscr{P}} \to \mathscr{P}$



FIGURE 1. Pillowcase orbifold.

with ramification data (η, ν) .

Denote by $\mathcal{Q}(\eta, \nu)$ the moduli space of quadratic differentials with singularity data $\{\nu_i - 2\}$ and $\{2\eta_i - 2\}$. Condition (1.6) guarantees that $\mathcal{Q}(\eta, \nu)$ is nonempty, and corresponds to genus zero.

Covers of the second type are "square tiled" in the following sense.

Consider the same partitions η , ν as above and a ramified cover

$$\pi_{\boxplus}: \hat{\mathscr{P}} \to \mathscr{P}$$

of the same degree 4d over the pillowcase orbifold \mathscr{P} with ramification data given as follows: $\pi_{\mathbb{H}}$ has profile $(2\eta, \nu, 2^{2d-|\eta|-|\nu|/2})$ over $0 \in \mathscr{P}$ and profile (2^{2d}) over the other three corners of \mathscr{P} . The cover $\pi_{\mathbb{H}}$ is unramified elsewhere. Applying the Riemann-Hurwitz formula and using relation (1.6) we see that covers with such ramification profile again have genus zero. The corresponding flat surface belongs to the same stratum $\mathscr{Q}(\eta, \nu)$ as before. Denote by $\operatorname{Cov}_{4d}^{0,\mathbb{H}}(\eta, \nu)$ the number of inequivalent degree 4d connected covers $\pi_{\mathbb{H}} : \widehat{\mathscr{P}} \to \mathscr{P}$ with ramification data (η, ν) as above.

Theorem 1.1 and the Theorem 1.3 below provide very simple asymptotic formulae for the Hurwitz numbers $\operatorname{Cov}_{4d}^{0}(\eta,\nu)$ and $\operatorname{Cov}_{4d}^{0,\boxplus}(\eta,\nu)$.

THEOREM 1.3. – For any ramification data (η, ν) satisfying condition (1.6) the numbers $\operatorname{Cov}_{4d}^{0}(\eta, \nu)$ and $\operatorname{Cov}_{4d}^{0,\boxplus}(\eta, \nu)$ of pillowcase covers of type (η, ν) admit the following limits:

(1.7)
$$\lim_{N \to \infty} \frac{1}{N^{\ell(\eta) + \ell(\nu) - 2}} \sum_{d=1}^{N} \operatorname{Cov}_{4d}^{0}(\eta, \nu) = 2^{\ell(\eta)} \cdot \frac{\operatorname{Vol} Q_{1}(\eta, \nu)}{2(\ell(\eta) + \ell(\nu) - 2)},$$

(1.8)
$$\lim_{N \to \infty} \frac{1}{N^{\ell(\eta) + \ell(\nu) - 2}} \sum_{d=1}^{N} \operatorname{Cov}_{4d}^{0, \boxplus}(\eta, \nu) = \frac{\operatorname{Vol} \mathcal{Q}_1(\eta, \nu)}{2(\ell(\eta) + \ell(\nu) - 2)},$$

where Vol $Q_1(\eta, \nu)$ is given by Equation (1.1).

Theorem 1.3 is proved in §B.2. Note that the more natural direct geometric approach to the counting of pillowcase covers leads to rather involved combinatorial problems. We present this alternative geometric approach in a separate paper [2].

REMARK 1.4. – There are several different combinatorial approaches to computing volumes of strata, based on counting (pillowcase) covers.

For the strata of Abelian differentials, the problem is solved in [25]; see also [51] for a more direct but much less efficient approach. Many of these combinatorial approaches can be pushed to produce some complicated expressions for the volumes in Theorem 1.1. Currently, the most efficient approach to calculation of volumes of strata of quadratic differentials (independently of genus) is suggested in [26]. The exact values of volumes of all strata up to dimension 11 are presented in [29] based on the algorithm of [26]; this result is close to limits of current computational capacities of modern computers in manipulating huge tables of characters. For an approach based on Kontsevich' solution to the Witten conjecture [32] see [2]; one more version developing ideas of Eskin and Okounkov is suggested in [45]; see also [12] for yet another approach. Paper [29] suggests a comparison of various approaches.

However, we were not able to get the simple expressions (1.1) using any of these methods. In fact, our proof of Theorem 1.1 is not purely combinatorial, but has analytic, geometrical and dynamical inputs (and is motivated by consideration of Lyapunov exponents). It thus remains a challenge to give a more direct proof of Theorem 1.1, in particular bypassing [16].

1.3. Counting trajectories of right-angled billiards

Currently it is not known whether there exists a single closed billiard trajectory in every obtuse triangle (see [46] for some progress in this direction and for further references). The situation with billiards in *rational* polygons (that is in polygons with angles which are rational multiples of π) is understood much better: trajectories of such billiards are related to geometry of certain compact flat surfaces with conical singularities, which are thoroughly studied starting with the landmark papers of H. Masur [33] and W. Veech [47]. In particular, it is known that a billiard in any rational polygon has infinitely many closed trajectories [34] (see also [31] for further applications), and, moreover, the number of trajectories of length at most L is bounded between c_1L^2 and c_2L^2 for some $0 < c_1 < c_2$ and for L large enough, see [35] and [36].

In the current paper we study families of right-angled billiards like the ones in Figures 2 and 3. Namely, we assume that the billiard table is a topological disk endowed with a flat metric, and that the boundary of the disk is piecewise geodesic such that the angle at

every corner of the boundary is an integer multiple of $\frac{\pi}{2}$. Note that by allowing integer multiples $k\pi/2$ with $k \ge 5$, we can obtain billiard tables which may not be embeddable in the plane (see Figure 2). In particular, we can consider helical right-angled billiards.



FIGURE 2. A Right-angled billiard table which is not embeddable into the plane.

We consider families of polygons sharing the same collection of interior corner angles $(\frac{\pi}{2}k_1, \frac{\pi}{2}k_2, \ldots, \frac{\pi}{2}k_n)$. Actually, it will be convenient to consider a slightly larger space $\mathscr{B}(k_1, \ldots, k_n)$ of "directional billiards" distinguishing a billiard table Π and the same table turned by angle ϕ . The measure in the space $\mathscr{B}(k_1, \ldots, k_n)$ is the product measure of Lebesgue measure arising from the side lengths and the angular measure $d\phi$.

We count the number of generalized diagonals of bounded length in such billiards (that is, the number of trajectories of bounded length which start in some fixed corner P_i and arrive to some fixed corner P_j , see Figure 3) and the number of closed billiard trajectories of bounded length. Note, that closed regular trajectories are never isolated in rational billiards: they always form bands of "parallel" closed trajectories of the same length, see Figure 3. Thus, when counting closed trajectories one actually counts the number of such bands. Sometimes, it is natural to count the bands with a weight which registers the thickness of the band, see e.g., Theorem 1.9 at the end of §1.3. By convention we always count *non-oriented* generalized diagonals and *non-oriented* closed billiard trajectories.

To give an idea of the general theorems stated in detail in §2 and developed in §4, we present the following representative results.

THEOREM 1.5. – For any right-angled billiard Π outside of a zero measure set in any family $\mathscr{B}(k_1, \ldots, k_n)$ the number $N_{ij}(\Pi, L)$ of generalized diagonals of length at most L joining a pair of fixed corners P_i, P_j with angles $\frac{\pi}{2}$ has the following quadratic asymptotics as $L \to \infty$:

(1.9)
$$N_{ij}(\Pi, L) \sim \frac{1}{2\pi} \cdot \frac{L^2}{\text{area of the billiard table}}$$

Theorem 1.5 is proved in §4.11, using the theorem proved by Jon Chaika in Appendix C.

The fact that this asymptotics does not depend at all on the billiard table is at the first glance counterintuitive. What is even more surprising is that it is universal: it is the same not only for almost all billiard tables inside each family, but it does not vary even from one family to another! In particular, though the shape of the two polygons of the same area in Figure 3 is quite different, the number of trajectories of length at most L joining the right-angle corner P_i to the right-angle corner P_j is approximately the same in both cases, and is

4° SÉRIE - TOME 49 - 2016 - Nº 6



FIGURE 3. A family $\mathcal{B}(k_1, \ldots, k_n)$ of right-angled polygons; a band of periodic trajectories on the left, and a generalized diagonal on the right.

approximately the same as the number of trajectories of length at most L joining two corners of the usual rectangular billiard of the same area when $L \gg 1$.

The situation becomes more complicated when we consider other types of corners of the billiard. Consider, for example, an L-shaped billiard table as on Figure 4. Let P_1, \ldots, P_5 be the right-angle corners of the L-shaped billiard, and let P_0 be the corner with the interior angle $\frac{3\pi}{2}$.



FIGURE 4. L-shaped billiard.

THEOREM 1.6. – For almost any L-shaped billiard Π the number $N_{i0}(\Pi, L)$ of generalized diagonals of length at most L joining a fixed corner P_i with angle $\frac{\pi}{2}$ and the corner P_0 with angle $\frac{3\pi}{2}$ has the following quadratic asymptotics as $L \to \infty$:

(1.10)
$$N_{i0}(\Pi, L) \sim \frac{2}{\pi} \cdot \frac{L^2}{\text{area of the billiard table}}$$

The proof of this theorem also relies in part on Theorem C.1 proved by Jon Chaika in Appendix C.

The naive intuition does not help: the angle $\frac{3\pi}{2}$ at the corner P_0 is *three* times larger than in the previous case, while the constant in the asymptotics for the number of generalized diagonals is *four* times larger than in the previous statement. Currently we have no idea how to obtain this factor 4 without using techniques of the Teichmüller geodesic flow, Lyapunov exponents of the Hodge bundle, and the computation of volumes of the moduli spaces of meromorphic quadratic differentials with at most simple poles on \mathbb{CP}^1 . Theorem 1.6 is proved in §4.9.

Using recently developed technology, one can prove weak asymptotic formulas similar to Theorem 1.5 and Theorem 1.6 for individual billiard tables. In particular, the following holds:

THEOREM 1.7. – Suppose Π is an L-shaped billiard table as in Figure 4. Let

$$a = \frac{|P_3 P_4|}{|P_1 P_2|}, \qquad b = \frac{|P_2 P_3|}{|P_4 P_5|}$$

Then,

(i) If a and b are both rational, or if there exists a non-square integer D > 0 such that $a, b \in \mathbb{Q}(\sqrt{D})$ and $a + \overline{b} = 1$ (where \overline{b} is the Galois conjugate of b), then there exist constants c_{ij} such that

(1.11)
$$N_{ij}(\Pi, L) \sim c_{ij} \frac{L^2}{\text{area of the billiard table}}$$

(ii) For any other L-shaped billiard table, we have the "weak asymptotic formulas"

$$N_{ij}(\Pi, L) \sim \frac{1}{2\pi} \cdot \frac{L^2}{\text{area of the billiard table}}$$

and

$$N_{i0}(\Pi, L) \sim \pi^2 \frac{2}{\pi} \cdot \frac{L^2}{\text{area of the billiard table}}.$$

The meaning of the "weak asymptotic" "~" is defined in §2.2.

In the case (i) the Siegel-Veech constants c_{ij} for rational values of parameters a, b can be computed by the formula due to E. Gutkin and C. Judge [30]. For $i, j \neq 0$ and $a, b \in \mathbb{Q}(\sqrt{D})$ the constants c_{ij} are computed by M. Bainbridge, see [6, Theorem 1.5 and §14].

Proof. – Theorem 1.7 is a compilation of several different results. In case (i), the polygon Π is a Veech polygon, which gives rise to a Teichmüller curve, see [9], [40]. The existence of an asymptotic formula such as (1.11) for such a situation was proved in the pioneering work of W. Veech [48].

Let

$$U = \begin{pmatrix} 1 & * \\ 0 & 1 \end{pmatrix} \subset \operatorname{SL}(2, \mathbb{R}), \qquad P = \begin{pmatrix} * & * \\ 0 & * \end{pmatrix} \subset \operatorname{SL}(2, \mathbb{R}).$$

The fact that weak asymptotic formulas such as those of part (ii) hold for any rational billiard table follows from [23, Theorem 2.12], which uses the general invariant measure classification theorem of [22] for the action of P on moduli space. However, to evaluate the constant for an arbitrary L-shaped table, one also has to appeal to the explicit classification of $SL(2, \mathbb{R})$ -invariant affine submanifolds in the moduli space of Abelian differentials in genus 2 due to C. McMullen, [41].

We note that asymptotic counting formulas for individual billiards are associated with invariant measure classification theorems on the action of subgroups of $SL(2, \mathbb{R})$ on (certain subsets of) the moduli space. In particular, when a measure classification theorem for the action of the subgroup U exists (e.g., in the case of a Teichmüller curve), one can get a strong asymptotic formula. Also, a measure classification theorem for the action of the subgroup P leads to a weak asymptotic formula.

For other examples when a classification of invariant measures for the action of U (and thus strong asymptotic formulas) are known see [20], [18], [10], [7]. All examples of individual billiard tables for which the (strong) quadratic asymptotics was known are, essentially, covered by several families of triangles depending on one integer parameter; by several sporadic triangles beyond these families; by a square with a specially located barrier; and by a family of L-shaped tables with or without a wall for special values of parameters of the L-shaped table.

In §2.2 for each family $\mathcal{B}_1(k_1, \ldots, k_n)$ of right-angled billiards we describe all geometric types of generalized diagonals and all closed billiard trajectories which can be found on a billiard Π outside of a zero measure set in $\mathcal{B}_1(k_1, \ldots, k_n)$. For such Π , and each such geometric type we prove (strong) quadratic asymptotics for the number of associated generalized diagonals (or of bands of closed billiard trajectories), and explicitly evaluate the constant in the quadratic asymptotics.

Theorem 1.8 below illustrates an application of the general Theorem 2.5 and of the general Theorems 4.3–4.8 to billiards more complicated than the L-shaped ones, see Figure 5.



FIGURE 5. A billiard table from the family $\mathscr{B}(4^m, 3^n, 1^{m+n+4})$.

By $\mathscr{B}(4^m, 3^n, 1^{m+n+4})$ we denote the family of right-angled billiards with *m* corners with angles 2π (endpoints of the walls); *n* corners with interior angles $3\pi/2$, and with the remaining n + m + 4 corners with interior angles $\pi/2$.

THEOREM 1.8. – Consider two distinct corners P_i, P_j of a billiard Π in any family $\mathscr{B}(4^m, 3^n, 1^{m+n+4})$. Assume that at least one of the interior angles $k_i \pi/2$ and $k_j \pi/2$ is different from $\pi/2$ (i.e., k_i, k_j are not simultaneously equal to 1).

For almost any Π , any generalized diagonal δ joining P_i to P_j and non parallel to a side of Π never bounds a band of closed trajectories. No other generalized diagonal in Π has a segment parallel to any segment of δ . For almost any Π , the number $N_{ij}(\Pi, L)$ of such generalized diagonals of length at most L has the following asymptotics as $L \to +\infty$:

 $N_{ij}(\Pi, L) \sim c_{ij} \cdot \frac{L^2}{\text{area of the billiard table}},$

where the constant c_{ij} depends only on the angles $k_i \pi/2$ and $k_j \pi/2$ at P_i and P_j correspondingly; its value is presented in the following table:

$$\begin{array}{c} \text{angle} \quad \frac{4\pi}{2} \quad \frac{3\pi}{2} \quad \frac{\pi}{2} \\ \hline \\ \frac{4\pi}{2} \quad \frac{9\pi}{10} \quad \frac{45\pi}{64} \quad \frac{9\pi}{32} \\ \hline \\ \frac{3\pi}{2} \quad \frac{45\pi}{64} \quad \frac{16}{3\pi} \quad \frac{2}{\pi} \\ \hline \\ \frac{\pi}{2} \quad \frac{9\pi}{32} \quad \frac{2}{\pi} \quad \frac{1}{2\pi} \end{array}$$

Note that the values of the constants do not depend neither on the numbers n or m of corners, nor on the particular shape of the billiard. The proof of this theorem also relies in part on Theorem C.1 proved by Jon Chaika in Appendix C.

We complete this section with an illustration of further counting problems where one can apply our techniques. Let $N_{area}(\Pi, L)$ denote the number of bands of closed periodic billiard trajectories of length at most L counted with a weight given by the normalized area of the band. More precisely, we count the area of overlapping domains of the band twice: the area of the band is naively measured as the area of the associated cylinder on the flat sphere, that is, the width of the band times the length of the closed trajectory, normalized by the area of the billiard table. Having measured the area of the band, we divide it by the area of the billiard table to get the weight of the band.

THEOREM 1.9. – For any billiard Π in any family $\mathscr{B}(k_1, \ldots, k_n)$ of right-angled billiards the weighted number $N_{\text{area}}(\Pi, L)$ of bands of closed billiard trajectories of length at most Lsatisfies the following weak asymptotics as $L \to \infty$:

$$N_{\text{area}}(\Pi, L) \quad ``\sim `` \frac{1}{16\pi} \sum_{j=1}^{n} \left(\frac{4}{k_j} - k_j\right) \cdot \frac{L^2}{\text{area of the billiard table}}$$

For almost any billiard Π in the same family, the asymptotics is, actually, exact:

(1.12)
$$N_{\text{area}}(\Pi, L) \sim \frac{1}{16\pi} \sum_{j=1}^{n} \left(\frac{4}{k_j} - k_j\right) \cdot \frac{L^2}{\text{area of the billiard table}}$$

The weak asymptotics for all billiards follows, as before, from [23, Theorem 2.12]. The strong asymptotics (1.12) is proved in §6.1, using Jon Chaika's Theorem C.1 which is proved in Appendix C. The constant in the corresponding counting function is directly related to the *Siegel-Veech area constant* for the corresponding stratum of meromorphic quadratic differentials on \mathbb{CP}^1 discussed in §1.1.

1.4. Right-angled billiard tables and quadratic differentials

Given a right-angled billiard Π in $\mathcal{B}(k_1, \ldots, k_n)$ we can glue a topological sphere from two superposed copies of Π identifying the boundaries of the two copies by isometries, see Figure 6. By construction the resulting topological sphere is endowed with a flat metric. Note that the metric is regular on the interior of the segments coming from the boundary of Π : one can unfold a neighborhood of any such point into a small regular flat domain.



FIGURE 6. Flat spheres glued from two copies of a right-angled billiard. The angle by which the billiard table is rotated with respect to the horizontal position encodes the "phase" of the corresponding quadratic differential. A general generalized diagonal in the polygon gives rise to two distinct saddle connections on the flat sphere.

However, the resulting flat metric has conical singularities with cone angles $\pi k_1, \ldots, \pi k_n$ at the points coming from the vertices of Π . By construction the linear holonomy of the flat metric with isolated singularities belongs to the group $\mathbb{Z}/2\mathbb{Z}$: the parallel transport along a short path encircling a conical point P_j brings a tangent vector \vec{v} either to itself or to $-\vec{v}$ depending on the parity of k_j .

It is known that a flat metric with isolated conical singularities and with holonomy in $\mathbb{Z}/2\mathbb{Z}$ on a closed surface defines a complex structure and a meromorphic quadratic differential qwith respect to this complex structure defined up to multiplication by a scalar $e^{i\phi}$. Choosing a line direction $\pm \vec{v}$ at some point of the resulting flat sphere as a "horizontal" direction we fix the scalar $e^{i\phi}$. In an appropriate flat local coordinate z outside of the conical points the resulting quadratic differential has the form $(dz)^2$. A conical singularity with a cone angle $k_i\pi$ corresponds to a zero of the quadratic differential of degree $k_i - 2$, where a "zero of degree -1" is a simple pole.

Actually, the two structures are synonymous: a meromorphic quadratic differential q with at most simple poles on a Riemann surface defines a canonical flat metric with isolated conical singularities, with linear monodromy in $\mathbb{Z}/2\mathbb{Z}$ and with a distinguished foliation by straight lines in the flat metric (see the original papers [33] and [47] or surveys [38] and [52]).

By construction closed billiard trajectories in Π are in canonical one-to-two correspondence with closed regular geodesics on the associated flat sphere, and generalized diagonals on Π are in the natural one-to-two correspondence with the *saddle connections* on the associated flat sphere, see Figure 6. Thus, the two counting problems are closely related.

It is known by work of Veech [49] and of Eskin-Masur [19] that almost all flat spheres in a given stratum $Q(d_1, \ldots, d_n)$ satisfy a quadratic asymptotic formula for the number of saddle connections. However, we cannot immediately translate this result to right-angled billiards. An elementary count shows that the space $\mathcal{B}(k_1, \ldots, k_n)$ has *real* dimension n-2, while the associated stratum $Q(k_1 - 2, \ldots, k_n - 2)$ has *complex* dimension n-2. Thus, flat spheres constructed from right-angled billiards form a subset of measure zero, and "almost all" results for the strata are not applicable to families of billiards. This is the common difficulty of translating results valid for flat surfaces to billiards.

In our specific case we are lucky enough to get a subspace of flat spheres "of billiard origin" which is transversal to the unstable foliation of the Teichmüller flow (see §3). This allows us to apply certain techniques of hyperbolic dynamics to obtain some ergodic results

in slightly weaker form. As a corollary we obtain the desired information on quadratic asymptotics in the counting problems for almost all billiards. The corresponding ergodic technique is presented in §6. A key tool we use is Theorem C.1 proved by Jon Chaika in Appendix C.

1.5. Reader's guide

The paper (like Caesar's Gaul) is composed of three parts. The reader interested only in the billiards may read only §2 (and optionally §3 and §6). The ergodic theorem we use in §6 is due to Jon Chaika, and is proved in Appendix C.

The part where we compute the volume of any stratum $Q_1(d_1, \ldots, d_n)$ of meromorphic quadratic differentials with at most simple poles on \mathbb{CP}^1 and where we compute the *Siegel-Veech constants* for these strata is independent from the rest of the paper. It is presented in §2.1, §§3.1–3.2 and in §§4–5 (with one verification in Appendix A).

Finally, Appendix B devoted to pillowcase covers is completely independent of the rest of the paper.

1.6. Historical remarks

The formula for the volume of the strata of quadratic differentials was guessed by M. Kontsevich more than a decade ago. At this time Formula (1.3) related to Lyapunov exponents was known experimentally. The Siegel-Veech constant (1.4) has especially simple form for the strata $Q(d, -1^{d+4})$ of quadratic differentials with a single zero and only simple poles on \mathbb{CP}^1 . Comparing (1.3) and a version of (1.4) M. Kontsevich obtained a conjectural formula for Vol $Q_1(d, -1^{d+4})$. Motivated by the simplicity of the resulting expression as a function of d he stated a guess that Vol $Q_1(d_1, \ldots, d_k)$ for any stratum in genus 0 might be expressed as a product of the corresponding expressions for all d_i .

Acknowledgments

The authors are grateful to M. Kontsevich for the conjecture on the volumes and for collaboration in the work on Lyapunov exponents essential for the current paper.

Part of this paper strongly relies on techniques developed in collaboration with H. Masur. We are grateful to him for his very important contribution.

We thank C. Boissy for the list of configurations of homologous saddle connections in genus 0, elaborated specifically for our needs, and for his pictures of configurations.

We thank P. Paule for the kind permission to use the "MultiSum" package, which was helpful in developing certain intuition with the combinatorial identity.

We thank E. Goujard who carefully read the first version of this paper, provided us with a list of typos, and indicated to us a missing factor in Formula (1.7) and necessity to prove Lemma B.3 as explained in Remark B.4.

We are grateful to G. Forni and to P. Hubert whose suggestions helped to improve the presentation and clarity of the arguments.

The authors are happy to thank IHÉS, IMJ, IUF, MPIM, MSRI, and the Universities of Chicago, of Illinois at Urbana-Champaign, of Rennes 1, and of Paris 7 for hospitality during the preparation of this paper.

1323

2. Configurations and Counting Theorems

2.1. Types of saddle connections and generalized diagonals

We distinguish the following four ways of getting generalized diagonals in a right-angled billiard. They correspond to four types of configurations of saddle connections on a flat sphere defined by a meromorphic quadratic differential with simple poles, see [21] and [39] for general information on *configurations of saddle connections* and [8] for specific case of \mathbb{CP}^1 .

I. Saddle connection joining distinct singularities. – In this situation (see Figure 7) we have a generalized diagonal joining a corner P_i with the inner angle $k_i \frac{\pi}{2}$, where $k_i \ge 3$, to a distinct corner P_j .



FIGURE 7. Type I. On the left: a generalized diagonal joining two distinct corners of the billiard, where at least one of the two corners has inner angle at least $\frac{3\pi}{2}$. It does not bound a band of closed trajectories. On the right: a saddle connection on \mathbb{CP}^1 joining a zero to a distinct zero (or to a pole).

The induced flat metric on \mathbb{CP}^1 has an associated saddle connection of the same length joining the zero P_i to the distinct zero (or simple pole) P_j .

II. Saddle connection joining a zero to itself. – This situation (see Figure 8) can happen only when we have a corner P_i with a corner angle $k_i \frac{\pi}{2}$ with $k_i \ge 4$. In this case we can have a generalized diagonal joining the corner P_i to itself such that it does not bound a band of closed regular trajectories.



FIGURE 8. Type II. On the left: a generalized diagonal returning to the same corner. For this type, it does not bound closed trajectories. On the right: the corresponding saddle connection joining a zero (of order at least 2) to itself.

For the induced flat metric on \mathbb{CP}^1 we get a corresponding saddle connection of the same length joining the zero P_i to itself such that the total angle $k_i\pi$ at the singularity P_i is split by the separatrix loop into two sectors having the angles strictly greater than π .

Proposition 2.1 below implies that for almost all flat surfaces in any stratum of meromorphic quadratic differentials with at most simple poles on \mathbb{CP}^1 each boundary component of *any* maximal cylinder filled with periodic geodesics contains exactly one saddle connection.

This implies that for such almost all flat surfaces the saddle connections of types I and II as above never belong to a boundary of such maximal flat cylinder. In the remaining two cases we consider saddle connections which do bound a cylinder of closed regular flat geodesics.

III. A "pocket". – In this situation (see Figure 9) we have a band of periodic trajectories. The boundary of the band is composed of two generalized diagonals. The first generalized diagonal joins a pair of corners P_i , P_j with inner angles $\frac{\pi}{2}$. The length of this saddle connection is twice shorter than the length of periodic billiard trajectory in the band. The second generalized diagonal joins a corner P_i with inner angle $k_i \frac{\pi}{2}$ with $k_i \ge 3$ to itself. The length of this saddle connection is the same as the length of periodic billiard trajectory in the band.



FIGURE 9. Type III. On the left: a band of closed trajectories bounded by two generalized diagonals. One of generalized diagonals joins two distinct corners with angles $\frac{\pi}{2}$; the other returns to the same corner. On the right: the corresponding "pocket" configuration with a cylinder bounded on one side by a saddle connection joining two simple poles, and by a saddle connection joining a zero to itself on the other side.

For the associated flat metric on \mathbb{CP}^1 we get a cylinder filled with closed regular trajectories. One of the boundary components of the cylinder degenerates to a saddle connection joining two simple poles P_i , P_j . Clearly, this saddle connection is twice shorter than the length of the periodic trajectories. The other boundary component is a saddle connection joining the zero P_l to itself. The total angle $k_l\pi$ at the singularity P_l is split by the separatrix loop into two sectors, such that the sector adjacent to the cylinder has angle π . The length of this saddle connection is the same as the length of the periodic trajectories in the cylinder.

IV. A "dumbbell". – In this last situation (see Figure 10) we again have a band of periodic trajectories. The boundary of the band is again composed of two generalized diagonals, but this time the first generalized diagonal joins the corner P_i with inner angle $k_i \frac{\pi}{2}$ to itself, and the second generalized diagonal joins the distinct corner P_j with inner angle $k_j \frac{\pi}{2}$ to itself. Both k_i, k_j are greater than or equal to 3. The length of each of these two generalized diagonals is the same as the length of every periodic billiard trajectory in the band.

For the associated flat metric on \mathbb{CP}^1 we get a cylinder filled with closed regular trajectories. On each of the boundary components of the cylinder we have a saddle connection joining the zero P_i (correspondingly P_j) to itself. The length of each of the two saddle connections is the same as the length of the periodic trajectories in the cylinder.

4° SÉRIE - TOME 49 - 2016 - Nº 6



FIGURE 10. Type IV. On the left: a band of periodic trajectories, such that each of the two bounding generalized diagonals returns to the same corner. On the right: a "dumbbell" composed of two flat spheres joined by a cylinder. Each boundary component of the cylinder is a saddle connection joining a zero to itself.

The following two propositions explain why we distinguish these four particular types of configurations (see more details in §3.2 which discusses a homological interpretation of these statements).

PROPOSITION 2.1. – Almost any flat surface S in any stratum $Q_1(d_1, \ldots, d_n)$ different from the pillowcase stratum $Q_1(-1^4)$ does not have a single pair of parallel saddle connections different from the pairs involved in configurations of types I, II, III, IV.

Proposition 2.1 is proved in §3.2. An analogous statement can be formulated for rightangled billiards.

PROPOSITION 2.2. – For almost any right-angled billiard in any family $\mathcal{B}(k_1, \ldots, k_n)$ the following property holds. Consider a pair of trajectories, where each trajectory is either a closed regular trajectory or a generalized diagonal. Suppose that these trajectories are not parallel to any side of the polygon. If some segment of the first trajectory is parallel to some segment of the second trajectory, then both trajectories make part of one of configurations I–IV described in 2.1.

Proposition 2.2 mimics Proposition 7.4 in [21]; it is proved in §3.3.

Configurations of saddle connections. – In addition to the type I–IV of a saddle connection, we may specify some extra combinatorial information, for example the indices ("names") of all singularities involved. For saddle connections of type IV, where a cylinder is joining two spheres, we specify not only the zeroes P_i and P_j at the boundary components of the cylinder, but we also specify the subcollections $P_{i_1}, \ldots, P_{i_{k_1}}$ and $P_{j_1}, \ldots, P_{j_{k_2}}$ of numbered zeroes an poles which get to the first and to the second sphere correspondingly. We call this information the *configuration* of a saddle connections involved as in types III and IV). By convention, the configuration of saddle connections includes its type. See also §3.2 for a homological interpretation of a configuration of saddle connections.

Configuration of a generalized diagonal. – By the configuration of the generalized diagonal we mean the configuration of the associated saddle connections in \mathbb{CP}^1 described in §1.4.

2.2. Counting Theorems

By the notation

$$N(L) \sim cL^2$$

we mean as customary,

$$\lim_{L \to \infty} \frac{N(L)}{L^2} = c.$$

We will also consider "weak asymptotic formulas"

$$N(L)$$
 "~" cL^2

which means

$$\lim_{L \to \infty} \frac{1}{L} \int_0^L N(e^t) e^{-2t} \, dt = c$$

The following theorem (which is a special case of results of [49] and [19]) establishes a strong asymptotic formula for almost all flat surfaces in a stratum. By convention we always count *non-oriented* saddle connections and *non-oriented* closed flat geodesics.

THEOREM 2.3. – For almost any flat surface S in any stratum $Q(d_1, \ldots, d_n)$ of meromorphic quadratic differentials with at most simple poles on \mathbb{CP}^1 the number $N_{\mathcal{C}}(S, L)$ of occurrences of saddle connections of length at most L and of fixed configuration \mathcal{C} , has quadratic asymptotics in L:

$$N_{\mathcal{C}}(S,L) \sim c_{\mathcal{C}} \cdot \frac{\pi L^2}{\text{area of } S}$$

The constants $c_{\mathcal{C}}$ are called Siegel-Veech constants. They depend only on the configuration \mathcal{C} and on d_1, \ldots, d_n . Their values are given in §4.

Theorem 2.3 is proved in §4.5. Note that Theorem 2.3 has no relation to billiards, it concerns only flat metrics on \mathbb{CP}^1 induced by meromorphic quadratic differentials with simple poles. In §1.4 we described how a right-angled billiard table Π canonically determines a meromorphic quadratic differential on \mathbb{CP}^1 . However, since the image of the resulting map $\mathcal{B}(k_1, \ldots, k_n) \rightarrow \mathcal{Q}(k_1 - 2, \ldots, k_n - 2)$ has measure 0 in $\mathcal{Q}(k_1 - 2, \ldots, k_n - 2)$, results such as Theorem 2.3 do not immediately imply anything about right-angled billiards. Nevertheless, we have the following:

THEOREM 2.4. – For almost any billiard table Π in any family $\mathscr{B}(k_1, \ldots, k_n)$ of rightangled billiards the number $N_{\mathscr{C}}(\Pi, L)$ of occurrences of generalized diagonals of configuration \mathscr{C} and of length at most L has quadratic asymptotics in L:

(2.1)
$$N_{\mathscr{C}}(\Pi, L) \sim \frac{c_{\mathscr{C}}}{4} \cdot \frac{\pi L^2}{\text{area of the billiard table }\Pi}$$

where the constants $c_{\mathcal{C}}$ are the corresponding Siegel-Veech constants $c_{\mathcal{C}}$ for the stratum $Q(k_1 - 2, ..., k_n - 2)$ in Theorem 2.3.

4° SÉRIE - TOME 49 - 2016 - Nº 6

Theorem 2.4 is proved in §6.1, using Jon Chaika's Theorem C.1 which is proved in Appendix C.

The factor of $\frac{1}{4}$ in (2.1) is explained as follows. Note that any generalized diagonal in the billiard table Π which is not parallel to one of the sides of Π canonically determines two symmetric saddle connections of the same type on the flat surface *S* glued from the two copies of Π , where the symmetry is the antiholomorphic involution, see Figure 6. Hence,

$$N_{\mathscr{C}}(\Pi,L)=rac{1}{2}N_{\mathscr{C}}(S,L).$$

Note also that by construction the area of S is twice the area of the billiard table Π .



FIGURE 11. A helical billiard corresponds to the stratum $Q(d, -1^{d+4})$.

Note that our billiard table does not need to be necessarily embeddable into the plane, say, we can consider a helical right-angled billiard as in Figure 11 (see also [11] for more general billiards of this type).

More precisely, by a *right-angled billiard table* we call a topological disc endowed with a flat metric having the following properties. The flat metric is allowed to have isolated conetype singularities in the interior of the disc with cone angles of the form $l_i\pi$, with $l_i \in \mathbb{N}$. The boundary of the disc is piecewise-geodesic in the flat metric, and the angles between the geodesic segments have the form $k_j\pi/2$, with $k_j \in \mathbb{N}$.

In fact, some version of Theorem 2.4 holds for individual billiards:

THEOREM 2.5. – Suppose Π is a billiard table from the family of right-angled billiards $\mathcal{B}(k_1, \ldots, k_n)$. Furthermore, suppose Π is such that the flat surface S glued from two copies of Π does not belong to any proper $\operatorname{GL}(2, \mathbb{R})$ -invariant affine submanifold of the stratum $Q(d_1, \ldots, d_k)$. Then, for any choice I–IV of configuration \mathcal{C} , the weak asymptotic formula

$$N_{\mathcal{C}}(\Pi, L) \sim \frac{c_{\mathcal{C}}}{4} \cdot \frac{\pi L^2}{\text{area of the billiard table }\Pi}$$

holds, where $c_{\mathcal{C}}$ is the Siegel-Veech constant corresponding to the configuration \mathcal{C} in the stratum $Q(k_1 - 2, ..., k_n - 2)$ (as in Theorem 2.3).

Proof. – The statement is an immediate corollary of [23, Theorem 2.12]. \Box

We note that a complete proof of [23, Theorem 2.12] involves the measure classification theorem of [22] and is well over 200 pages long, and yields *weak* asymptotic formulas. The proof of Theorem 2.4 is much shorter, and uses special features of right-angled billiards, namely Proposition 3.3. However, Theorem 2.4 is an almost everywhere statement, and does not imply any type of asymptotic formula for an individual billiard table.

We also note that for most other families of billiards, almost-everywhere statements like Theorem 2.4 are not available (since the analogue of Proposition 3.3 fails.)

3. Billiards in right-angled polygons and quadratic differentials

In §3.1 we describe the cohomological coordinates in a stratum of quadratic differentials. We proceed in §3.2 with a reminder of the notions of \hat{h} omologous saddle connections and a configuration of \hat{h} omologous saddle connections.

In §3.3 we analyze the canonical embedding of the space of (directional) right-angled billiards $\mathscr{B}(k_1, \ldots, k_n)$ into the corresponding space $\mathscr{Q}(k_1 - 2, \ldots, k_n - 2)$ of meromorphic quadratic differentials on \mathbb{CP}^1 . Namely, we prove in Proposition 3.3 that its image projects surjectively onto the unstable foliation of the Teichmüller geodesic flow, which allows us to apply certain ergodic techniques of hyperbolic dynamics not only to flat surfaces from $\mathscr{Q}(k_1 - 2, \ldots, k_n - 2)$ but to billiards from $\mathscr{B}(k_1, \ldots, k_n)$.

We complete §3 with a proof of Proposition 2.2.

3.1. Coordinates in a stratum of quadratic differentials

Consider a meromorphic quadratic differential ψ having zeroes of arbitrary multiplicities but only simple poles on \mathbb{CP}^1 . Let P_1, \ldots, P_n be its singular points (zeros and simple poles). Consider the minimal branched double covering $p : \hat{S} \to \mathbb{CP}^1$ such that the induced quadratic differential $p^*\psi$ on the hyperelliptic surface \hat{S} is a square of an Abelian differential $p^*\psi = \omega^2$.

The zeros $\hat{P}_1, \ldots, \hat{P}_N$ of the resulting Abelian differential ω correspond to the zeros of ψ in the following way: every zero $P \in \mathbb{CP}^1$ of ψ of odd order is a ramification point of the covering, so it produces a single zero $\hat{P} \in \hat{S}$ of ω ; every zero $P \in \mathbb{CP}^1$ of ψ of even order is a regular point of the covering, so it produces two zeros $\hat{P}^+, \hat{P}^- \in \hat{S}$ of ω . Every simple pole of ψ defines a branching point of the covering; this point is a regular point of ω .

Consider the subspace $H_1^-(\hat{S}, \{\hat{P}_1, \dots, \hat{P}_N\}; \mathbb{Z})$ of the relative homology of the cover with respect to the collection of zeroes $\{\hat{P}_1, \dots, \hat{P}_N\}$ of ω which is antiinvariant with respect to the induced action of the hyperelliptic involution. We are going to construct a basis in this subspace (in complete analogy with a usual basis of absolute cycles for a hyperelliptic surface).

We can always enumerate the singular points P_1, \ldots, P_n of ψ in such a way that the first point P_1 and the last point P_n correspond to simple poles.

Chose now a simple oriented broken line P_1, \ldots, P_{n-1} on \mathbb{CP}^1 joining consecutively all the singular points of ψ except the last one. For every arc $[P_i, P_{i+1}]$ of this broken line, $i = 1, \ldots, n - 2$, the difference of their two preimages defines a relative cycle in $H_1^-(\hat{S}, \{\hat{P}_1, \ldots, \hat{P}_N\}; \mathbb{Z})$. By construction such a cycle is antiinvariant with respect to the hyperelliptic involution.

LEMMA 3.1. – The resulting collection of integer relative cycles antiinvariant with respect to the involution forms a basis of cycles in $H_1^-(\hat{S}, \{\hat{P}_1, \ldots, \hat{P}_N\}; \mathbb{Z})$.

4° SÉRIE - TOME 49 - 2016 - Nº 6

1329



FIGURE 12. Basis of cycles in $H_1^-(\hat{S}, \{\hat{P}_1, \dots, \hat{P}_N\}; \mathbb{Z})$. Note that the cycle corresponding to the very last slit is omitted.

Proof. – Let $\Sigma = {\hat{P}_1, \ldots, \hat{P}_N}$. By construction the set Σ is invariant under the involution (as a set, not pointwise), so we can consider the exact sequence of the pair $\Sigma \subset S$ separately for invariant and antiinvariant homology subgroups. Consider the following fragment of such exact sequence:

$$H_1^-(S;\mathbb{Z}) \to H_1^-(\tilde{S},\Sigma;\mathbb{Z}) \to H_0^-(\Sigma;\mathbb{Z}).$$

First note that the image of $H_1^-(S;\mathbb{Z})$ is spanned by our collection of relative cycles. To verify this statement, it is sufficient to consider sums of relative cycles corresponding to consecutive segments $[P_{i_k}, P_{i_k+1}], [P_{i_k+1}, P_{i_k+2}], \ldots, [P_{i_k+\ell_k}, P_{i_{k+1}}]$, where P_{i_k} and $P_{i_{k+1}}$ located at the extremities of the corresponding broken line are poles or zeroes of odd orders while all $P_{i_k+1}, P_{i_k+2}, \ldots, P_{i_k+\ell_k}$ located in between are zeroes of even orders. In this way we get a standard basis of absolute cycles of a hyperelliptic surface.

Now note that the homomorphism $H_1^-(\hat{S}, \Sigma; \mathbb{Z}) \to H_0^-(\Sigma; \mathbb{Z})$ is surjective. Note also that 0-cycles of the form $[\hat{P}_i^+] - [\hat{P}_i^-]$ generate the group $H_0^-(\Sigma; \mathbb{Z})$, and that any such 0-cycle can be obtained as a boundary of an appropriate sum of the relative 1-cycles in our collection.

These two observations imply that our collection of cycles is a generating set in $H_1^-(\hat{S}, \Sigma; \mathbb{Z})$. It remains to note that its cardinality coincides with the rank of the free Abelian group $H_1^-(\hat{S}, \Sigma; \mathbb{Z})$.

Note that a preimage of a simple pole does not belong to the set $\hat{P}_1, \ldots, \hat{P}_N$. Thus, a preimage of an arc $[P_i, P_{i+1}]$ having a simple pole as one of the endpoints does not define a cycle in $H_1(\hat{S}, \{\hat{P}_1, \ldots, \hat{P}_N\}; \mathbb{Z})$. However, since a simple pole is always a branching

point, the *difference* of the preimages of such arc is already a well-defined relative cycle in $H_1(\hat{S}, \{\hat{P}_1, \dots, \hat{P}_N\}; \mathbb{Z})$.

Let $Q(d_1, \ldots, d_n)$ be the ambient stratum for the meromorphic quadratic differential (\mathbb{CP}^1, ψ) . The subspace $H^1_-(\hat{S}, \{\hat{P}_1, \ldots, \hat{P}_N\}; \mathbb{C})$ in the relative *cohomology* antiinvariant with respect to the natural involution defines local coordinates in the stratum, see [15].

3.2. Ĥomologous saddle connections

We follow the exposition in [39] introducing the notions of a *rigid* collection of saddle connections and of $\hat{h}omologous$ saddle connections. Consider a flat sphere S corresponding to a meromorphic quadratic differential (\mathbb{CP}^1, ψ) with at most simple poles. Any saddle connection on the flat sphere S persists under small deformations of S inside $Q(\alpha)$. It might happen that any deformation of a given flat surface which shortens some specific saddle connection necessarily shortens some other saddle connections. We say that a collection $\{\gamma_1, \ldots, \gamma_n\}$ of saddle connections is *rigid* if any sufficiently small deformation of the flat surface inside the stratum preserves the proportions $|\gamma_1| : |\gamma_2| : \ldots : |\gamma_n|$ of the lengths of all saddle connections in the collection.

Consider the canonical double cover \hat{S} over S defined in §3.1. Given a saddle connection γ on S choose an orientation of γ and let γ', γ'' be its lifts to the double cover \hat{S} endowed with the orientation inherited from γ . If $[\gamma'] = -[\gamma'']$ as cycles in $H_1(\hat{S}, \{\hat{P}_1, \dots, \hat{P}_N\}; \mathbb{Z})$ we let $[\hat{\gamma}] := [\gamma']$, otherwise we define $[\hat{\gamma}]$ as $[\hat{\gamma}] := [\gamma'] - [\gamma'']$. It immediately follows from the above definition that the cycle $[\hat{\gamma}]$ defined by a saddle connection γ is always *primitive* in $H_1(\hat{S}, \{\hat{P}_1, \dots, \hat{P}_N\}; \mathbb{Z})$ and belongs to $H_1^-(\hat{S}, \{\hat{P}_1, \dots, \hat{P}_N\}; \mathbb{Z})$.

Following [39] we introduce the following

DEFINITION 3.2. – The saddle connections γ_1, γ_2 on a flat surface S defined by a quadratic differential q are \hat{h} omologous if $[\hat{\gamma}_1] = [\hat{\gamma}_2]$ in $H_1(\hat{S}, \hat{P}; \mathbb{Z})$ under an appropriate choice of orientations of γ_1, γ_2 . (The notion "homologous in the relative homology with local coefficients defined by the canonical double cover induced by a quadratic differential" is unbearably bulky, so we introduced an abbreviation " \hat{h} omologous". We stress that the circumflex over the "h" is quite meaningful: as it is indicated in the definition, the corresponding cycles are homologous on the double cover.)

Note that since there is no canonical way to enumerate the preimages γ', γ'' of a saddle connection γ on the double cover, the cycle $[\hat{\gamma}]$ is defined only up to a sign, even when we fix the orientation of γ . Thus, γ_1 is homologous to γ_2 if and only if $[\hat{\gamma}_1] = \pm [\hat{\gamma}_2]$.

PROPOSITION (H. Masur, A. Z.). – Let S be a flat surface corresponding to a meromorphic quadratic differential q with at most simple poles. A collection $\gamma_1, \ldots, \gamma_n$ of saddle connections on S is rigid if and only if all saddle connections $\gamma_1, \ldots, \gamma_n$ are homologous.

There is an obvious geometric test for deciding when saddle connections γ_1, γ_2 on a translation surface S are homologous: it is sufficient to check whether $S \setminus (\gamma_1 \cup \gamma_2)$ is connected or not (provided $S \setminus \gamma_1$ and $S \setminus \gamma_2$ are connected). It is slightly less obvious to check whether saddle connections γ_1, γ_2 on a flat surface S with nontrivial linear holonomy are homologous or not. In particular, a pair of closed saddle connections might be homologous in the usual sense, but not homologous; a pair of closed saddle connections might be

ĥomologous even if one of them represents a loop homologous to zero, and the other does not; finally, a saddle connection joining a pair of *distinct* singularities might be ĥomologous to a saddle connection joining a singularity to itself, or joining another pair of distinct singularities. The following statement provides a geometric criterion for deciding when two saddle connections are ĥomologous.

PROPOSITION (H. Masur, A. Z.). – Let S be a flat surface corresponding to a meromorphic quadratic differential q with at most simple poles. Two saddle connections γ_1, γ_2 on S are homologous if and only if they have no interior intersections and one of the connected components of the complement $S \setminus (\gamma_1 \cup \gamma_2)$ has trivial linear holonomy. Moreover, if such a component exists, it is unique.

Now everything is ready for the proof of Proposition 2.1.

Proof of Proposition 2.1. – Configurations I and II involve a single saddle connection. Using the above criterion it is immediate to check that all saddle connections involved in configurations III and IV are \hat{h} omologous. Thus, these configurations are rigid, and we can find them on almost every flat surface in the stratum.

Theorem 2.2 in [8] applies general results from [39] to classify all possible configurations of \hat{h} omologous saddle connections on \mathbb{CP}^1 , and shows that there are no such configurations different from types I–IV.

To complete the proof it remains to apply Proposition 4 from [39] which claims that for almost every flat surface in any stratum, two saddle connections are parallel if and only if they are \hat{h} omologous. This statement is proved following the lines of Proposition 7.4 in [21]; see also the analogous proof of Proposition 2.2 in §3.3 below.

3.3. The subspace of billiards

Consider now the map

$$\mathscr{B}(k_1,\ldots,k_n) \hookrightarrow \mathscr{Q}(k_1-2,\ldots,k_n-2).$$

In the chosen coordinates in $H^1_-(\hat{S}, \{\hat{P}_1, \dots, \hat{P}_N\}; \mathbb{C})$ the image of a directional billiard Π is presented by a point

(3.1)
$$\left(2 \int_{P_1}^{P_2} dz, \dots, 2 \int_{P_{n-2}}^{P_{n-1}} dz \right) = \left(2|P_1P_2|e^{i\phi}, 2|P_2P_3|e^{(k_2\pi)/2 + i\phi}, \dots, 2|P_{n-3}P_{n-2}|e^{(k_2 + \dots + k_{n-2})\pi/2 + i\phi} \right)$$

The components of the projection of this vector to the $H^1_-(\hat{S}, \{\hat{P}_1, \dots, \hat{P}_N\}; \mathbb{R})$ are of the form

$$\pm 2\sin(\phi)|P_iP_{i+1}|$$
 or $\pm 2\cos(\phi)|P_iP_{i+1}|$

depending on the parity of $k_2 + \ldots + k_i$. Thus, for ϕ different from an integer multiple of $\pi/2$ the composition map $T_* \mathcal{B} \to H^1_-(\hat{S}, \{\hat{P}_1, \ldots, \hat{P}_N\}; \mathbb{R})$ is a surjective map. We have proved

PROPOSITION 3.3. – Consider the canonical local embedding

$$\mathscr{B}(k_1,\ldots,k_n) \hookrightarrow \mathscr{Q}(k_1-2,\ldots,k_n-2).$$

For almost all directional billiards in $\mathcal{B}(k_1, \ldots, k_n)$ the projection of the tangent space $T_*\mathcal{B}(k_1, \ldots, k_n)$ to the unstable subspace of the Teichmüller geodesic flow is a surjective map.

We complete this section with a proof of Proposition 2.2.

Proof of Proposition 2.2. – By assumption we do not consider generalized diagonals and closed billiard trajectories parallel to the sides of the polygon. First note that without loss of generality we can consider only generalized diagonals: any closed regular trajectory makes part of a band which is bounded on both sides by a (chain of) generalized diagonals, see Figure 3.

Let $l_m = |P_m P_{m+1}|$ for m = 1, ..., n-2. Recall that l_i are the independent coordinates in the space $\mathscr{B}(k_1, ..., k_n)$. Unfolding the billiard along a generalized diagonal we see that every generalized diagonal (non parallel to one of the sides of the polygon) defines a relation

$$\frac{\sum a_i l_i}{\sum b_j l_j} = \tan(\phi),$$

where $0 < \phi < \pi/2$; the sum in the numerator is taken over the vertical sides of the polygon; the sum in the denominator is taken over the horizontal sides; and all a_i and b_j are integers. Since the second generalized diagonal has a segment going in the same direction ϕ , it also defines a relation

$$rac{\sum c_i l_i}{\sum d_j l_j} = an(\phi),$$

where the sum in the numerator is taken over the vertical sides of the polygon; the sum in the denominator is taken over the horizontal sides; and all c_i and d_j are integers.

Each generalized diagonal determines a saddle connection γ on the corresponding flat sphere, which in turn defines a cycle $\pm \hat{\gamma} \in H_1^-(\hat{S}, \{\hat{P}_1, \dots, \hat{P}_N\}; \mathbb{Z})$. Moreover, up to appropriate choice of signs of the basis vectors in the basis from §3.1 the cycle corresponding to the first generalized diagonal has the form $\hat{c}_1 := \sum a_i \hat{\gamma}_i + \sum b_j \hat{\gamma}_j$ and the cycle corresponding to the second generalized diagonal has the form $\hat{c}_2 := \sum c_i \hat{\gamma}_i + \sum d_j \hat{\gamma}_j$.

Assume that the two generalized diagonals do not make part of any of configurations I– IV. By the result of Boissy [8] there are no configurations of homologous saddle connections on \mathbb{CP}^1 other than configurations I–IV. This implies that the corresponding saddle connections are not homologous, and, hence, the cycles \hat{c}_1 and \hat{c}_2 are not proportional. This implies that the relation

$$\frac{\sum a_i l_i}{\sum b_j l_j} = \frac{\sum c_i l_i}{\sum d_j l_j}$$

is a nontrivial relation on coordinates l_1, \ldots, l_{n-2} . Thus, the set, satisfying this condition, has measure zero. Taking a union over the countable collection of possible conditions (countable, because we have to consider all possible collections of integers a_i, b_j, c_i, d_j) we still get a set of measure zero.

4° SÉRIE – TOME 49 – 2016 – Nº 6

4. Values of the Siegel-Veech constants

In this section, we derive formulas for the Siegel-Veech constant of each configuration of saddle connections. There are two kinds of formulas. The first kind expresses the Siegel-Veech constant as a ratio of volumes of strata, with explicit combinatorial coefficients. These formulas will be stated and proved in this section. The second kind of formula gives the Siegel-Veech constants as numbers (depending only on the stratum and the configuration). They are proved by plugging the expression (1.1) from Theorem 1.1 into the formula of the first kind. We also present these formulas here; however, Theorem 1.1 will only be proved in §5. For this reason we have attempted to separate the formulas which depend on Theorem 1.1 from the formulas which do not.

The results obtained in this section are based on techniques developed in the papers [19], [21], and [39] written in collaboration with H. Masur.

4.1. Normalization of the volume element

Recall that for any flat surface S in any stratum $Q(d_1, \ldots, d_k)$ we have a canonical ramified double cover $\hat{S} \to S$ such that the induced quadratic differential on the Riemann surface \hat{S} is a global square of a holomorphic Abelian differential. We have seen in §3.1 that the subspace $H_{-}^{1}(\hat{S}, \{\hat{P}_{1}, \ldots, \hat{P}_{N}\}; \mathbb{C})$ antiinvariant with respect to the induced action of the hyperelliptic involution on relative cohomology provides local coordinates in the corresponding stratum $Q(d_{1}, \ldots, d_{n})$ of quadratic differentials. We define a lattice in $H_{-}^{1}(\hat{S}, \{\hat{P}_{1}, \ldots, \hat{P}_{N}\}; \mathbb{C})$ as the subset of those linear forms which take values in $\mathbb{Z} \oplus i\mathbb{Z}$ on $H_{-}^{1}(\hat{S}, \{\hat{P}_{1}, \ldots, \hat{P}_{N}\}; \mathbb{Z})$.

We define the volume element $d\mu$ on $Q(d_1, \ldots, d_k)$ as the linear volume element in the vector space $H^1_-(\hat{M}^2_g, \{\hat{P}_1, \ldots, \hat{P}_N\}; \mathbb{C})$ normalized in such way that the fundamental domain of the above lattice has volume 1.

We warn the reader that for N > 1 this lattice is a proper sublattice of index 4^{N-1} of the lattice

$$H^1_{-}(\widehat{S}, \{\widehat{P}_1, \dots, \widehat{P}_N\}; \mathbb{C}) \cap H^1(\widehat{S}, \{\widehat{P}_1, \dots, \widehat{P}_N\}; \mathbb{Z} \oplus i\mathbb{Z}).$$

Indeed, if a flat surface S defines a lattice point for our choice of the lattice, then the holonomy vector along a saddle connection joining distinct singularities can be half-integer. (However, the holonomy vector along any *closed* saddle connection is still always integer.)

The choice of one or another lattice is a matter of convention. Our choice makes formulae relating enumeration of pillowcase covers to volumes simpler; see Appendix B. Another advantage of our choice is that the volumes of the strata $Q(d, -1^{d+4})$ and of the hyperelliptic components of the corresponding strata of Abelian differentials are the same (up to the factors responsible for the numbering of zeroes and of simple poles).

CONVENTION 4.1. – Similar to the case of Abelian differentials we now choose a real hypersurface $Q_1(d_1, \ldots, d_k)$ of flat surfaces of fixed area in the stratum $Q(d_1, \ldots, d_k)$. We abuse notation by denoting by $Q_1(d_1, \ldots, d_k)$ the space of flat surfaces of area 1/2 (so that the canonical double cover has area 1).

The volume element $d\mu$ in the embodying space $\mathcal{Q}(d_1, \ldots, d_k)$ induces naturally a volume element $d\mu_1$ on the hypersurface $\mathcal{Q}_1(d_1, \ldots, d_k)$ in the following way. There is a natural \mathbb{C}^* -action on $\mathcal{Q}(d_1, \ldots, d_k)$: having $\lambda \in \mathbb{C}^*$ we associate to the flat surface $S = (\mathbb{C}P^1, q)$ the flat surface

(4.1)
$$\lambda \cdot S := (\mathbb{C}P^1, \lambda^2 \cdot q).$$

In particular, we can represent any $S \in Q(d_1, \ldots, d_k)$ as $S = rS_{(1)}$, where $r \in \mathbb{R}_+$, and where $S_{(1)}$ belongs to the "hyperboloid": $S_{(1)} \in Q_1(d_1, \ldots, d_k)$. Geometrically this means that the metric on S is obtained from the metric on $S_{(1)}$ by rescaling with linear coefficient r. In particular, vectors associated to saddle connections on $S_{(1)}$ are multiplied by r to give vectors associated to corresponding saddle connections on S. It means also that area(S) = $r^2 \cdot \operatorname{area}(S_{(1)}) = r^2/2$, since $\operatorname{area}(S_{(1)}) = 1/2$. We define the *volume element* $d\mu_1$ on the "hyperboloid" $Q_1(d_1, \ldots, d_k)$ by disintegration of the volume element $d\mu$ on $Q(d_1, \ldots, d_k)$:

(4.2)
$$d\mu = r^{2n-1} \, dr \, d\mu_1 \,,$$

where

$$2n = \dim_{\mathbb{R}} \mathcal{Q}(d_1, \dots, d_k) = 2 \dim_{\mathbb{C}} \mathcal{Q}(d_1, \dots, d_k) = 2(k-2).$$

Using this volume element we define the total volume of the stratum $Q_1(d_1, \ldots, d_k)$:

(4.3)
$$\operatorname{Vol} \mathcal{Q}_1(d_1, \dots, d_k) := \int_{\mathcal{Q}_1(d_1, \dots, d_k)} d\mu_1$$

For a subset $E \subset Q_1(d_1, \ldots, d_k)$ we let $C(E) \subset Q_1(d_1, \ldots, d_k)$ denote the "cone" based on E:

(4.4)
$$C(E) := \{ S = rS_{(1)} \mid S_{(1)} \in E, \ 0 < r \le 1 \}.$$

Our definition of the volume element on $Q_1(d_1, \ldots, d_k)$ is consistent with the following normalization:

(4.5)
$$\operatorname{Vol}(\mathcal{Q}_1(d_1,\ldots,d_k)) = \dim_{\mathbb{R}} \mathcal{Q}(d_1,\ldots,d_k) \cdot \mu(C(\mathcal{Q}_1(d_1,\ldots,d_k)))$$

where $\mu(C(Q_1(d_1,\ldots,d_k)))$ is the total volume of the "cone" $C(Q_1(d_1,\ldots,d_k)) \subset Q(d_1,\ldots,d_k)$ measured by means of the volume element $d\mu$ on $Q(d_1,\ldots,d_k)$ defined above.

4.2. $SL(2, \mathbb{R})$ -action

There is an action of $SL(2, \mathbb{R})$ on the moduli space of quadratic differentials that preserves the stratification, and moreover, preserves ([33, 47]) the measures on Q and Q_1 described above. Recall that a quadratic differential q determines (and is determined by) by an atlas of charts to \mathbb{C} whose transition maps are of the form $z \mapsto \pm z + c$. Since $SL(2, \mathbb{R})$ acts on \mathbb{C} via linear maps on \mathbb{R}^2 , given a quadratic differential q and a matrix $g \in SL(2, \mathbb{R})$, define the quadratic differential $g \cdot q$ via post-composition of charts with g. This action generalizes the action of $SL(2, \mathbb{R})$ on the space of (unit-area) flat tori $SL(2, \mathbb{R})/SL(2, \mathbb{Z})$. Note that $SL(2, \mathbb{R})$ preserves the area of the quadratic differential q, and in particular it preserves the level surface $Q_1(d_1, \ldots, d_k)$.

4.3. Strata of surfaces with marked points

In this section we shall also consider the strata $Q_1(\alpha)$ of flat surfaces $S = (\mathbb{CP}^1, q)$ where we mark a regular point on the surface. Say, $Q_1(2, 1^2, 0, -1^8)$ will denote the stratum of meromorphic quadratic differentials on \mathbb{CP}^1 with one zero of order 2, two zeroes of order 1 denoted by 1^2 , eight simple poles -1^8 , and one additional marked point: "zero of order 0".

Let $\alpha = \{d_1, \ldots, d_k\}$ be a set with multiplicities, where $d_i \in \{-1, 1, 2, 3, \ldots\}$ for $i = 1, \ldots, k$, and $\sum d_i = -4$. A stratum with a marked point $\mathcal{Q}(0, d_1, \ldots, d_k)$ has the natural structure of a fiber bundle over the corresponding stratum without marked points $\mathcal{Q}(d_1, \ldots, d_k)$. This bundle has the surface S (punctured at all singularities P_1, \ldots, P_k) as a fiber over the "point" $S \in \mathcal{Q}(d_1, \ldots, d_k)$. Clearly, the dimension of the "universal curve" $\mathcal{Q}(0, d_1, \ldots, d_k)$ satisfies

(4.6)
$$\dim_{\mathbb{C}} \mathcal{Q}(0, d_1 \dots, d_k) = \dim_{\mathbb{C}} \mathcal{Q}(d_1 \dots, d_k) + 1 = k - 1.$$

By convention we *always* mark a point on a flat torus. We denote the corresponding stratum $\mathcal{H}(0)$; it has dimension two: dim_C $\mathcal{H}(0) = 2$.

The natural measure on the stratum $Q(0, d_1 \dots, d_k)$ with marked points disintegrates into a product measure, where the measure $d\mu_0$ along the fiber is proportional to the Lebesgue measure on S induced by the flat metric on S, and the measure on the base $Q(d_1 \dots, d_k)$ is the natural measure $d\mu_1$ on the corresponding stratum taken without marked points.

When the flat structure on S is defined by a *quadratic* differential the measure of the fiber S is different from the measure of the analogous fiber S with the flat structure defined by an *Abelian* differential. Namely, by Convention 4.1 the area of the surface S in terms of our flat metric defined by the quadratic differential is 1/2. Note also, that a saddle connection γ joining a zero and a marked point and having half-integer linear holonomy $\pm \text{hol}(\gamma) \in \mathbb{R}^2$ defines an integer cycle $\hat{\gamma} \in H_1^-(\hat{M}_g^2, \{\hat{P}_1, \dots, \hat{P}_N\}; \mathbb{Z})$. Hence, our choice of the fundamental domain of the lattice in the relative cohomology $H_-^1(\hat{M}_g^2, \{\hat{P}_1, \dots, \hat{P}_N\}; \mathbb{C})$ described in §4.1 implies that the component $d\mu_0$ of the disintegrated measure along the fiber S is

$$(4.7) d\mu_0 = 4dx \, dy$$

i.e., 4 times the standard Lebesgue measure coming from the flat metric. This gives $\mu_0(S) = 2$ for the total measure of each fiber, which implies the following relation between the volumes of the strata:

(4.8)
$$\operatorname{Vol} Q_1(0, d_1, \dots, d_k) = 2 \operatorname{Vol} Q_1(d_1, \dots, d_k).$$

Recall that v(0) = 2, see (1.2); so this is coherent with Formula (1.1) for the volume.

4.4. Volume of a stratum of disconnected flat surfaces

It will be convenient to consider the strata $Q(\alpha') = Q(\alpha'_a) \times Q(\alpha'_b)$, of closed flat surfaces S having two components $S_a \sqcup S_b$ of prescribed types. Such strata play especially important role in the context of the *principal boundary* discussed in §4.6. In the consideration below each of α'_a, α'_b might contain an entry "0" or not. In other words, the strata $Q(\alpha'_a), Q(\alpha'_b)$ are allowed to have a marked point.

CONVENTION 4.2. – Using notation $\alpha' = \alpha'_a \sqcup \alpha'_b$ for the strata $Q(\alpha')$ of disconnected surfaces we assume that we keep track of how α' is partitioned into α'_a and α'_b .

We shall need the expressions for the volume element and for the total volume of such strata. The corresponding expressions for the strata of Abelian differentials were obtained in §6.2 pp. 81–82 in [21]. Though the corresponding formula translates to the strata of quadratic differentials without any difficulties we present this simple calculation since it is very instructive in view of calculation of Siegel-Veech constants performed below.

We write
$$S_i = r_i S_i^{(1)}$$
, where area $\left(S_i^{(1)}\right) = \frac{1}{2}$; $i \in \{a, b\}$. Then area $\left(S_i\right) = r_i^2 \cdot \frac{1}{2}$. Let
 $n_i := \dim_{\mathbb{C}} \mathcal{Q}(\alpha'_i)$; $n := \dim_{\mathbb{C}} \mathcal{Q}(\alpha') = n_a + n_b$.

Let $d\mu^a$ (correspondingly $d\mu^b$) be the volume element on the stratum $Q(\alpha'_a)$ (correspondingly $Q(\alpha'_b)$). Let $d\mu^a_1$ (correspondingly $d\mu^b_1$) be the hypersurface volume element on the "unit hyperboloid" $Q_1(\alpha'_a)$ (correspondingly $Q_1(\alpha'_b)$). We have

$$d\mu(S) = d\mu^a(S_a) \cdot d\mu^b(S_b) = r_a^{2n_a - 1} r_b^{2n_b - 1} dr_a \, dr_b \, d\mu_1^a \, d\mu_1^b.$$

Set

$$W = \operatorname{Vol} \mathcal{Q}_1(\alpha'_a) \cdot \operatorname{Vol} \mathcal{Q}_1(\alpha'_b).$$

Then,

$$\mu(C(\mathcal{Q}_1(\alpha')) = W \cdot \int_{\substack{r_a^2 + r_b^2 \le 1\\r_a > 0; r_b > 0}} r_a^{2n_a - 1} r_b^{2n_b - 1} \, dr_a \, dr_b = W \cdot \frac{1}{4} \frac{(n_a - 1)!(n_b - 1)!}{n!},$$

where we have left the computation of the integral over the disk as an exercise. Hence, applying (4.5) we get

(4.9)
$$\operatorname{Vol} \mathcal{Q}_{1}(\alpha') = 2n \cdot \mu(C(\mathcal{Q}_{1}(\alpha')))$$
$$= \frac{1}{2} \cdot \frac{(\dim_{\mathbb{C}} \mathcal{Q}(\alpha'_{a}) - 1)!(\dim_{\mathbb{C}} \mathcal{Q}(\alpha'_{b}) - 1)!}{(\dim_{\mathbb{C}} \mathcal{Q}(\alpha') - 1)!} \cdot \operatorname{Vol} \mathcal{Q}_{1}(\alpha'_{a}) \cdot \operatorname{Vol} \mathcal{Q}_{1}(\alpha'_{b}).$$

Repeating literally the same arguments we obtain the corresponding formula for the volume elements:

(4.10)
$$d\mu_1 = \frac{1}{2} \cdot \frac{(\dim_{\mathbb{C}} \mathcal{Q}(\alpha'_a) - 1)! (\dim_{\mathbb{C}} \mathcal{Q}(\alpha'_b) - 1)!}{(\dim_{\mathbb{C}} \mathcal{Q}(\alpha') - 1)!} \cdot d\mu_1^a \, d\mu_1^b.$$

4.5. Reduction to ergodic theory

In this section we recall the strategy given in [19] to obtain the quadratic asymptotics in Theorem 2.3.

Fix an unordered collection (d_1, \ldots, d_n) of integers $d_i \in \mathbb{N} \cup \{-1\}, i = 1, \ldots, n$, satisfying $\sum_{i=1}^n d_i = -4$, and let Q_1 denote the stratum $Q_1(d_1, \ldots, d_n)$. Note that every such stratum is nonempty and connected. Let μ_1 denote the canonical PSL $(2, \mathbb{R})$ -invariant measure on Q_1 . Fix a configuration \mathcal{C} as in §2.1. To each saddle connection we associate a holonomy vector in the Euclidean plane \mathbb{R}^2 having the same length and the same line direction as the saddle connection. By convention the configuration III is represented by the closed saddle connection joining a zero to itself (the holonomy vector associated to the partner saddle connections are not oriented, the holonomy vector is defined up to a sign, so we actually consider a pair of opposite holonomy vectors $\pm \vec{v}$. Given a flat surface $S = (\mathbb{CP}^1, q) \in Q_1$, let $V_{\mathcal{C}}(S)$ be the set of holonomy vectors of saddle connections whose configuration is \mathcal{C} . For any flat surface S the set $V_{\mathcal{C}}(S)$ is a discrete subset of \mathbb{R}^2 . We are interested in the asymptotics of the number

(4.11)
$$N_{\mathscr{C}}(S,L) = \frac{1}{2} \big| V_{\mathscr{C}}(S) \cap B(0,L) \big|,$$

of saddle connections of type \mathcal{C} on the flat surface S of length at most L. The weight 1/2 in the above expression compensates the fact that each saddle connection is represented by two holonomy vectors $\pm \vec{v}$.

In the remainder of §6, the stratum Q and the configuration C are fixed. We will often omit C from the notation, and we will use the abbreviated notation q for the flat surface $S = (\mathbb{CP}^1, q)$.

4.5.1. Siegel-Veech formulas. – Given $f \in C_c(\mathbb{R}^2)$, define the Siegel-Veech transform $\widehat{f} : \mathcal{Q}_1 \to \mathbb{R}$ by

(4.12)
$$\widehat{f}(q) = \frac{1}{2} \sum_{v \in V_{\mathcal{C}}(q)} f(v)$$

We have the Siegel-Veech formula ([49], Theorem 0.5). There is a constant (called the Siegel-Veech constant) $b_{\mathcal{C}}(Q)$ so that:

(4.13)
$$\frac{1}{\mu_1(\mathcal{Q}_1)} \int_{\mathcal{Q}_1} \widehat{f}(q) d\mu_1(q) = b_{\mathcal{C}}(\mathcal{Q}) \int_{\mathbb{R}^2} f(x, y) \, dx dy.$$

Let

$$g_t = \begin{pmatrix} e^t & 0\\ 0 & e^{-t} \end{pmatrix}$$
$$r_{\theta} = \begin{pmatrix} \cos\theta & \sin\theta\\ -\sin\theta & \cos\theta \end{pmatrix}.$$

Let f be (a smoothed version of) the indicator function of the trapezoid \mathcal{T} defined by the points

$$(1,1), (-1,1), (1/2, 1/2), (-1/2, 1/2).$$

Note that the area of this trapezoid is 3/4.

We then have, for $t \gg 0$, and any $v \in \mathbb{R}^2$ ([19], Lemma 3.4):

(4.14)
$$\frac{1}{2\pi} \int_0^{2\pi} f(g_t r_\theta v) d\theta \approx \begin{cases} \frac{e^{-2t}}{\pi} & e^t/2 \le ||v|| \le e^t \\ 0 & \text{otherwise.} \end{cases}$$

(See [19] for the exact meaning of \approx .) Heuristically, the integral measures the proportion of angles θ so that $r_{\theta}v \in g_{-t}\mathcal{T}$. The trapezoid $g_{-t}\mathcal{T}$ has vertices at

$$(e^{-t}, e^t), (-e^{-t}, e^t), (e^{-t}/2, e^t/2), (-e^{-t}/2, e^t/2).$$

The range of (inverse) slopes is of size $2e^{-2t}$, and thus the length of the interval of θ 's satisfying $r_{\theta}v \in g_{-t}\mathcal{T}$ is also of size $2e^{-2t}$, if v has length in between $e^t/2$ and e^t , and zero otherwise. Dividing by 2π to get the proportion, we obtain (4.14). Combining (4.12) and (4.14), we obtain

(4.15)
$$\frac{e^{2t}}{2\pi} \int_0^{2\pi} \widehat{f}(g_t r_\theta q) d\theta \approx \frac{1}{\pi} \left(N_{\mathscr{C}}(q, e^t) - N_{\mathscr{C}}(q, e^t/2) \right).$$

ANNALES SCIENTIFIQUES DE L'ÉCOLE NORMALE SUPÉRIEURE

1337

4.5.2. Equidistribution results. - The equation (4.15) reduces the problem of studying

$$\lim_{t \to \infty} e^{-2t} N_{\mathcal{C}}(q, e^t)$$

to that of studying the limiting behavior of

$$\frac{1}{2\pi}\int_0^{2\pi}\widehat{f}(g_tr_\theta q)d\theta.$$

Assuming this limit exists, and is equal to c, a geometric series calculation shows

$$\lim_{t \to \infty} e^{-2t} N_{\mathscr{C}}(q, e^t) = \frac{4}{3}\pi c.$$

Assuming further that Lebesgue measure supported on the circles $\{g_t r_\theta q\}_{0 \le \theta < 2\pi}$ converges, as $t \to \infty$, to the absolutely continuous $\operatorname{SL}(2, \mathbb{R})$ -invariant measure μ_1 on \mathcal{Q}_1 , we would have that $c = \frac{1}{\mu_1(\mathcal{Q}_1)} \int_{\mathcal{Q}_1} \widehat{f}(q) d\mu_1(q)$, and then using (4.13), we would obtain, since the area of the trapezoid is 3/4,

$$\lim_{t \to \infty} e^{-2t} N_{\mathcal{C}}(q, e^t) = \pi b_{\mathcal{C}}(Q).$$

In fact, this is the approach used in [19]. There, the key tool is a general ergodic theorem on $SL(2, \mathbb{R})$ -actions, proved by A. Nevo [43] which shows

$$\lim_{t \to \infty} \int_0^{2\pi} \widehat{f}(g_t r_\theta q) \, d\theta = \frac{1}{\mu_1(\mathcal{Q}_1)} \int_{\mathcal{Q}_1} \widehat{f}(q) \, d\mu_1(q),$$

for almost every $q \in Q$. However, since the set of billiards has measure 0, this does not yield any information about them. We will instead use Theorem C.1 to obtain our results.

4.6. Siegel-Veech constants and the principal boundary of strata

In this section we present a strategy for evaluating the Siegel-Veech constants. This strategy was successfully applied in [21] to compute all Siegel-Veech constants for all connected components of the strata of Abelian differentials. In this section we present the general scheme elaborated in [21] and developed in [39]. In the further sections we adjust it to the concrete cases of configurations of saddle connections I–IV described in §2.1.

Fix a stratum $Q(\alpha)$ of meromorphic quadratic differentials on \mathbb{CP}^1 , where $\alpha = \{d_1, \ldots, d_k\}$. Consider a configuration \mathcal{C} of one of the types I–IV (in the case of general strata in higher genus it would be any configuration of homologous saddle connections). We have seen in §4.5 that to each flat surface $S \in Q(\alpha)$ we can associate a discrete subset $V_{\mathcal{C}}(S) \subset \mathbb{R}^2$ of holonomy vectors of saddle connections whose configuration is \mathcal{C} . By construction the set $V_{\mathcal{C}}(S)$ is centrally symmetric with respect to the origin. To any function f with compact support on \mathbb{R}^2 Formula (4.12) associates its *Siegel-Veech transform* \hat{f} defined on the stratum Q. By Definition (4.12), choosing the characteristic function χ_L of a closed disk of radius L centered at the origin of \mathbb{R}^2 as a function f, we get as $\hat{\chi}_L(S)$ the counting function $N_{\mathcal{C}}(S, L)$ of the number of saddle connections of type \mathcal{C} and of length at most L on the flat surface S defined by (4.11).

Applying Siegel-Veech Formula (4.13) we obtain

(4.16)
$$\frac{1}{\mu_1(\mathcal{Q}_1)} \int_{\mathcal{Q}_1} \widehat{\chi}_L(S) \, d\mu_1(S) = b_{\mathcal{C}}(\mathcal{Q}) \int_{\mathbb{R}^2} \chi_L(x,y) \, dx dy = b_{\mathcal{C}}(\mathcal{Q}) \cdot \pi L^2.$$

4° SÉRIE – TOME 49 – 2016 – Nº 6

By the results of A. Eskin and H. Masur (Theorems 2.1 and 2.2 in [19]), for almost all flat surfaces S in the stratum Q_1 one has

(4.17)
$$N_{\mathscr{C}}(S,L) = \widehat{\chi}_L(S) \sim b_{\mathscr{C}} \cdot \pi L^2$$

with the same constant $b_{\mathcal{C}}$ as in (4.16).

Formula (4.16) can be applied to $\hat{\chi}_L$ for any value of L. In particular, instead of taking large L we can choose a very small $L = \varepsilon \ll 1$. The corresponding function $\hat{\chi}_{\varepsilon}(S)$ counts how many (collections of) ε -short saddle connections (closed geodesics) of the type \mathcal{C} we can find on a flat surface $S \in Q$.

Consider a subset $Q_1^{\varepsilon}(\mathcal{C}) \subset Q_1$ of surfaces of area 1/2 having a saddle connection shorter than ε . Consider a smaller subset $Q_1^{\varepsilon,\text{thin}} \subset Q_1^{\varepsilon}$ of those surfaces of area 1/2 in Q_1 which have at least two distinct collections of homologous saddle connections of type \mathcal{C} and of length at most ε . Finally, define $Q_1^{\varepsilon,\text{thick}}$ as the complement $Q_1^{\varepsilon} - Q_1^{\varepsilon,\text{thin}}$.

For the flat surfaces S outside of the subset $\mathcal{Q}_1^{\varepsilon}(\mathcal{C})$ there are no saddle connections of the type \mathcal{C} shorter than ε , so $\widehat{\chi}_{\varepsilon}(S) = 0$ for such surfaces. For surfaces S from the subset $\mathcal{Q}_1^{\varepsilon,\text{thick}}(\mathcal{C})$ there is exactly one collection like this, so $\widehat{\chi}_{\varepsilon}(S) = 1$. Finally, for the surfaces S from the remaining subset $\mathcal{Q}_1^{\varepsilon,\text{thin}}(\mathcal{C})$ one has $\widehat{\chi}_{\varepsilon}(S) \ge 1$. Though $\widehat{\chi}_{\varepsilon}(S)$ might be large on $\mathcal{Q}_1^{\varepsilon,\text{thin}}$ the measure of this subset is so small that

$$\int_{\mathcal{Q}_1^{\varepsilon, \min}(\mathcal{C})} \widehat{\chi}_{\varepsilon}(S) \ d\mu_1 = o(\varepsilon^2)$$

(see Lemma 7.3 in [21] based on Theorem 5.1 in [19]. Formally speaking, Lemma 7.3 in [21] is formulated for the strata of Abelian differentials; however, the proof applies to strata of quadratic differentials without any modifications).

Hence

$$\int_{\mathcal{Q}_1} \widehat{\chi}_{\varepsilon}(S) \ d\mu_1 = \operatorname{Vol} \mathcal{Q}_1^{\varepsilon, \operatorname{thick}}(\mathcal{C}) \ + \ o(\varepsilon^2).$$

This latter volume is almost the same as the volume Vol $Q_1^{\varepsilon}(\mathcal{C})$, namely, by Corollary 7.2 in [21] based on formula (7) on page 518 in Section 10 of [37], one has Vol $Q_1^{\varepsilon}(\mathcal{C}) =$ Vol $Q_1^{\varepsilon,\text{thick}}(\mathcal{C}) + o(\varepsilon^2)$. (Once again, though Corollary 7.2 in [21] is formulated for strata of Abelian differentials the proof applies without any changes to strata of quadratic differentials. See also Theorem 1.4 in [44] generalizing the result of [37].)

Taking into consideration that

$$\int_{\mathbb{R}^2} \chi_{\varepsilon}(x, y) \, dx \, dy = \pi \varepsilon^2$$

and applying Siegel-Veech Formula (4.16) we get

$$\frac{\operatorname{Vol} \mathcal{Q}_1^{\varepsilon}(\mathcal{C})}{\operatorname{Vol} \mathcal{Q}_1} + o(\varepsilon^2) = b_{\mathcal{C}} \cdot \pi \varepsilon^2$$

which implies the following formula for the Siegel-Veech constant $b_{\mathcal{C}}$:

(4.18)
$$b_{\mathcal{C}} = \lim_{\varepsilon \to 0} \frac{1}{\pi \varepsilon^2} \cdot \frac{\operatorname{Vol} \, \mathcal{Q}_1^{\varepsilon}(\mathcal{C})}{\operatorname{Vol} \, \mathcal{Q}_1}.$$

We complete this section by establishing an elementary relation between the Siegel-Veech constant $b_{\mathcal{C}}$ used in §6 and in §4.6 and the Siegel-Veech constant $c_{\mathcal{C}}$ used in §2. Recall that counting function (4.17)

$$N_{\mathscr{C}}(S,L) \sim b_{\mathscr{C}} \cdot \pi L^2$$

counts the number of saddle connections of type \mathscr{C} of length at most L on the flat surface $S \in \mathcal{Q}_1$. By convention 4.1 surfaces from \mathcal{Q}_1 have area 1/2. Thus, applying the asymptotic Formula (2.3) from Theorem 2.3 to the flat surface $S \in \mathcal{Q}_1$ we get

$$N_{\mathcal{C}}(S,L) \sim c_{\mathcal{C}} \cdot \frac{\pi L^2}{\text{area of } S} = 2c_{\mathcal{C}} \cdot \pi L^2,$$

which implies that

 $(4.19) b_{\mathscr{C}} = 2c_{\mathscr{C}}.$

4.7. Principal boundary

When saddle connections of a configuration \mathscr{C} are contracted by a continuous deformation, the limiting flat surface decomposes into one or several connected components represented by nondegenerate flat surfaces S'_1, \ldots, S'_m . Let the initial surface S belong to a stratum $\mathscr{Q}(\alpha)$, where α is the set with multiplicities $\{d_1, \ldots, d_k\}$. Let $\mathscr{Q}(\alpha'_j)$ be the stratum ambient for S'_j . The stratum $\mathscr{Q}(\alpha'_{\mathscr{C}}) = \mathscr{Q}(\alpha'_1) \sqcup \cdots \sqcup \mathscr{Q}(\alpha'_m)$ of disconnected flat surfaces $S'_1 \sqcup \cdots \sqcup S'_m$ is referred to as a *principal boundary* stratum of the stratum $\mathscr{Q}(\alpha)$. The principal boundary of any connected component of any stratum of Abelian differentials is described in [21]; the principal boundaries of strata of quadratic differentials are described in [39].

The papers [21], [39] also present the inverse construction. Consider any flat surface $S' := S'_1 \sqcup \cdots \sqcup S'_m \in \mathcal{Q}(\alpha'_{\mathcal{C}})$ in the principal boundary of $\mathcal{Q}(\alpha)$; consider a vector $\vec{v} \in \mathbb{R}^2 \simeq \mathbb{C}$ such that $\|\vec{v}\| \leq \varepsilon$. One can reconstruct a flat surface $S \in \mathcal{Q}(\alpha)$ endowed with a collection of saddle connections of the type \mathcal{C} such that the linear holonomy along saddle connections is represented by $\pm \vec{v}$, and such that the degeneration of S contracting the saddle connections in the collection gives the surface S'. When the configuration \mathcal{C} does not involve any cylinders, any flat surface $S' \in \mathcal{Q}_1$ and any holonomy vector \vec{v} define the surface $S \in \mathcal{Q}_1^{\varepsilon}(\mathcal{C})$, basically, up to some finite order ambiguity which can be explicitly computed. Moreover, the measure in $\mathcal{Q}_1^{\varepsilon}(\mathcal{C})$ disintegrates as the measure in $\mathcal{Q}_1(\alpha'_{\mathcal{C}})$ times the measure $d\mu_0$ in the space of parameters of the deformation. The latter space can be viewed as a finite cover of the space of holonomy vectors $\pm \vec{v}$, that is the quotient of the disk D_{ε}^2/\pm of radius ε over the central symmetry. As a result we get

(4.20)
$$\operatorname{Vol}\left(\mathcal{Q}_{1}^{\varepsilon}(\mathcal{C})\right) = (\operatorname{explicit factor}) \cdot \pi \varepsilon^{2} \cdot \operatorname{Vol} \mathcal{Q}_{1}(\alpha_{\mathcal{C}}') + o(\varepsilon^{2}).$$

Thus, in order to compute the constant $b_{\mathcal{C}}$ by Formula (4.18) it is sufficient to express the volume of Vol $Q_1(\alpha')$ in terms of the volumes Vol $Q_1(\alpha'_1), \ldots,$ Vol $Q_1(\alpha'_m)$, and to compute the explicit factor, responsible for the fixed finite number of flat surfaces $S \in Q_1^{\varepsilon}(\alpha)$ which correspond to a fixed flat surface $S' \in Q(\alpha'_{\mathcal{C}})$ in the boundary stratum and to a fixed holonomy vector \vec{v} . The first problem is simple; the answer to this problem is given in §4.4; the second problem is solved for configurations I–IV in the remaining part of §4.

4° SÉRIE - TOME 49 - 2016 - Nº 6



FIGURE 13. Breaking up a zero into two. In the particular case, when one of the newborn singularities is a simple pole, we can slit along the resulting saddle connection of length δ to get a surface with geodesic boundary of length 2δ .

The situation for configurations which involve a cylinder is slightly more complicated, but similar to the previous one. In both cases, applying Formula (4.18) and (4.20) we express the constant $b_{\mathcal{C}}$ as

(4.21)
$$b_{\mathcal{C}} = (\text{explicit combinatorial factor}) \cdot \frac{\prod_{j=1}^{k} \operatorname{Vol} \mathcal{Q}_{1}(\alpha'_{j})}{\operatorname{Vol} \left(\mathcal{Q}_{1}(\alpha)\right)}.$$

4.8. Surgeries on a flat surface

Consider a flat surface $S' \in Q_1(\alpha'_{\mathcal{C}})$ in a stratum of meromorphic quadratic differentials with at most simple poles on \mathbb{CP}^1 , possibly with a marked point. Fix some zero, or a simple pole (or the marked regular point) P_i . Consider a vector $\pm \vec{v} \in \mathbb{R}^2$, defined up to reversing the direction. Assume that \vec{v} is much shorter than the shortest saddle connection on S.

The papers [21] and [39] describe how to perform a small deformation of the surface S' breaking up the chosen singularity P_i of degree d_i into two singularities P'_i, P''_i of any two prescribed degrees d'_i and d''_i satisfying the relation $d'_i + d''_i = d_i$, where $d_i, d'_i, d''_i \in \{-1, 0, 1, 2, ...\}$. The deformation can be performed in such way that the holonomy vector of the resulting tiny saddle connection joining the newborn singularities P'_i, P''_i is exactly $\pm \vec{v}$. This deformation is described in details in Sections 8.1–8.2 in [21] and in Section 6.3 in [39]. When at least one of d'_i, d''_i is even, the deformation is local: it does not change the metric outside of a small neighborhood of P_i and it does not change the area of the flat surface. When both d'_i, d''_i are odd the deformation involves some arbitrariness and involves some small change of the area of the flat surface. A discussion in the original papers [21] and [39] explains why both issues might be neglected in our calculations.

The cone angles at the distinguished singularity is equal to $\pi(d_i + 2)$. Thus, there are $(d_i + 2)$ geodesic rays in linear direction $\pm \vec{v}$ adjacent to P_i . Take a small disk D_{ε}^2 of radius ε centered in the origin and consider its quotient D_{ε}^2/\pm over the action of central symmetry. Letting the vector $\pm \vec{v}$ vary in D_{ε}^2/\pm and taking care of normalization (4.7) of the measure $d\mu_0$ on D_{ε}^2/\pm we get a set of parameters of measure

(4.22)
$$(d+2) \cdot 4 \cdot \frac{\pi \varepsilon^2}{2} = 2(d+2) \cdot \pi \varepsilon^2.$$

For this configuration the "(explicit factor)" in (4.20) equals 2(d+2).

Consider now a particular case, when one of the newborn singularities P'_i, P''_i , say, P''_i is a simple pole. Since $d'_i + d''_i = d_i \ge -1$, the singularities P'_i, P''_i cannot be simple poles simultaneously. Making a slit along the short saddle connection joining P'_i to P''_i we create a surface \mathring{S} with geodesic boundary. Note that the cone angle at the singularity P''_i was π . This

means, that after opening up a slit, the point P''_i becomes a regular point of the boundary of \mathring{S} , see Figure 13. In other words, the boundary of \mathring{S} corresponds to a single closed geodesic with linear holonomy $\pm \vec{v}$.

Parallelogram construction

In order to construct the subset $Q_1^{\varepsilon}(\mathcal{C})$ corresponding to configuration II, we need another surgery. Given a flat surface $S' \in Q_1(\alpha')$ in a stratum of meromorphic quadratic differentials with at most simple poles on \mathbb{CP}^1 , given a pair of singularities P', P'' on S' and given a short vector $\pm \vec{v} \in \mathbb{R}^2$, we construct a surface with two boundary components creating a pair of small holes adjacent to the chosen singularities P', P''. The surgery is performed in such way that the holes have geodesic boundary with linear holonomy $\pm \vec{v}$. Let d', d'' be the degrees of singularities P', P'' respectively. The corresponding cone angles are $\pi(d'+2)$ and $\pi(d''+2)$. Thus, there are (d' + 2) geodesic rays in linear direction $\pm \vec{v}$ adjacent to P' and (d'' + 2)geodesic rays in linear direction $\pm \vec{v}$ adjacent to P''.

The corresponding surgery is described in Section 12.2 in [21] and in Section 6.1 in [39] as the "parallelogram construction". This is a nonlocal construction, so it is not canonical, and it changes slightly the area of the surface. Up to this ambiguity (which can be neglected in our computations as explained in [21] and in [39]), given the data as above, there are (d' + 2)(d'' + 2) ways to construct the described surface with boundary \mathring{S} . Take a small disk D_{ε}^2 of radius ε centered in the origin and consider its quotient D_{ε}^2/\pm over the action of central symmetry. Let the vector $\pm \vec{v}$ vary in D_{ε}^2/\pm . Note that contrary to the previous case, the saddle connection is now closed. Thus the measure along the fiber has the form

$$d\mu_0 = dx \, dy$$

and not the form (4.7) as before. This implies that for this configuration the set of parameters of deformation having holonomy vectors in D_{ϵ}^2/\pm has the measure

(4.23)
$$(d'+2)(d''+2) \cdot \frac{\pi\varepsilon^2}{2}.$$

For this configuration the "(explicit factor)" in (4.20) equals $\frac{(d'+2)(d''+2)}{2}$.

4.9. Type I: A simple saddle connection joining a fixed zero to a fixed pole or to a distinct fixed zero

Now we finally pass to explicit computation of the Siegel-Veech constants following the strategy described above.

Throughout the rest of this section $Q(d_1, \ldots, d_k)$ denotes any stratum of meromorphic quadratic differentials with at most simple poles on \mathcal{P} different from the stratum $Q(-1^4)$ of pillowcases.

THEOREM 4.3. – For the configuration C of saddle connections of type I, i.e., for saddle connections joining a fixed pair P_i , P_j of distinct singularities of orders d_i , d_j , the Siegel-Veech constant c_C is expressed as follows:

(4.24)
$$c_{\mathcal{C}} = (d_i + d_j + 2) \frac{\operatorname{Vol} Q_1(d_i + d_j, d_1, d_2, \dots, \widehat{d_i}, \dots, \widehat{d_j}, \dots, d_k)}{\operatorname{Vol} Q_1(d_1, \dots, d_k)}.$$

4° SÉRIE - TOME 49 - 2016 - Nº 6

After plugging in Theorem 1.1, we get:

(4.25)
$$c_{\mathcal{C}} = \frac{(d_i + d_j + 2)!! \ (d_i + 1)!! \ (d_j + 1)!!}{(d_i + d_j + 1)!! \ d_i!! \ d_j!!} \cdot \begin{cases} \frac{2}{\pi^2} & \text{when both } d_i, d_j \text{ are odd} \\ \frac{1}{2} & \text{otherwise.} \end{cases}$$

Proof of (4.24). – The principal boundary $Q_1(\alpha'_{\mathcal{C}})$ for this particular configuration \mathcal{C} is obtained by collapsing the saddle connection joining singularities of degrees d_i and d_j . This operation merges two singularities to a single one of degree $d = d_i + d_j$. Thus,

 $\alpha'_{\mathcal{C}} = \{d_i + d_j, d_1, d_2, \dots, \widehat{d_i}, \dots, \widehat{d_j}, \dots, d_k\}.$

By (4.20)

$$\operatorname{Vol}\left(\mathcal{Q}_1^{\varepsilon}(\mathcal{C})\right) = (\operatorname{explicit}\,\operatorname{factor})\cdot\pi\varepsilon^2\cdot\operatorname{Vol}\,\mathcal{Q}_1(\alpha_{\mathcal{C}}') \ + \ o(\varepsilon^2)$$

where the "(explicit factor) $\cdot \pi \varepsilon^2$ " in Formula (4.20) stands for the measure of the space of parameters of deformation corresponding to holonomy vectors in D_{ε}^2/\pm . This measure was computed in (4.22). Thus, we can rewrite (4.20) as

$$\operatorname{Vol}\left(\mathcal{Q}_{1}^{\varepsilon}(\mathcal{C})\right) = 2(d+2) \cdot \pi \varepsilon^{2} \cdot \operatorname{Vol}\left(\mathcal{Q}_{1}(d_{i}+d_{j},d_{1},d_{2},\ldots,\widehat{d_{i}},\ldots,\widehat{d_{j}},\ldots,d_{k})\right) + o(\varepsilon^{2}).$$

Applying (4.18) and (4.19) to the above expression we obtain (4.24).

We are ready to give a proof of Theorem 1.6 (based on (4.25) which would be proved in §5).

Proof of Theorem 1.6. – Let $Q(d_1, \ldots, d_k) = Q(1, -1^5)$. Let $d_i = 1, d_j = -1$. Applying (4.25) we get $c_I = 8/\pi^2$. Applying Theorem 2.4 to the L-shaped billiard as in Figure 4 we get the coefficient $\frac{2}{\pi}$ in the weak asymptotics of the number of generalized diagonals joining a fixed corner with angle $\frac{\pi}{2}$ with the corner with angle $\frac{3\pi}{2}$, and thus prove Formula 1.10 and Theorem 1.6.

4.10. Type II: A simple saddle connection joining a zero to itself

The configuration \mathscr{C} of type II consists of a single separatrix loop emitted from a fixed zero P_i of order d_i such that the total angle $(d_i + 2)\pi$ at the singularity P_i is split by the separatrix loop into two sectors having the angles $(d'_i + 3)\pi$ and $(d''_i + 3)\pi$. We assume that $d'_i, d''_i \ge -1$, so we *do not* have any cylinders filled with periodic geodesics for this configuration. The angles satisfy the natural relation

$$d'_i + d''_i = d_i - 4$$
 $d'_i, d''_i \ge -1$

which implies, in particular, that $d_i \ge 2$.

Our saddle connection separates the original surface S into two parts. Let $P_{i_1}, \ldots, P_{i_{k_1}}$ be the list of singularities (zeroes and poles) which belong to the first part and let $P_{j_1}, \ldots, P_{j_{k_2}}$ be the list of singularities (zeroes and poles) which belong to the second part. This information is part of the configuration of this saddle connection.

We assume that the initial surface S does not have any marked points; as usual we denote by d_n the order of the singularity P_n . The set with multiplicities $\{d_1, \ldots, d_k\}$ representing the orders of all singularities (zeroes and poles) on S can be obtained as a disjoint union of the following subsets:

$$\{d_1,\ldots,d_k\} = \{d_{i_1},\ldots,d_{i_{k_1}}\} \sqcup \{d_{j_1},\ldots,d_{j_{k_2}}\} \sqcup \{d_i\}.$$

ANNALES SCIENTIFIQUES DE L'ÉCOLE NORMALE SUPÉRIEURE

THEOREM 4.4. – The Siegel-Veech constant $c_{\mathcal{C}}$ for this configuration is expressed as follows:

$$(4.26) \quad c_{\mathcal{C}} = \frac{(d'_{i}+2)(d''_{i}+2)}{8} \cdot \\ \cdot \frac{(\dim_{\mathbb{C}} \mathcal{Q}(d'_{i}, d_{i_{1}}, \dots, d_{i_{k_{1}}}) - 1)! (\dim_{\mathbb{C}} \mathcal{Q}(d''_{i}, d_{j_{1}}, \dots, d_{j_{k_{2}}}) - 1)!}{(\dim_{\mathbb{C}} \mathcal{Q}(d_{1}, d_{2}, \dots, d_{k}) - 2)!} \cdot \\ \cdot \frac{\operatorname{Vol} \mathcal{Q}_{1}(d'_{i}, d_{i_{1}}, \dots, d_{i_{k_{1}}}) \cdot \operatorname{Vol} \mathcal{Q}_{1}(d''_{i}, d_{j_{1}}, \dots, d_{j_{k_{2}}})}{\operatorname{Vol} \mathcal{Q}_{1}(d_{1}, \dots, d_{k})}.$$

After plugging in Theorem 1.1 we get:

$$(4.27) \quad c_{\mathcal{C}} = \frac{1}{8} \cdot \frac{(d'_i + 2)!! \ (d''_i + 2)!! \ (d_i + 1)!!}{(d'_i + 1)!! \ (d''_i + 1)!! \ d_i!!} \\ \cdot \frac{(k_1 - 2)! \ (k_2 - 2)!}{(k - 4)!} \cdot \begin{cases} 1 & \text{when both } d'_i, d''_i \text{ are odd} \\ \frac{4}{\pi^2} & \text{otherwise.} \end{cases}$$

Proof of (4.26). - Let

$$\alpha'_a := \{d'_i, d_{i_1}, \dots d_{i_{k_1}}\} \qquad \alpha'_b := \{d''_i, d_{j_1}, \dots d_{j_{k_2}}\} \qquad \alpha'_{\mathcal{C}} := \alpha'_a \sqcup \alpha'_b.$$

Contracting a saddle connection of type II and detaching the resulting singular flat surface into two components we get a disconnected flat surface $S' = S'_a \sqcup S'_b$, where $S' \in \mathcal{Q}(\alpha'_{\mathcal{C}})$. The stratum of disconnected surfaces $\mathcal{Q}(\alpha'_{\mathcal{C}})$ is the principal boundary for configuration II. By (4.20)

$$\operatorname{Vol}\left(\mathcal{Q}_{1}^{\varepsilon}(\mathcal{C})\right) = (\operatorname{explicit factor}) \cdot \pi \varepsilon^{2} \cdot \operatorname{Vol} \mathcal{Q}_{1}(\alpha_{\mathcal{C}}') + o(\varepsilon^{2}).$$

By (4.9) we have

$$\operatorname{Vol} \mathcal{Q}_{1}(\alpha_{\mathscr{C}}') = \frac{1}{2} \cdot \frac{(\operatorname{dim}_{\mathbb{C}} \mathcal{Q}(\alpha_{a}') - 1)!(\operatorname{dim}_{\mathbb{C}} \mathcal{Q}(\alpha_{b}') - 1)!}{(\operatorname{dim}_{\mathbb{C}} \mathcal{Q}(\alpha_{\mathscr{C}}') - 1)!} \cdot \operatorname{Vol} \mathcal{Q}_{1}(\alpha_{a}') \cdot \operatorname{Vol} \mathcal{Q}_{1}(\alpha_{b}').$$

Note that by definition $\dim_{\mathbb{C}} \mathcal{Q}(\alpha'_{\mathcal{C}}) = \dim_{\mathbb{C}} \mathcal{Q}(\alpha'_a) + \dim_{\mathbb{C}} \mathcal{Q}(\alpha'_a)$. Hence

$$\dim_{\mathbb{C}} \mathcal{Q}(\alpha'_{\mathcal{C}}) = ((k_1+1)-2) + ((k_2+1)-2) = (k_1+k_2) - 2 = k - 3 = \dim_{\mathbb{C}} \mathcal{Q}(\alpha) - 2.$$

The "(explicit factor) $\cdot \pi \varepsilon^2$ " in Formula (4.20) stands for the measure of the space of parameters of deformation corresponding to holonomy vectors in D_{ε}^2/\pm . For configuration \mathscr{C} of type II this measure was computed in (4.23). Thus, we can rewrite (4.20) as

$$\operatorname{Vol}\left(\mathcal{Q}_{1}^{\varepsilon}(\mathcal{C})\right) = \frac{(d'+2)(d''+2)}{2} \cdot \pi \varepsilon^{2} \cdot \frac{1}{2} \cdot \frac{(\dim_{\mathbb{C}} \mathcal{Q}(d'_{i}, d_{i_{1}}, \dots, d_{i_{k_{1}}}) - 1)! (\dim_{\mathbb{C}} \mathcal{Q}(d''_{i}, d_{j_{1}}, \dots, d_{j_{k_{2}}}) - 1)!}{(\dim_{\mathbb{C}} \mathcal{Q}(d_{1}, d_{2}, \dots, d_{k}) - 2)!} \cdot \operatorname{Vol} \mathcal{Q}_{1}(d''_{i}, d_{j_{1}}, \dots, d_{j_{k_{2}}}).$$

Applying (4.18) and (4.19) to the above expression we obtain (4.26).

4° SÉRIE – TOME 49 – 2016 – Nº 6
4.11. A "pocket", i.e., a cylinder bounded by a pair of poles

Consider a configuration \mathcal{C} of type III where we have a single cylinder filled with closed regular geodesics, such that the cylinder is bounded by a saddle connection joining a fixed pair of simple poles P_{j_1}, P_{j_2} on one side and by a separatrix loop emitted from a fixed zero P_i of order $d_i \ge 1$ on the other side. This information is considered to be part of the configuration. By convention, the affine holonomy associated to this configuration corresponds to the closed geodesic and *not* to the saddle connection joining the two simple poles. (Such a saddle connection is twice as short as the closed geodesic.)

THEOREM 4.5. – The Siegel-Veech constant $c_{\mathcal{C}}$ for this configuration is expressed as follows:

(4.28)
$$c = \frac{d_i}{2(\dim_{\mathbb{C}} \mathcal{Q}(d_1, \dots, d_k) - 2)} \cdot \frac{\operatorname{Vol} \mathcal{Q}_1(d_1, d_2, \dots, d_i - 2, \dots, d_k)}{\operatorname{Vol} \mathcal{Q}_1(d_1, \dots, d_i, \dots, d_k)}$$

After plugging in Theorem 1.1, we get

(4.29)
$$c_{\mathcal{C}} = \frac{d_i + 1}{2(k-4)} \cdot \frac{1}{\pi^2}$$

Proof of (4.28). – Let $\alpha'_{\mathcal{C}} = \{d_1, \ldots, d_{i-1}, d_i - 2, d_{i+1}, \ldots, d_k\}$. Consider a configuration of type III with a short saddle connection γ joining a zero of degree d_i to itself. Contracting γ we get a flat surface S' in the principal boundary stratum $\mathcal{Q}(\alpha'_{\mathcal{C}})$.

To go backwards, we need to create a hole on S' with geodesic boundary having holonomy $\pm \vec{v}$ and attach a cylindrical "pocket" to this hole; see the right picture in Figure 9. The cone angle at the singularity P_i of degree $(d_i - 2)$ is $\pi \cdot d_i$. Thus, having a surface $S' \in Q_1(\alpha'_{\mathcal{C}})$ and a vector $\pm \vec{v} \in D_{\varepsilon}^2/\pm$ there are d_i rays in line direction $\pm \vec{v}$ adjacent to the singularity P_i .

Note, however, that now a deformation involves not only the surface S' from the principal boundary and a holonomy vector $\pm \vec{v}$, but also additional parameters describing the geometry of the "pocket". Geometrically, a "pocket" is equivalent to a flat cylinder endowed with a distinguished line direction and with a marked point on each of the boundary components. Thus, in addition to the holonomy vector $\pm \vec{v}$ representing the waist curve, it is parameterized by the *height h* of the cylinder and by the *twist t* of the cylinder, $0 \le t < |\vec{v}|$. Parameters h and t record the information about the holonomy along a saddle connection joining the zero P_i on one side of the cylinder to one of the simple poles, say, P_{j_1} on the other side of the cylinder. The flat area of a "pocket" $T(\pm \vec{v}, h, t)$ equals $|\vec{v}| \cdot h$.

The measure $d\mu$ in $Q^{\varepsilon, \text{thick}}(\alpha)$ disintegrates into the product measure $d\mu'$ on $Q(\alpha')$ and the measure $d\nu$ on the "space of pockets" \mathcal{R} ,

$$d\mu(S) = d\mu(S') \cdot d\nu(T).$$

The parameter $\pm \vec{v}$ corresponds to the holonomy along a *closed* saddle connection, while the parameters (h, t) correspond to holonomy along a saddle connection joining distinct singularities. Hence, the resulting measure on the space of parameters defining a "pocket" is

$$d\nu(T) = d\vec{v} \cdot 4dhdt.$$

Following Convention 4.1 we denote by \mathscr{R}_1 the hypersurface of pockets of area $\frac{1}{2}$. Let $S \in \mathscr{Q}_1(\alpha)$. We denote by $rS \in \mathscr{Q}(\alpha)$ the surface proportional to the initial one with the

coefficient r; in particular, $area(rS) = r^2/2$, see Convention 4.1. We use similar notations $r_S S'$ and $r_T T$ for surfaces from $Q(\alpha')$ and from \mathcal{R} correspondingly. We recall that the volume elements in the strata and the area elements on the corresponding "unit hyperboloids" are related as follows, see (4.2):

$$d\mu = r^{2n-1} dr \ d\mu_1, \quad \text{where } n = \dim_{\mathbb{C}} \mathcal{Q}(\alpha) = 2(k-2)$$
$$d\mu = r_S^{2n_S-1} dr_S \ d\mu'_1, \text{ where } n_S = \dim_{\mathbb{C}} \mathcal{Q}(\alpha') = 2(k-4)$$
$$d\nu = r_T^{2n_T-1} dr_T \ d\nu_1, \text{ where } n_T = \dim_{\mathbb{C}} \mathcal{R} = 2.$$

Let $S' \in Q_1(\alpha')$. Consider a surface $r_S S'$, where $0 < r_S \leq 1$; it has area $r_S^2/2$. Define $\Omega(\varepsilon, r_S) \subset \mathcal{R}$ to be the set of pockets, such that performing an appropriate surgery to $r_S S'$ and pasting in a "pocket" from $\Omega(\varepsilon, r_S)$ we get a surface $S \in C(Q_1^{\varepsilon}(\alpha))$. Ignoring a negligible change of the area of the surface $r_S S'$ after creating a hole, we get the following two constraints. The first constraint imposes the bound on the area $r_T^2/2$ of a pocket $r_T T$, where $T \in \mathcal{R}_1$: the total area of the compound surface S should be at most 1/2, so $r_S^2 + r_T^2 \leq 1$. The second constraint imposes a bound on the length of the waist curve of the cylinder: after rescaling proportionally the compound surface S to let it have area $\frac{1}{2}$ we should get a waist curve of length at most ε . Thus, the waist curve of the original cylinder should be at most $\varepsilon \sqrt{r_S^2 + r_T^2}$. Clearly, the set $\Omega(\varepsilon, r_S)$ does not depend on the particular surface $r_s S' \in Q(\alpha')$, but only on the parameters r_S and ε .

We have seen that there are d_i rays in line direction $\pm \vec{v}$ adjacent to the singularity P_i . Using the above notations we can represent the volume of a cone in $Q_1(\alpha)$ over $Q_1^{\varepsilon}(\alpha, \mathcal{C})$ (see (4.4) for the definition of a *cone*) as

(4.30)
$$\mu(C(\mathcal{Q}_1^{\varepsilon}(\alpha,\mathcal{C}))) = d_i \cdot \operatorname{Vol} \mathcal{Q}_1(\alpha') \cdot \int_0^1 \nu_T(\Omega(\varepsilon,r_S)) r_S^{2n_S-1} dr_S + o(\varepsilon^2).$$

Denote by $\text{Cusp}(\varepsilon)$ the volume of the ε -thin part of the "unit hyperboloid" in the space of "pockets":

$$\operatorname{Cusp}(\varepsilon) := \operatorname{Vol} \mathscr{R}_1^{\varepsilon}.$$

From the definition of the subset $\Omega(\varepsilon, r_S)$ it immediately follows that its volume is expressed by the following integral

(4.31)
$$\nu_T(\Omega(\varepsilon, r_S)) = \int_0^{\sqrt{1-r_S^2}} r_T^{2n_T-1} \cdot \operatorname{Cusp}\left(\frac{\varepsilon \cdot \sqrt{r_S^2 + r_T^2}}{r_T}\right) dr_T.$$

Thus, we need to evaluate the following integral

(4.32)
$$\mu(C(\mathcal{Q}^{\varepsilon}(\alpha, \mathcal{C}))) = d_i \cdot \operatorname{Vol}(\mathcal{Q}_1(\alpha')) \cdot \int_0^1 r_S^{2n_S - 1} dr_S \int_0^{\sqrt{1 - r_S^2}} r_T^{2n_T - 1} \cdot \operatorname{Cusp}\left(\frac{\varepsilon \cdot \sqrt{r_S^2 + r_T^2}}{r_T}\right) dr_T + o(\varepsilon^2).$$

Lемма 4.6. –

$$\operatorname{Cusp}(\varepsilon) = 2\pi\varepsilon^2.$$

Proof. – We first evaluate the volume $\nu(C(\mathscr{R}_1^{\varepsilon}))$ of the corresponding cone. Pockets belonging to this cone are described by the following conditions:

$$\begin{cases} h \cdot |\vec{v}| \le 1/2 \\ |\vec{v}| \le \varepsilon \cdot \sqrt{2h \cdot |\vec{v}|}. \end{cases}$$

Hence

$$\nu(C(\mathcal{R}_1^{\varepsilon})) = \int_{D_{\varepsilon}^2/\pm} d\vec{v} \int_{w/(2\varepsilon^2)}^{1/(2w)} 2dh \int_0^w 2dt = 4\pi \int_0^{\varepsilon} w\left(\frac{1}{2w} - \frac{w}{2\varepsilon^2}\right) w \, dw = \frac{\pi\varepsilon^2}{2},$$

where $w = |\vec{v}|$. It remains to apply (4.5):

$$\nu(C(\mathscr{R}_1^{\varepsilon})) = \dim_{\mathbb{R}} \mathscr{R} \cdot \operatorname{Vol}(\mathscr{R}_1^{\varepsilon})$$

and to note that $\dim_{\mathbb{R}} \mathcal{R} = 4$.

Having found the expression

$$\operatorname{Cusp}\left(\frac{\varepsilon \cdot \sqrt{r_S^2 + r_T^2}}{r_T}\right) = 2\pi\varepsilon^2 \cdot \frac{r_S^2 + r_T^2}{r_T^2}$$

we can rewrite the integral (4.32) as

(4.33) $\mu(C(\mathcal{Q}^{\varepsilon}(\alpha, \mathcal{C}))) = d_i \cdot \operatorname{Vol} \mathcal{Q}_1(\alpha') \cdot$

$$2\pi\varepsilon^2 \cdot \int_0^1 r_S^{2n_S-1} dr_S \int_0^{\sqrt{1-r_S^2}} r_T^{2n_T-1} \cdot \frac{r_S^2 + r_T^2}{r_T^2} dr_T + o(\varepsilon^2).$$

Taking into consideration that $n_T = \dim_{\mathbb{C}} \mathcal{R} = 2$ we compute the above integrals and get

(4.34)
$$\mu(C(\mathcal{Q}_1^{\varepsilon}(\alpha, \mathcal{C}))) = d_i \cdot \operatorname{Vol} \mathcal{Q}_1(\alpha') \cdot \frac{2\pi\varepsilon^2}{2n_S(2n_S + 4)} + o(\varepsilon^2).$$

It remains to note that

$$\operatorname{Vol} \mathcal{Q}_{1}^{\varepsilon}(\alpha, \mathcal{C}) = \frac{1}{\dim_{\mathbb{R}} \mathcal{Q}(\alpha)} \cdot \mu(C(\mathcal{Q}_{1}^{\varepsilon}(\alpha, \mathcal{C}))),$$

see (4.5), and that

$$\dim_{\mathbb{R}} \mathcal{Q}(\alpha) = 2 \dim_{\mathbb{C}} \mathcal{Q}(\alpha) = 2(\dim_{\mathbb{C}} \mathcal{Q}(\alpha') + 2) = 2n_S + 4$$

to get

(4.35)
$$\operatorname{Vol} \mathcal{Q}_{1}^{\varepsilon}(\alpha, \mathcal{C}) = \pi \varepsilon^{2} \cdot \frac{d_{i}}{\dim_{\mathbb{C}} \mathcal{Q}(\alpha) - 2} \cdot \operatorname{Vol} \mathcal{Q}_{1}(\alpha') + o(\varepsilon^{2}).$$

Applying (4.18) and (4.19) to the above expression we obtain (4.28).

The rest of the discussion in §4.11 also depends on Theorem 1.1. Consider a slightly more general configuration: as before we consider a fixed pair of simple poles P_{j_1}, P_{j_2} , but this time we do not specify which zero we have at the base of the cylinder. Clearly the corresponding Siegel-Veech constant $c_{j_1,j_2}^{\text{pocket}}$ is equal to the sum of the Siegel-Veech constants considered above over all zeroes P_i on our surface S:

$$c_{j_1,j_2}^{\text{pocket}} = \sum_{i=1 \mid d_i \ge 1}^k c_i.$$

ANNALES SCIENTIFIQUES DE L'ÉCOLE NORMALE SUPÉRIEURE

The following corollary follows immediately from the Formula (4.29) above.

COROLLARY 4.7. – For any stratum of meromorphic quadratic differentials with at most simple poles and with no marked points on \mathbb{CP}^1 and for every fixed pair P_{j_1} , P_{j_2} of simple poles, the Siegel-Veech constant $c_{j_1,j_2}^{\text{pocket}}$ is equal to

(4.36)
$$c_{j_1,j_2}^{\text{pocket}} = \frac{1}{2\pi^2}.$$

Proof. – By assumption the stratum $Q(d_1, \ldots, d_k)$ does not contain marked points. We can order d_i in the reverse lexicographic order, so that d_1, \ldots, d_m are positive (i.e., correspond to the zeroes) and d_{m+1}, \ldots, d_{m+n} are equal to -1 (i.e., correspond to the simple poles).

Since we live on \mathbb{CP}^1 we have $\sum_{i=1}^k d_i = -4$ which is equivalent to $\sum_{i=1}^m d_i = n-4$. Hence,

$$c_{j_1,j_2}^{\text{pocket}} = \frac{1}{2(k-4)} \sum_{i=1}^m (d_i+1) \frac{1}{\pi^2}$$
$$= \frac{1}{2(n+m-4)} \Big(\sum_{i=1}^m d_i + \sum_{i=1}^m 1 \Big) \frac{1}{\pi^2} = \frac{1}{2(n+m-4)} (n-4+m) \frac{1}{\pi^2} = \frac{1}{2\pi^2}. \square$$

Proof of Theorem 1.5. – Note that Theorem 1.5 counts the number of generalized diagonals joining two fixed corners of a right-angled billiard, while in the "pocket" configuration we count the number of closed flat geodesics on the induced cylinder, which are twice longer. Rescaling, we get an extra factor 4 for the counting problem in this alternative normalization. Applying Theorem 2.4, and taking into consideration the factor $\frac{1}{4}$ in Formula (2.1) we get the answer

$$N_{ij}(\Pi, L)$$
 "~" $\frac{1}{4} \cdot 4 \cdot c_{i,j}^{\text{pocket}} \frac{\pi L^2}{\text{area of the billiard table }\Pi}$.

Plugging in expression (4.36) for $c_{i,j}^{\text{pocket}}$ we get Formula (1.10).

4.12. A "dumbbell", i.e., a simple cylinder separating the sphere and joining a pair of distinct zeroes

Consider a configuration \mathscr{C} of type IV, where we have a single cylinder filled with closed regular geodesics, such that the cylinder is bounded by a separatrix loop on each side. We assume that the separatrix loop bounding the cylinder on one side is emitted from a fixed zero P_i of order $d_i \ge 1$ and that the separatrix loop bounding the cylinder on the other side is emitted from a fixed zero P_j of order $d_j \ge 1$.

Such a cylinder separates the original surface S in two parts; let $P_{i_1}, \ldots, P_{i_{k_1}}$ be the list of singularities (zeroes and simple poles) which get to the first part and $P_{j_1}, \ldots, P_{j_{k_2}}$ be the list of singularities (zeroes and simple poles) which get to the second part. In particular, we have $i \in \{i_1, \ldots, i_{k_1}\}$ and $j \in \{j_1, \ldots, j_{k_2}\}$. We assume that S does not have any marked points. Denoting as usual by d_n the order of the singularity P_n we can represent the sets with multiplicities $\alpha := \{d_1, \ldots, d_k\}$ as a disjoint union of the two subsets

 $\{d_1,\ldots,d_k\} = \{d_{i_1},\ldots,d_{i_{k_1}}\} \sqcup \{d_{j_1},\ldots,d_{j_{k_2}}\}.$

This information is considered to be part of the configuration.

THEOREM 4.8. – The Siegel-Veech constant $c_{\mathcal{C}}$ for this configuration is expressed as follows:

$$(4.37) \quad c_{\mathcal{C}} = \frac{d_i \cdot d_j}{4} \\ \cdot \frac{(\dim_{\mathbb{C}} \mathcal{Q}(d_{i_1}, \dots, d_i - 2, \dots, d_{i_{k_1}}) - 1)! (\dim_{\mathbb{C}} \mathcal{Q}(d_{j_1}, \dots, d_j - 2, \dots, d_{j_{k_2}}) - 1)!}{(\dim_{\mathbb{C}} \mathcal{Q}(d_1, d_2, \dots, d_k) - 2)!} \\ \cdot \frac{\operatorname{Vol} \mathcal{Q}_1(d_{i_1}, \dots, d_i - 2, \dots, d_{i_{k_1}}) \cdot \operatorname{Vol} \mathcal{Q}_1(d_{j_1}, \dots, d_j - 2, \dots, d_{j_{k_2}})}{\operatorname{Vol} \mathcal{Q}_1(d_1, \dots, d_k)}.$$

Plugging in Theorem 1.1 we get:

(4.38)
$$c_{\mathcal{C}} = \frac{(d_i+1)(d_j+1)}{2} \cdot \frac{(k_1-3)!(k_2-3)!}{(k-4)!} \cdot \frac{1}{\pi^2}.$$

Proof of (4.37). – The proof is completely analogous to computation of the Siegel-Veech constant for configuration III. Denote by α'_a the set with multiplicities obtained from $\{d_{i_1}, \ldots, d_{i_{k_1}}\}$ by replacing the entry d_i by $d_i - 2$. Similarly, denote by α'_b the set with multiplicities obtained from $\{d_{j_1}, \ldots, d_{j_{k_2}}\}$ by replacing the entry d_j by $d_j - 2$. Define $\alpha' := \alpha'_a \sqcup \alpha'_b$. Contracting the two saddle connections we get a disconnected flat surface S'in the principal boundary stratum $Q(\alpha')$.

Given a flat surface $S' \in Q(\alpha')$ and a holonomy vector $\pm \vec{v}$ we have d_i separatrix rays in direction $\pm \vec{v}$ adjacent to the point P_i of S' and d_j separatrix rays in direction $\pm \vec{v}$ adjacent to the point P_j .

Following line-by-line the proof of (4.28) in the previous section we get an expression for Vol $Q_1^{\varepsilon}(\alpha, \mathcal{C})$ completely analogous to (4.35): the only adjustment consists in replacing the factor d_i by the product $d_i d_j$:

$$\operatorname{Vol} \mathcal{Q}_1^{\varepsilon}(\alpha, \mathcal{C}) = \pi \varepsilon^2 \cdot \frac{d_i d_j}{\dim_{\mathbb{C}} \mathcal{Q}(\alpha) - 2} \cdot \operatorname{Vol} \mathcal{Q}_1(\alpha') + o(\varepsilon^2).$$

Applying expression (4.9) from §4.4 for Vol $Q_1(\alpha')$ and taking into consideration that $\dim_{\mathbb{C}} Q(\alpha') = \dim_{\mathbb{C}} Q(\alpha) - 2$ we can rewrite the latter expression as

$$\operatorname{Vol} \mathcal{Q}_{1}^{\varepsilon}(\alpha, \mathcal{C}) = \pi \varepsilon^{2} \cdot \frac{d_{i}d_{j}}{2} \\ \cdot \frac{(\dim_{\mathbb{C}} \mathcal{Q}(d_{i_{1}}, \dots, d_{i} - 2, \dots, d_{i_{k_{1}}}) - 1)!(\dim_{\mathbb{C}} \mathcal{Q}(d_{j_{1}}, \dots, d_{j} - 2, \dots, d_{j_{k_{2}}}) - 1)!}{(\dim_{\mathbb{C}} \mathcal{Q}(d_{1}, d_{2}, \dots, d_{k}) - 2)!} \\ \cdot \operatorname{Vol} \mathcal{Q}_{1}(d_{i_{1}}, \dots, d_{i} - 2, \dots, d_{i_{k_{1}}}) \cdot \operatorname{Vol} \mathcal{Q}_{1}(d_{j_{1}}, \dots, d_{j} - 2, \dots, d_{j_{k_{2}}}) + o(\varepsilon^{2}).$$

Applying (4.18) and (4.19) to the above expression we obtain (4.37).

4.13. Siegel-Veech constant $c_{\rm area}$

Consider an SL(2, \mathbb{R})-invariant manifold in a stratum of Abelian differentials or a PSL(2, \mathbb{R})-invariant manifold in a stratum of quadratic differentials. Denote by c_{cyl} the associated Siegel-Veech constant responsible for counting the maximal cylinders of closed

geodesics and denote by c_{area} the Siegel-Veech constant responsible for counting the cylinders of closed geodesics counted with weight

$$\frac{\text{(area of the cylinder)}}{\text{(area of the surface)}}$$

In [50] Ya. Vorobets proved the following result:

THEOREM (Vorobets, 2003). – For any connected component $\mathcal{H}^{comp}(\alpha)$ of any stratum of Abelian differentials the ratio c_{area}/c_{cyl} of the corresponding Siegel-Veech constants satisfies the following relation:

$$rac{c_{ ext{area}}}{c_{ ext{cyl}}} = rac{1}{2g - 2 + n} = rac{1}{\dim_{\mathbb{C}} \mathcal{H}(\alpha) - 1}$$

Note that a configuration of \hat{h} omologous saddle connections of \mathbb{CP}^1 involves at most one cylinder. The following proposition states the Vorobets formula for individual configurations involving cylinders for strata of meromorphic quadratic differentials with simple poles on \mathbb{CP}^1 .

PROPOSITION 4.9. – For any stratum $Q_1(d_1, \ldots, d_n)$ of meromorphic quadratic differentials with simple poles on \mathbb{CP}^1 and for any admissible configuration \mathcal{C} of saddle connections involving a cylinder the following equality holds:

$$\frac{c_{\text{area}}(\mathcal{C})}{c(\mathcal{C})} = \frac{1}{\dim_{\mathbb{C}} \mathcal{Q}(d_1, \dots, d_n) - 1} = \frac{1}{n-3}.$$

Proof. – The proof consists in an elementary adjustment of the computation from the previous two sections. We will present the computation of $c_{\text{area}}(\mathcal{C})$ for the "pocket configuration" (configuration of type III) following the analogous computation in §4.11. The computation for the configuration of type IV is completely analogous and is omitted.

This time we have to compute the integral of the ratio $\frac{r_T^2}{r_S^2 + r_T^2}$ of the area $r_T^2/2$ of the cylinder over the total area $(r_S^2 + r_T^2)/2$ of the entire surface. We integrate this expression over $Q_1^{\varepsilon}(\alpha, \mathcal{C})$. Note that this ratio is the same for proportional surfaces. Thus we can integrate with respect to the corresponding cone $C(Q_1^{\varepsilon}(\alpha, \mathcal{C}))$:

$$\int_{\mathcal{Q}_1^\varepsilon(\alpha,\mathcal{C})} \frac{r_T^2}{r_S^2 + r_T^2} \ d\mu_1 = \dim_{\mathbb{R}} \mathcal{Q}(\alpha) \cdot \int_{C(\mathcal{Q}_1^\varepsilon(\alpha,\mathcal{C}))} \frac{r_T^2}{r_S^2 + r_T^2} \ d\mu(S).$$

Moreover, the ratio of the corresponding Siegel-Veech constants satisfies

(4.39)
$$\frac{c_{\text{area}}(\alpha, \mathcal{C})}{c_{\text{cyl}}(\alpha, \mathcal{C})} = \lim_{\varepsilon \to 0} \frac{\int_{C(\mathcal{Q}_{1}^{\varepsilon}(\alpha, \mathcal{C}))} \frac{r_{T}}{r_{S}^{2} + r_{T}^{2}} d\mu(S)}{\int_{C(\mathcal{Q}_{1}^{\varepsilon}(\alpha, \mathcal{C}))} d\mu(S)}$$

The denominator

$$\int_{C(\mathcal{Q}_1^\varepsilon(\alpha,\mathcal{C}))} d\mu(S) = \mu(C(\mathcal{Q}_1^\varepsilon(\alpha,\mathcal{C})))$$

of the above ratio is given by (4.33). To evaluate the integral in the numerator we modify (4.33) by multiplying the function inside the integral by an extra factor $r_T^2/(r_S^2 + r_T^2)$ obtaining:

(4.40)
$$\int_{C(Q_{1}^{\varepsilon}(\alpha,\mathcal{C}))} \frac{r_{T}^{2}}{r_{S}^{2} + r_{T}^{2}} d\mu(S) = d_{i} \cdot \operatorname{Vol}(Q_{1}(\alpha')) \cdot 2\pi\varepsilon^{2} \cdot \int_{0}^{1} r_{S}^{2n_{S}-1} dr_{S} \int_{0}^{\sqrt{1-r_{S}^{2}}} r_{T}^{2n_{T}-1} dr_{T} + o(\varepsilon^{2}).$$

Taking into consideration that $n_T = \dim_{\mathbb{C}} \mathscr{R} = 2$ and evaluating the latter integral we obtain

(4.41)
$$\int_{C(\mathcal{Q}_1^{\varepsilon}(\alpha,\mathcal{C}))} \frac{r_T^2}{r_S^2 + r_T^2} \ d\mu(S) = d_i \cdot \operatorname{Vol} \mathcal{Q}_1(\alpha') \cdot \frac{4\pi\varepsilon^2}{2n_S(2n_S + 2)(2n_S + 4)} + o(\varepsilon^2).$$

Plugging (4.41) and (4.34) into expression (4.39) and recalling the definition

$$n_S = \dim_{\mathbb{C}} \mathcal{Q}(\alpha') = \dim_{\mathbb{C}} \mathcal{Q}(\alpha) - 2$$

we obtain

$$\frac{c_{\text{area}}(\alpha, \mathcal{C})}{c_{\text{cyl}}(\alpha, \mathcal{C})} = \frac{2}{2n_S + 2} = \frac{1}{\dim_{\mathbb{C}} \mathcal{Q}(\alpha) - 1},$$

which completes the proof of Proposition 4.9.

1351

Proposition 4.9 immediately implies the following statement.

COROLLARY 4.10. – For any stratum $Q_1(d_1, \ldots, d_n)$ of meromorphic quadratic differentials with simple poles on \mathbb{CP}^1 the Siegel-Veech constant c_{area} is expressed in terms of the Siegel-Veech constants of configurations as follows:

$$c_{ ext{area}} = rac{1}{n-3} \cdot \sum_{\substack{ ext{configurations } \mathscr{C} \\ ext{containing a cylinder}}} c_{\mathscr{C}}.$$

5. Computation of the volumes of the moduli spaces

In this section, we prove Theorem 1.1. The approach taken here is somewhat indirect.

5.1. An identity for the Siegel-Veech constant

The idea is to prove Formula (1.1) in Theorem 1.1 for the volume by induction, using the formulas expressing Siegel-Veech constants in terms of the volumes. Namely, by [16, Theorem 3] one has:

$$c_{\text{area}}(\mathcal{Q}(d_1,\ldots,d_k)) = -\frac{1}{8\pi^2} \sum_{j=1}^k \frac{d_j(d_j+4)}{d_j+2}.$$

On the other hand, by the Vorobets formula applied to \mathbb{CP}^1 (see Corollary 4.10 in §4.13) one has

$$c_{ ext{area}}(\mathcal{Q}(d_1,\ldots,d_k)) = rac{1}{\dim_{\mathbb{C}}\mathcal{Q}(d_1,\ldots,d_k) - 1} \cdot \sum_{\substack{ ext{Configurations } \mathcal{C}\\ ext{containing a cylinder}}} c_{\mathcal{C}}$$

In view of §2.1, for \mathbb{CP}^1 there are exactly two configurations containing a cylinder: a "pocket" and a "dumbbell". The formulas for the Siegel-Veech constants were given in §4.

Taking into consideration that $\dim_{\mathbb{C}} \mathcal{Q}(d_1, \ldots, d_k) - 1 = k - 3$, for any collection d_1, \ldots, d_k of integers in $\{-1, 1, 2, 3\}$ satisfying the relation

$$\sum_{i=1}^{k} d_i = -4$$

we get the following identity:

(5.1)
$$-\frac{1}{8\pi^2} \sum_{j=1}^k \frac{d_j(d_j+4)}{d_j+2} = \frac{1}{k-3} \cdot \left(\sum_{\substack{\text{"pocket"} \\ \text{configurations}}} c_{\mathcal{C}} + \sum_{\substack{\text{"dumbbell"} \\ \text{configurations}}} c_{\mathcal{C}} \right).$$

If we plug in the expressions (4.28) and (4.37) into (5.1), we get a formula of the form:

(5.2) Vol $Q_1(d_1, \ldots, d_n)$ = Explicit polynomial in volumes of simpler strata.

The formulas (5.2) clearly determine the volumes. Thus, to prove Theorem 1.1, it is enough to show that the expressions for the volumes given by Theorem 1.1 satisfy the recurrence relation (5.2), or equivalently to prove the following:

THEOREM 5.1. – *The explicit expressions* (4.29) *and* (4.38) *for the Siegel-Veech constants satisfy the identity* (5.1).

The proof of the Theorem 5.1 is quite involved and is done in Appendix A. This completes the proof of Theorem 1.1.

6. Counting trajectories and ergodic theory on moduli space

In this section we will prove Theorem 2.4. We modify appropriately the strategy from §4.5 to obtain the asymptotic formula (2.1). The key tool is Theorem C.1 proved by Jon Chaika in Appendix C.

6.1. Pointwise asymptotics

To understand the asymptotics for any set of special trajectories for the flat metric associated to $q \in Q_1$, we use (4.15) to reduce the problem to understanding

(6.1)
$$\frac{1}{2\pi} \int_0^{2\pi} \widehat{f}(g_t r_\theta q) d\theta,$$

where \hat{f} is the indicator function of the trapezoid defined in §4.5.1. We are particularly interested in the differentials q_{Π} , $\Pi \in \mathcal{B}$. If $\hat{f} \in \mathcal{L}_c$ (in the notation of Appendix C), we could directly apply Theorem C.1 to conclude that

$$\lim_{t \to \infty} \frac{1}{2\pi} \int_0^{2\pi} \widehat{f}(g_t r_\theta q_\Pi) d\theta = \frac{3}{4} b_{\mathscr{C}}(\mathscr{Q})$$

for almost every $\Pi \in \mathcal{B}$. Following an argument from [20], we will approximate \hat{f} by such functions. Fix $\varepsilon > 0$, let $h_{\varepsilon} : \mathcal{Q}_1 \to \mathbb{R}$ be a continuous function with

(6.2)
$$h_{\varepsilon}(q) = \begin{cases} 1 & l(q) > \varepsilon, \\ 0 & l(q) < \varepsilon/2 \end{cases}$$

1353

Here, l(q) denotes the length of the shortest saddle connection on q. The function h_{ε} is a smoothed version of the indicator function of the compact part of the stratum Q_1 . Given $\phi : Q_1 \to \mathbb{R}$, define

(6.3)
$$(A_t\phi)(q) = \frac{1}{2\pi} \int_0^{2\pi} \phi(g_t r_\theta q) d\theta$$

For any $q \in Q_1$,

(6.4)
$$\left(A_t(\widehat{f}h_{\varepsilon})\right)(q) \le \left(A_t\widehat{f}\right)(q) = \left(A_t(\widehat{f}h_{\varepsilon})\right)(q) + \left(A_t(\widehat{f}(1-h_{\varepsilon}))\right)(q) + \left(A_t(\widehat{f}(1$$

We follow [20, p.435, proof of Theorem 2.4]. Fix $1 > \eta > \delta > 0$. [19, Theorem 5.1] shows there is a $C(\delta)$ so that for all $q \in Q$

(6.5)
$$\widehat{f}(q) \le \frac{C(\delta)}{l(q)^{1+\delta}}.$$

On the other hand, $1 - h_{\varepsilon}(q) > 0$ implies $l(q) \leq \varepsilon$, so

$$\widehat{f}(q)(1-h_{\varepsilon}(q)) \leq \widehat{f}(q) \leq \frac{C(\delta)}{l(q)^{1+\eta}} \cdot l(q)^{\eta-\delta} \leq \varepsilon^{\eta-\delta} \frac{C(\delta)}{l(q)^{1+\eta}}.$$

Thus,

$$\left(A_t(\widehat{f}(1-h_{\varepsilon}))\right)(q) \le C(\delta)\varepsilon^{\eta-\delta}\left(A_tl^{-1-\eta}\right)(q).$$

[19, Theorem 5.2] states that for $\eta < 1$, there is a $C_1 = C_1(\eta, \Pi)$ so that for all t > 0,

$$\left(A_t l^{-1-\eta}\right)(q_{\Pi}) < C_1(\eta, \Pi).$$

Since $\hat{f}h_{\varepsilon}$ is continuous and compactly supported, for any Π from the set of full measure to which Theorem C.1 applies we get

$$\lim_{t\to\infty} \left(A_t(\widehat{f}h_{\varepsilon}) \right)(q_{\Pi}) = \frac{1}{\mu_1(\mathcal{Q}_1)} \int_{\mathcal{Q}_1} \widehat{f}h_{\varepsilon}(q) \, d\mu_1(q).$$

So we have

(6.6)
$$\liminf_{t \to \infty} A_t \widehat{f}(q_{\Pi}) \ge \frac{1}{\mu_1(\mathcal{Q}_1)} \int_{\mathcal{Q}_1} \widehat{f}h_{\varepsilon}(q) \, d\mu_1(q)$$

and

(6.7)
$$\limsup_{t \to \infty} A_t \widehat{f}(q_{\Pi}) \le \frac{1}{\mu_1(\mathcal{Q}_1)} \int_{\mathcal{Q}_1} \widehat{f}h_{\varepsilon}(q) \, d\mu_1(q) \, + \, C(\delta)C_1(\eta, \Pi)\varepsilon^{\eta-\delta}.$$

Combining (6.6) and (6.7) and letting $\varepsilon \to 0$, we obtain, as desired, Theorem 2.4.

We complete this section with the proof of Theorem 1.9 from §1.3.

Proof of Theorem 1.9. – The proof is completely analogous to the proof of Theorem 2.4 above; we just have to carefully follow the normalization which is different from the previous case.

By [16, Theorem 3] one has:

(6.8)
$$c_{\text{area}}(\mathcal{Q}(k_1-2,\ldots,k_n-2)) = -\frac{1}{8\pi^2} \sum_{j=1}^n \frac{(k_j-2)(k_j+2)}{k_j} = \frac{1}{8\pi^2} \sum_{j=1}^n \left(\frac{4}{k_j} - k_j\right)$$

The length of a closed trajectory in Π is the same as the length of the associated closed geodesic on the covering flat sphere S. By definition, the area of the band of closed trajectories on Π is the same as the area of each of the maximal cylinders on \mathbb{CP}^1 . However, the flat area of S is twice the area of Π . Thus, the ratio

(area of the band of periodic trajectories on Π)

(area of Π)

is twice larger then the corresponding ratio

(area of the maximal cylinder of periodic geodesics on S)

Taking into consideration that the bands of closed trajectories in the polygon Π are in the natural one-to-two correspondence with the maximal cylinders of closed regular geodesics on the covering flat sphere S, see Figure 6, we get

$$N_{\text{area}}(\Pi, L) = N_{\text{area}}(S, L).$$

It remains to note that

$$N_{\text{area}}(S,L) \sim c_{\text{area}}(\mathcal{Q}) \cdot \frac{L^2}{\text{area}(S)} = \frac{c_{\text{area}}(\mathcal{Q})}{2} \cdot \frac{L^2}{\text{area}(\Pi)}$$

to conclude that the constant in the weak quadratic asymptotic (1.12) in Theorem 1.9 is one half of the Siegel-Veech constant $c_{area}(Q)$ from (6.8).

Appendix A Proof of combinatorial identity

In this appendix, we prove Theorem 5.1. Our proof follows the following scheme. In Section A.1 we rewrite the conjectural identity (5.1) in a more detailed form (A.1) and then applying elementary algebraic manipulations we rewrite it again in the form (A.2). In Section A.2 we rearrange the summation in (A.2) and in Section A.3 we introduce multiindex notation. Combining this rearrangement with new notation we rewrite the conjectural identity in the form (A.3).

In Section A.4 we introduce generating functions $F(\mathbf{s})$ and $G(\mathbf{s})$ as power series in (multi)variable \mathbf{s} with coefficients involved in the conjectural combinatorial identity (A.3). The desired combinatorial identity (A.3) now wraps to the identity $F^2(\mathbf{s}) \stackrel{?}{=} G(\mathbf{s})$. At this stage we have just gained a more concise and convenient form for the conjectural identity, nothing serious has happened.

In Section A.5 we introduce an entire collection of auxiliary generating functions $M_a(s)$ indexed by a positive integer *a* (and depending on an integer multiindex parameter **b**):

$$M_a(\mathbf{s}) := \sum_{\mathbf{k} \in (\mathbb{Z}^{\ge 0})^m} A(a; \mathbf{b}; \mathbf{k}) \mathbf{s}^{\mathbf{k}},$$

where the *Mohanty coefficient* $A(a; \mathbf{b}; \mathbf{k})$ is defined in Section A.5. By a theorem of Mohanty [42], *all* these generation functions $M_a(\mathbf{s})$ are expressed in terms of $M_1(\mathbf{s})$ denoted by $z(\mathbf{s}) := M_1(\mathbf{s})$. Moreover, *all* these generation functions are expressed in terms of $z(\mathbf{s})$ in a very simple way, namely,

$$M_a(\mathbf{s}) = z^a(\mathbf{s}).$$

4° SÉRIE – TOME 49 – 2016 – Nº 6

By the same theorem of Mohanty, the basic generating function z(s) satisfies the functional relation

(*)
$$1 - z + \sum_{i=1}^{m} s_i z^{b_i} = 0.$$

Note that this relation is *polynomial* in the basic generating function z and in formal variables $\mathbf{s} = (s_1, \ldots, s_m)$.

The strategy of the proof is to express our generating functions F(s) and G(s) as polynomials in Mohanty functions $M_a(s) = z^a(s)$ and formal variables s. As soon as we get the corresponding expressions for F (Lemma A.3 in Section A.6) and G (Lemma A.4 in Section A.7) we express the difference $G - F^2$ as a polynomial in z and formal variables s and show (Theorem A.2) that in the resulting polynomial one can factor out the square of expression (*). Since by Mohanty's Theorem this expression is identically zero, this proves that $G - F^2$ is identically zero, which completes the proof of Theorem 5.1.

A.1. General identity to prove

Let d_1, \ldots, d_m be the degrees of zeroes only. The number *n* of simple poles is expressed as $n = 4 + \sum_{i=1}^{m} d_i$. The total number k = m + n of all singularities is, thus, expressed as $k = 4 + \sum_{i=1}^{m} (d_i + 1)$.

Recall that all zeroes and poles are *named*. A "pocket" configuration is uniquely defined by a choice of a distinguished zero (at the base of the cylinder) and by a choice of an unordered pair of simple poles (corners of the "pocket"); all choices of a zero and of a pair of poles are admissible. When the distinguished zero at the base of the cylinder has degree d_i , Formula (4.29) gives the following value for the Siegel-Veech constant $c_{\mathcal{C}}$ for an individual "pocket" configuration (with distinguished zero and distinguished pair of fixed poles):

$$c_{\mathcal{C}} = \frac{d_i + 1}{2\left(\sum_{i=1}^{m} (d_i + 1)\right)} \cdot \frac{1}{\pi^2},$$

where we have replaced k-4 in the denominator of Formula (4.29) by $k-4 = \sum_{i=1}^{m} (d_i+1)$. For each choice of the zero in the "pocket" configuration there are

$$\binom{n}{2} = \binom{4 + \sum_{i=1}^{m} d_i}{2}$$

ways to choose a pair of distinguished poles. Hence, the total impact of all "pocket" configurations to the right-hand-side of (5.1) (based on Formula (4.29)) can be written as

$$\begin{aligned} \frac{1}{k-3} \cdot \sum_{\substack{\text{"pocket"}\\\text{configurations}}} c_{\mathcal{C}} \\ &= \frac{1}{(1+\sum_{i=1}^{m} (d_{i}+1))} \cdot \begin{pmatrix} 4+\sum_{i=1}^{m} d_{i} \\ 2 \end{pmatrix} \sum_{i=1}^{m} \frac{d_{i}+1}{2(\sum_{i=1}^{m} (d_{i}+1))} \cdot \frac{1}{\pi^{2}} \\ &= \frac{1}{2\pi^{2}} \cdot \frac{1}{(1+\sum_{i=1}^{m} (d_{i}+1))} \cdot \begin{pmatrix} 4+\sum_{i=1}^{m} d_{i} \\ 2 \end{pmatrix}. \end{aligned}$$

A "dumbbell" configuration \mathscr{C} is uniquely defined by a choice of the following data. We need to choose zeroes go to one part of the "dumbbell"; all the remaining zeroes go to the complementary part. In other words, we have to consider all partitions of the set $\{1, \ldots, m\}$ enumerating the zeroes into two nonempty complementary subsets $\{i_1, \ldots, i_{m_1}\} \sqcup \{j_1, \ldots, j_{m_2}\}$. For each such partition we have to consider all possible choices of a distinguished zero (at the base of the cylinder) in each of the two groups. After that, we have to choose $n_1 = 2 + \sum_{i=1}^{m_1} d_i$ out of $n = 4 + \sum_{i=1}^{m} d_i$ simple poles which go to the first part of the "dumbbell"; the remaining simple poles go to the other part. When all these data are chosen and when the distinguished two zeros (one in each of the two groups) at the base of the cylinder have degrees d_i, d_j , Formula (4.38) gives the following value for the Siegel-Veech constant $c_{\mathcal{C}}$ for the individual "dumbbell" configuration:

$$c_{\mathcal{C}} = \frac{(d_i+1)(d_j+1)}{2} \cdot \frac{(-1+\sum_{i=1}^{m_1}(d_i+1))! \left(-1+\sum_{j=1}^{m_2}(d_j+1)\right)!}{(\sum_{i=1}^m (d_i+1))!} \cdot \frac{1}{\pi^2}.$$

Here we have replaced k in the denominator of (4.38) by $k = 4 + \sum_{i=1}^{m} (d_i + 1)$, and have replaced k_1 and k_2 in the numerator of (4.38) by $k_1 = m_1 + n_1 = 2 + \sum_{i=1}^{m_1} (d_i + 1)$ and by $k_2 = m_2 + n_2 = 2 + \sum_{j=1}^{m_2} (d_j + 1)$ correspondingly.

For each partition of the set $\{1, \ldots, m\}$ enumerating the zeroes into two nonempty subsets $\{i_1, \ldots, i_{m_1}\} \sqcup \{j_1, \ldots, j_{m_2}\}$ (which makes part of the "dumbbell" configuration) there are

$$\frac{\left(4 + \sum_{i=1}^{m} d_i\right)!}{\left(2 + \sum_{i=1}^{m_1} d_i\right)!\left(2 + \sum_{j=1}^{m_2} d_j\right)!}$$

ways to partition the simple poles between two parts of the "dumbbell". Taking into consideration this counting and plugging in the explicit conjectural expressions (4.29) and (4.38) for the Siegel-Veech constants $c_{\mathcal{C}}$ into (5.1) we observe that the right-hand-side of (5.1) can be read as

$$\begin{aligned} \frac{1}{k-3} \cdot \left(\sum_{\substack{\text{"pocket"}\\\text{configurations}}} c_{\mathcal{C}} + \sum_{\substack{\text{"dumbbell"}\\\text{configurations}}} c_{\mathcal{C}} \right) \stackrel{?}{=} \\ & = \frac{1}{2\pi^2} \cdot \frac{1}{(1+\sum_{i=1}^m (d_i+1))} \cdot \left(\left(\frac{4+\sum_{i=1}^m d_i}{2} \right) \right. \right. \\ & + \sum_{1 \le i < j \le m} (d_i+1)(d_j+1) \sum_{\substack{\text{partitions of } \{1, \dots, m\} \text{ into}\\ \{i_1, \dots, i_{m_1}\} \sqcup \{j_1, \dots, j_{m_2}\} \\ \text{ such that } i \text{ is in the second subset}} \\ & \cdot \frac{\left(-1 + \sum_{i=1}^m (d_i+1) \right)! \left(-1 + \sum_{j=1}^m (d_j+1) \right)!}{\left(\sum_{i=1}^m (d_i+1) \right)!} \right) \end{aligned}$$

Multiplying both parts of the conjectural identity by the common factor

$$4\pi^2 \cdot \left(1 + \sum_{i=1}^m (d_i + 1)\right)$$

e binomial coefficient $\begin{pmatrix} 4 + \sum_{i=1}^m d_i \\ 2 \end{pmatrix}$ coming from the "pocket" configuration
hand-side of the identity and simplifying the resulting expressions we get the

moving the

to the lefthe following conjectural identity:

$$(A.1) \quad \left(6 + \sum_{i=1}^{m} \frac{d_i(d_i+1)}{d_i+2}\right) \cdot \left(1 + \sum_{i=1}^{m} (d_i+1)\right) - \left(4 + \sum_{i=1}^{m} d_i\right) \left(3 + \sum_{i=1}^{m} d_i\right)$$
$$\stackrel{?}{=} 2 \cdot \frac{\left(4 + \sum_{i=1}^{m} d_i\right)!}{\left(\sum_{i=1}^{m} (d_i+1)\right)!} \cdot \sum_{1 \le i < j \le m} (d_i+1)(d_j+1)$$
$$\cdot \sum_{\substack{partitions of \{1, \dots, m\} \text{ into} \\ \{r_1, \dots, r_{m_1}\} \sqcup \{s_1, \dots, s_{m_2}\} \\ \text{such that } i \text{ is in the first subset} \\ \text{and } j \text{ is in the serond subset}}} \frac{\left(-1 + \sum_{i=1}^{m} (d_{r_i}+1)\right)! \cdot \left(-1 + \sum_{j=1}^{m_2} (d_{s_j}+1)\right)!}{\left(2 + \sum_{i=1}^{m_1} d_{r_i}\right)! \cdot \left(2 + \sum_{j=1}^{m_2} d_{s_j}\right)!}$$

This is the identity which we need to prove.

Changing the order of the summation we can first sum over all possible partitions of the set of indices $\{1, \ldots, m\}$ and having chosen the partition we consider all possible ways to select a distinguished element *i* in the first subset and a distinguished element *j* in the second subset. Note, however, that we will see each of the elements of the above sum twice. Thus, collecting the resulting sums we can rewrite the sum in the right-hand-side of the above

expression as follows:

$$\sum_{\substack{\text{partitions of } \{1,...,m\} \text{ into} \\ \text{two nonempty sets} \\ \{r_1,...,r_{m_1}\} \sqcup \{s_1,...,s_{m_2}\}}} \frac{\left(-1 + \sum_{i=1}^{m_1} (d_{r_i} + 1)\right)! \cdot \left(-1 + \sum_{j=1}^{m_2} (d_{s_j} + 1)\right)!}{\left(2 + \sum_{i=1}^{m_1} d_{r_i}\right)! \cdot \left(2 + \sum_{j=1}^{m_2} d_{s_j}\right)!} \\ \cdot \left(\sum_{1 \le i \le m_1} (d_{r_i} + 1)\right) \left(\sum_{1 \le j \le m_2} (d_{s_j} + 1)\right) \\ = \sum_{\substack{\text{partitions of } \{1,...,m\} \text{ into} \\ \text{two nonempty sets} \\ \{r_1,...,r_{m_1}\} \sqcup \{s_1,...,s_{m_2}\}}} \frac{\left(\sum_{1 \le j \le m_2} (d_{s_j} + 1)\right)}{\left(2 + \sum_{i=1}^{m_1} d_{r_i}\right)! \cdot \left(\sum_{j=1}^{m_2} (d_{s_j} + 1)\right)!}\right)}$$

Hence, we can rewrite the right-hand-side in (A.1) as a sum over the ratios of binomial coefficients:

$$\frac{\left(4 + \sum_{i=1}^{m} d_{i}\right)!}{\left(\sum_{i=1}^{m} (d_{i}+1)\right)!} \cdot \sum_{\substack{\text{partitions of } \{1, \dots, m\} \text{ into} \\ \text{two nonempty sets} \\ \{r_{1}, \dots, r_{m_{1}}\} \sqcup \{s_{1}, \dots, s_{m_{2}}\}} \frac{\left(\sum_{i=1}^{m_{1}} (d_{r_{i}}+1)\right)! \cdot \left(\sum_{j=1}^{m_{2}} (d_{s_{j}}+1)\right)!}{\left(2 + \sum_{i=1}^{m_{1}} d_{r_{i}}\right)! \cdot \left(2 + \sum_{j=1}^{m_{2}} d_{s_{j}}\right)!}\right)}$$
$$= \sum_{\substack{\text{partitions of } \{1, \dots, m\} \text{ into} \\ \text{two nonempty sets} \\ \{r_{1}, \dots, r_{m_{1}}\} \sqcup \{s_{1}, \dots, s_{m_{2}}\}}} \frac{\left(4 + \sum_{i=1}^{m} d_{i}\right)}{\left(\sum_{i=1}^{m} (d_{i}+1)\right)}\right)}.$$

Finally, omitting the conditions that the subsets of the partition are nonempty, we get two extra terms. It is immediate to verify that their sum is equal to

$$\left(4 + \sum_{i=1}^{m} d_i\right) \left(3 + \sum_{i=1}^{m} d_i\right)$$

and, thus, we can rewrite the needed conjectural identity (A.1) as follows:

(A.2)
$$\left(6 + \sum_{i=1}^{m} \frac{d_i(d_i+1)}{d_i+2}\right) \cdot \left(1 + \sum_{i=1}^{m} (d_i+1)\right)$$

$$\stackrel{?}{=} \sum_{\substack{\text{partitions of } \{1,...,m\} \text{ into} \\ \text{two complementary sets} \\ \{r_1,...,r_{m_1}\} \sqcup \{s_1,...,s_{m_2}\}} \frac{\left(4 + \sum_{i=1}^{m} d_i\right)}{\left(2 + \sum_{l=1}^{m} d_l + 1\right)}}{\left(\sum_{l=1}^{m} (d_l+1)\right)}.$$

We will show that (A.2) is valid for any nonempty collection of nonnegative integers $\{d_1, \ldots, d_m\}$.

 $4^{\circ} \ S\acute{E}RIE - TOME \ 49 - 2016 - N^{o} \ 6$

A.2. Identity in terms of multinomial coefficients

Let n_d be the total number of entries d in the set (with multiplicities) $\{d_1, \ldots, d_m\}$. The left-hand-side of conjectural identity (A.2) can be expressed as

$$\left(6 + \sum_{d} \frac{d(d+1)}{d+2} n_d\right) \cdot \left(1 + \sum_{d} (d+1)n_d\right).$$

The right-hand-side can be represented in terms of n_d as

$$\frac{(4+\sum_{d} d\cdot n_{d})!}{(\sum_{d} (d+1)n_{d})!} \cdot \sum_{k_{1}=0}^{n_{1}} \sum_{k_{2}=0}^{n_{2}} \cdots \binom{n_{1}}{k_{1}} \binom{n_{2}}{k_{2}} \frac{(\sum_{d} (d+1)k_{d})! \cdot (\sum_{d} (d+1)(n_{d}-k_{d}))!}{(2+\sum_{d} d\cdot k_{d})! \cdot (2+\sum_{d} d\cdot (n_{d}-k_{d}))!}$$

$$= \frac{n_{1}!n_{2}! \dots (4+\sum_{d} d\cdot n_{d})!}{(\sum_{d} (d+1)n_{d})!} \cdot \sum_{k_{1}=0}^{n_{1}} \sum_{k_{2}=0}^{n_{2}}$$

$$\cdots \frac{(\sum_{d} (d+1)k_{d})!}{k_{1}!k_{2}! \dots (2+\sum_{d} d\cdot k_{d})!} \cdot \frac{(\sum_{d} (d+1)(n_{d}-k_{d}))!}{(n_{1}-k_{1})!(n_{2}-k_{2})! \dots (2+(\sum_{d} d\cdot (n_{d}-k_{d}))!}.$$

Note now that the common factor is (up to four missing factors) is a multinomial coefficient and that the bottom line is a product of two "complementary" multinomial coefficients (with two missing factors each).

A.3. Notation

To simplify the otherwise complicated factorials and terms, we introduce some notation: $\mathbf{k} = (k_1, \ldots, k_m), \mathbf{s} = (s_1, \ldots, s_m), \text{ and } \mathbf{d} = (d_1, \ldots, d_m) \text{ are all } m$ -tuples. We will think of $\mathbf{k}, \mathbf{d} \in (\mathbb{Z}^{\geq 0})^m$, and \mathbf{s} as variables. We write

$$\mathbf{1} = (1, 1, \dots, 1) := \sum_{i=1}^{m} \mathbf{e}_i,$$

where \mathbf{e}_i are the standard basis vectors. Let *n* denote an integer.

Inner Product:

$$\mathbf{k} \cdot \mathbf{d} := \sum_{i=1}^m k_i d_i$$

is the standard inner product.

Factorials:

$$\mathbf{k}! := \Pi_{i=1}^m k_i!$$

Multinomial Coefficients:

$$\binom{n}{\mathbf{k}} := \binom{n}{k_1, \dots, k_m, n - \mathbf{k} \cdot \mathbf{1}}$$

Deletion of variables: Here, we can have i = j:

$$\mathbf{k}^i = \mathbf{k} - \mathbf{e}_i, \mathbf{k}^{i,j} = \mathbf{k} - \mathbf{e}_i - \mathbf{e}_j$$

Powers:

$$\mathbf{s^k} = \Pi_{i=1}^m s_i^{k_i}$$

ANNALES SCIENTIFIQUES DE L'ÉCOLE NORMALE SUPÉRIEURE

1359

We redefine notations **d** and *m* denoting from now on by **d** the original set $\{d_1, \ldots, d_m\}$ with suppressed multiplicities. In other words, we define the new **d** as the set of distinct entries of the original set $\{d_1, \ldots, d_m\}$. We also redefine *m* denoting by *m* the cardinality of the new set **d**. Applying manipulations performed in §A.2 we can rewrite the identity we need to prove in the following way:

(A.3)
$$\frac{6 + \sum_{i=1}^{m} \frac{d_i(d_i+1)}{d_i+2} n_i}{\left(2 + (\mathbf{d}+1) \cdot \mathbf{n}\right) \cdot \left(3 + (\mathbf{d}+1) \cdot \mathbf{n}\right) \cdot \left(4 + (\mathbf{d}+1) \cdot \mathbf{n}\right)} \cdot \begin{pmatrix} 4 + (\mathbf{d}+1) \cdot \mathbf{n} \\ \mathbf{n} \end{pmatrix}$$
$$\stackrel{?}{=} \sum_{\mathbf{k}=0}^{\mathbf{n}} \frac{1}{\left(1 + (\mathbf{d}+1) \cdot \mathbf{k}\right) \left(2 + (\mathbf{d}+1) \cdot \mathbf{k}\right)} \cdot \begin{pmatrix} 2 + (\mathbf{d}+1) \cdot \mathbf{k} \\ \mathbf{k} \end{pmatrix}$$
$$\cdot \frac{1}{\left(1 + (\mathbf{d}+1) \cdot (\mathbf{n}-\mathbf{k})\right) \left(2 + (\mathbf{d}+1)(\mathbf{n}-\mathbf{k})\right)} \cdot \begin{pmatrix} 2 + (\mathbf{d}+1) \cdot \mathbf{k} \\ \mathbf{k} \end{pmatrix}.$$

A.4. Generating Functions

We define

$$F(\mathbf{s}) := \sum_{\mathbf{k} \in (\mathbb{Z}^{\geq 0})^m} \frac{\binom{2 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}}{\mathbf{k}}}{(1 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k})(2 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k})} \mathbf{s}^{\mathbf{k}},$$

and

$$G(\mathbf{s}) := \sum_{\mathbf{k} \in (\mathbb{Z}^{\geq 0})^m} \frac{6 + \sum_{i=1}^m \frac{d_i(d_i+1)}{d_i+2} k_i}{(2 + (\mathbf{d}+1) \cdot \mathbf{k})(3 + (\mathbf{d}+1) \cdot \mathbf{k})(4 + (\mathbf{d}+1) \cdot \mathbf{k})} \cdot \binom{4 + (\mathbf{d}+1) \cdot \mathbf{k}}{\mathbf{k}} \mathbf{s}^{\mathbf{k}}.$$

In terms of these generating functions, the conjectural identity to be proved becomes

 $F^2 \stackrel{?}{=} G.$

A.5. Mohanty's Formula

Our main tools are the combinatorial identities developed by Mohanty [42]. We recall formulas (31) and (32) of [42], in our own notation. Given $a \in \mathbb{N}$, $\mathbf{b}, \mathbf{k} \in (\mathbb{Z}^{\geq 0})^m$, define the *Mohanty coefficient*

$$A(a; \mathbf{b}; \mathbf{k}) := \frac{a}{a + \mathbf{b} \cdot \mathbf{k}} \binom{a + \mathbf{b} \cdot \mathbf{k}}{\mathbf{k}}.$$

We have

THEOREM A.1 (Mohanty [42], (31) and (32)). - With notation as above, we have

$$\sum_{\mathbf{k}\in(\mathbb{Z}^{\geq 0})^m} A(a;\mathbf{b};\mathbf{k})\mathbf{s}^{\mathbf{k}} = z^a,$$

where

$$1 - z + \sum_{i=1}^{m} s_i z^{b_i} = 0.$$

1361

Since we will use only one **b**, namely $\mathbf{b} = \mathbf{d} + \mathbf{1}$, we will abbreviate the Mohanty coefficient by defining $A(a; \mathbf{k}) = A(a; \mathbf{d} + \mathbf{1}; \mathbf{k})$.

In the rest of the appendix we prove:

THEOREM A.2. - We have

$$F^2 = G.$$

More precisely,

(A.4)
$$G(\mathbf{s}) = F^2(\mathbf{s}) - \frac{1}{4}z^2 \left(1 - z + \sum_{i=1}^m s_i z^{d_i + 1}\right)^2,$$

where z is as in Mohanty's Formula Theorem A.1 for $A(a; \mathbf{k}) = A(a; \mathbf{d} + 1; \mathbf{k})$, so

$$1 - z + \sum_{i=1}^{m} s_i z^{d_i + 1} = 0.$$

To prove this formula, we will derive formulas for F (§A.6) and G (§A.7), and show (A.4).

A.6. Formula for F

Our first lemma is the formula for *F*:

LEMMA A.3. – We have

$$F(\mathbf{s}) = \sum_{i=1}^{m} \frac{s_i}{d_i + 2} z^{d_i + 2} - \frac{1}{2} z^2 + z,$$

where

$$1 - z + \sum_{i=1}^{m} s_i z^{d_i + 1} = 0.$$

Proof. – We expand the right hand side using Mohanty's formula, and equalize the s^k terms of the right hand and left hand sides. The right hand side expands, term-by-term, as:

$$\begin{split} \sum_{i=1}^{m} \frac{s_i}{d_i+2} z^{d_i+2} &\longmapsto \sum_{i=1}^{m} \frac{s_i}{d_i+2} \frac{d_i+2}{d_i+2 + (\mathbf{d}+1) \cdot \mathbf{k}} \binom{d_i+2 + (\mathbf{d}+1) \cdot \mathbf{k}}{\mathbf{k}} \\ & \frac{1}{2} z^2 \longmapsto \frac{1}{2} \frac{2}{2 + (\mathbf{d}+1) \cdot \mathbf{k}} \binom{2 + (\mathbf{d}+1) \cdot \mathbf{k}}{\mathbf{k}} \\ & z \longmapsto \frac{1}{1 + (\mathbf{d}+1) \cdot \mathbf{k}} \binom{1 + (\mathbf{d}+1) \cdot \mathbf{k}}{\mathbf{k}} . \end{split}$$

The s^k terms of each of the second and third expressions can be read off directly. For the first, we have:

$$\sum_{i=1}^{m} \frac{s_i}{d_i+2} z^{d_i+2} \longmapsto \sum_{i=1}^{m} \frac{1}{d_i+2+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k}^i} \binom{d_i+2+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k}^i}{\mathbf{k}^i}.$$

Observing that

$$a + d_i + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}^i = a - 1 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k},$$

we can re-write this as

$$\sum_{i=1}^{m} \frac{s_i}{d_i+2} z^{d_i+2} \longmapsto \sum_{i=1}^{m} \frac{1}{1+(\mathbf{d}+\mathbf{1})\cdot \mathbf{k}} \binom{1+(\mathbf{d}+\mathbf{1})\cdot \mathbf{k}}{\mathbf{k}^i}.$$

Thus, our identity reduces to showing that :

$$\frac{\binom{2+(\mathbf{d}+1)\cdot\mathbf{k}}{\mathbf{k}}}{(1+(\mathbf{d}+1)\cdot\mathbf{k})(2+(\mathbf{d}+1)\cdot\mathbf{k})}$$

is the sum of

$$\sum_{i=1}^{m} \frac{1}{1 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}} \binom{1 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}}{\mathbf{k}^{i}},$$

and

$$\frac{1}{1+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k}}\binom{1+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k}}{\mathbf{k}} - \frac{1}{2+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k}}\binom{2+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k}}{\mathbf{k}}.$$

Multiplying through by

$$(1 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k})(2 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}),$$

our identity reduces to showing that

$$\binom{2+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k}}{\mathbf{k}}$$

equals

$$(2 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}) \left(\binom{1 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}}{\mathbf{k}} + \sum_{i=1}^{m} \binom{1 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}}{\mathbf{k}^{i}} \right) - (1 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}) \binom{2 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}}{\mathbf{k}}.$$

Moving the last term to the left hand side, and canceling the resulting $(2 + (d + 1) \cdot k)$, our identity reduces to:

$$\binom{2+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k}}{\mathbf{k}} = \binom{1+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k}}{\mathbf{k}} + \sum_{i=1}^{m} \binom{1+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k}}{\mathbf{k}^{i}},$$

which is the basic identity for multinomial coefficients

$$\binom{n}{\mathbf{k}} = \binom{n-1}{\mathbf{k}} + \sum_{i=1}^{m} \binom{n-1}{\mathbf{k}^{i}},$$

with $n = 2 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}$.

A.7. Formula for G

Our second main lemma is a formula for G:

4° SÉRIE – TOME 49 – 2016 – Nº 6

1363

LEMMA A.4. – We have

$$G(\mathbf{s}) = \frac{3}{4}z^2 - \frac{1}{2}z^3 + \frac{1}{2}\left(\sum_{i=1}^m \frac{d_i s_i z^{d_i+4}}{d_i+2} - \sum_{i=1}^m \frac{(d_i-2)s_i z^{d_i+3}}{d_i+2} - \sum_{i,j=1}^m \frac{d_i (d_i+4)s_i s_j z^{4+d_i+d_j}}{(d_i+2)(4+d_i+d_j)}\right).$$

Before we prove this lemma, we prove Theorem A.2 assuming it.

A.8. Proof of Theorem A.2

We want to show (A.4) assuming Lemma A.4 (and Lemma A.3), that is, we want to show

(A.5)
$$G(\mathbf{s}) \stackrel{?}{=} F^2(\mathbf{s}) - \frac{1}{4}z^2 \left(1 - z + \sum_{i=1}^m s_i z^{d_i + 1}\right)^2.$$

Expanding $F^2(\mathbf{s})$ using

$$F(\mathbf{s}) = \sum_{i=1}^{m} \frac{s_i}{d_i + 2} z^{d_i + 2} - \frac{1}{2} z^2 + z,$$

we obtain

$$F^{2}(\mathbf{s}) = \left(\sum_{i=1}^{m} \frac{s_{i}}{d_{i}+2} z^{d_{i}+2}\right)^{2} + \left(z - \frac{1}{2}z^{2}\right)^{2} + 2\left(\sum_{i=1}^{m} \frac{s_{i}}{d_{i}+2} z^{d_{i}+2}\right) \left(z - \frac{1}{2}z^{2}\right).$$

Expanding this expression for $F^2(\mathbf{s})$, expanding the second term in the right-hand side of (A.5) and simplifying, we obtain three types of terms in the resulting expression for the right-hand side of (A.5):

Simple powers:
$$\frac{3}{4}z^2 - \frac{1}{2}z^3$$
.
Single sums: $\sum_{i=1}^m \left(\frac{2}{d_i+2} - \frac{1}{2}\right)s_i z^{d_i+3} + \sum_{i=1}^m \left(\frac{1}{2} - \frac{1}{d_i+2}\right)s_i z^{d_i+4}$.
Double sum: $\sum_{i,j=1}^m \left(\frac{1}{(d_i+2)(d_j+2)} - \frac{1}{4}\right)s_i s_j z^{4+d_i+d_j}$.

Expanding Lemma A.4 in a similar fashion, we have the corresponding terms for G(s):

Simple powers:
$$\frac{3}{4}z^2 - \frac{1}{2}z^3$$
.
Single sums: $\sum_{i=1}^m \left(\frac{2-d_i}{2(d_i+2)}\right) s_i z^{d_i+3} + \sum_{i=1}^m \left(\frac{d_i}{2(d_i+2)}\right) s_i z^{d_i+4}$.
Double sum: $-\sum_{i,j=1}^m \left(\frac{d_i(d_i+4)}{2(d_i+2)} \frac{1}{4+d_i+d_j}\right) s_i s_j z^{4+d_i+d_j}$.

A quick inspection shows that the simple powers and single sums are equal. For the double sum, we need to combine the (i, j) and (j, i) terms in both sums (note that the terms are identical in the F^2 expansion, but not in the G expansion), and check their equality. The F^2 term is thus

$$-2\left(\frac{1}{4} - \frac{1}{(d_i+2)(d_j+2)}\right)$$

and the G term is

$$-\frac{1}{4+d_i+d_j}\left(\frac{d_i(d_i+4)}{2(d_i+2)}+\frac{d_j(d_j+4)}{2(d_j+2)}\right).$$

To check their equality, we reorganize and obtain:

$$\frac{-4 + (d_i + 2)(d_j + 2)}{2(d_i + 2)(d_j + 2)} \stackrel{?}{=} \frac{1}{2(4 + d_i + d_j)} \left(\frac{d_i(d_i + 4)(d_j + 2) + d_j(d_j + 4)(d_i + 2)}{(d_i + 2)(d_j + 2)} \right).$$

Cancelling and cross-multiplying, this reduces to

$$(4+d_i+d_j)(d_id_j+2d_i+2d_j) = (d_i^2+4d_i)(d_j+2) + (d_j^2+4d_j)(d_i+2),$$

which is easily verified.

A.9. Proof of Lemma A.4

Recall that the s^k term for G is

(A.6)
$$\frac{6 + \sum_{i=1}^{m} \frac{d_i(d_i+1)}{d_i+2}k_i}{(2 + (\mathbf{d}+1) \cdot \mathbf{k})(3 + (\mathbf{d}+1) \cdot \mathbf{k})(4 + (\mathbf{d}+1) \cdot \mathbf{k})} \binom{4 + (\mathbf{d}+1) \cdot \mathbf{k}}{\mathbf{k}} \binom{4 + (\mathbf{d}+1) \cdot \mathbf{k}}{\mathbf{k}}.$$

We observe
$$\frac{d_i(d_i+1)}{(d_i+2)} = d_i - 1 + \frac{2}{d_i+2},$$

and write
$$3$$

$$6 = \frac{3}{2} \left((4 + \mathbf{d} \cdot \mathbf{k}) - \mathbf{d} \cdot \mathbf{k} \right)$$

Using these, we rewrite the term (A.6) as the product of three terms:

Numerator: $\left(\frac{3}{2}(4 + \mathbf{d} \cdot \mathbf{k}) - \sum_{i=1}^{m} \left(\frac{1}{2}d_i + 1 - \frac{2}{d_i+2}\right)k_i\right)$. Partial Fractions: $\left(\frac{1}{2+(\mathbf{d}+1)\cdot\mathbf{k}} - \frac{1}{3+(\mathbf{d}+1)\cdot\mathbf{k}}\right)$. Multinomial Coefficient: $\frac{1}{4+(\mathbf{d}+1)\cdot\mathbf{k}}\binom{4+(\mathbf{d}+1)\cdot\mathbf{k}}{\mathbf{k}} = \frac{(3+(\mathbf{d}+1)\cdot\mathbf{k})!}{\mathbf{k}!(4+\mathbf{d}\cdot\mathbf{k})!}$. We consider terms from this triple product in turn.

A.9.1. k_i -terms. – First, we consider the individual term

$$\left(\frac{1}{2}d_i+1-\frac{2}{d_i+2}\right)k_i\left(\frac{1}{2+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k}}-\frac{1}{3+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k}}\right)\frac{(3+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k})!}{\mathbf{k}!(4+\mathbf{d}\cdot\mathbf{k})!}.$$

Keeping the $\left(\frac{1}{2}d_i + 1 - \frac{2}{d_i+2}\right)$ term outside for now, and considering only the first part of the difference, we are interested in

$$k_i \frac{1}{2 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}} \frac{(3 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k})!}{\mathbf{k}! (4 + \mathbf{d} \cdot \mathbf{k})!} = (3 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}) \frac{(1 + (\mathbf{d} + \mathbf{1})\mathbf{k})!}{\mathbf{k}^i! (4 + \mathbf{d} \cdot \mathbf{k})!}$$

Expanding

$$(3 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}) = (4 + \mathbf{d} \cdot \mathbf{k}) + \mathbf{1} \cdot \mathbf{k}^{i},$$

we first consider

$$\begin{split} (4 + \mathbf{d} \cdot \mathbf{k}) \frac{(1 + (\mathbf{d} + \mathbf{1})\mathbf{k})!}{\mathbf{k}^{i}!(4 + \mathbf{d} \cdot \mathbf{k})!} &= \frac{(1 + (\mathbf{d} + \mathbf{1})\mathbf{k})!}{\mathbf{k}^{i}!(3 + \mathbf{d} \cdot \mathbf{k})!} \\ &= \frac{(d_i + 2 + (\mathbf{d} + \mathbf{1})\mathbf{k}^i)!}{\mathbf{k}^{i}!(d_i + 3 + \mathbf{d} \cdot \mathbf{k}^i)!} \\ &= \frac{1}{d_i + 3} \frac{d_i + 3}{d_i + 3 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}^i} \binom{d_i + 3 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}^i}{\mathbf{k}^i} \\ &= \frac{1}{d_i + 3} A(d_i + 3; \mathbf{k}^i). \end{split}$$

Now expanding $\mathbf{1} \cdot \mathbf{k}^i = (k_i - 1) + \sum_{j \neq i} k_j$, we have the terms

$$\begin{aligned} (k_i - 1) \frac{(1 + (\mathbf{d} + \mathbf{1})\mathbf{k})!}{\mathbf{k}^i!(4 + \mathbf{d} \cdot \mathbf{k})!} &= \frac{1}{4 + 2d_i} \frac{4 + 2d_i}{4 + 2d_i + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}^{i,i}} \binom{4 + 2d_i + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}^{i,i}}{\mathbf{k}^{i,i}} \\ &= \frac{1}{4 + 2d_i} A(4 + 2d_i; \mathbf{k}^{i,i}) \end{aligned}$$

and

$$\begin{split} (k_j) \frac{(1 + (\mathbf{d} + \mathbf{1})\mathbf{k})!}{\mathbf{k}^{i}!(4 + \mathbf{d} \cdot \mathbf{k})!} &= \frac{1}{4 + d_i + d_j} \frac{4 + d_i + d_j}{4 + d_i + d_j + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}^{i,j}} \binom{4 + d_i + d_j + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}^{i,j}}{\mathbf{k}^{i,j}} \\ &= \frac{1}{4 + d_i + d_j} A(4 + d_i + d_j; \mathbf{k}^{i,j}). \end{split}$$

Collecting all of these, we have

$$\begin{split} k_i \frac{1}{2 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}} \frac{(3 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k})!}{\mathbf{k}! (4 + \mathbf{d} \cdot \mathbf{k})!} \\ &= \frac{1}{d_i + 3} A(d_i + 3; \mathbf{k}^i) + \sum_{j=1}^m \frac{1}{4 + d_i + d_j} A(4 + d_i + d_j; \mathbf{k}^{i,j}). \end{split}$$

Next, we work with the factor

$$\begin{split} -k_i \frac{1}{3 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}} \frac{(3 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k})!}{\mathbf{k}!(4 + \mathbf{d} \cdot \mathbf{k})!} &= -\frac{(2 + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k})!}{\mathbf{k}^i!(4 + \mathbf{d} \cdot \mathbf{k})!} \\ &= \frac{(3 + d_i + (\mathbf{d} + \mathbf{1}) \cdot \mathbf{k}^i)!}{\mathbf{k}^i!(4 + d_1 + \mathbf{d} \cdot \mathbf{k}^i)!} \\ &= -\frac{1}{d_i + 4} A(4 + d_i; \mathbf{k}^i). \end{split}$$

These k_i terms come with the factor of

$$\left(\frac{1}{2}d_i + 1 - \frac{2}{d_i + 2}\right) = \frac{d_i(d_i + 4)}{2(d_i + 2)},$$

so we have that their total contribution is:

,

$$\sum_{i=1}^{m} \frac{d_i(d_i+4)}{2(d_i+2)} \left(\frac{A(d_i+3;\mathbf{k}^i)}{d_i+3} - \frac{A(d_i+4;\mathbf{k}^i)}{d_i+4} + \sum_{j=1}^{m} \frac{A(4+d_i+d_j;\mathbf{k}^{i,j})}{4+d_i+d_j} \right).$$

Summing over $\mathbf{k} \in (\mathbb{Z}^{\geq 0})^m$, we obtain, using Mohanty's formula,

(A.7)
$$\sum_{i=1}^{m} \frac{d_i(d_i+4)}{2(d_i+2)} \left(\frac{s_i z^{d_i+3}}{d_i+3} - \frac{s_i z^{d_i+4}}{d_i+4} + \sum_{j=1}^{m} \frac{s_i s_j z^{4+d_i+d_j}}{4+d_i+d_j} \right).$$

A.9.2. $\frac{3}{2}(4 + \mathbf{d} \cdot \mathbf{k})$ -terms. – We now expand the $\frac{3}{2}(4 + \mathbf{d} \cdot \mathbf{k})$ -terms, keeping $\frac{3}{2}$ on the outside for now. That is, we consider

$$(4+\mathbf{d}\cdot\mathbf{k})\left(\frac{1}{2+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k}}-\frac{1}{3+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k}}\right)\frac{(3+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k})!}{\mathbf{k}!(4+\mathbf{d}\cdot\mathbf{k})!}$$

As above, we first work with the term

$$\begin{aligned} (4+\mathbf{d}\cdot\mathbf{k}) \frac{1}{2+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k}} \frac{(3+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k})!}{\mathbf{k}!(4+\mathbf{d}\cdot\mathbf{k})!} &= (3+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k}) \frac{(1+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k})!}{\mathbf{k}!(3+\mathbf{d}\cdot\mathbf{k})!} \\ &= ((3+\mathbf{d}\cdot\mathbf{k})+\mathbf{1}\cdot\mathbf{k}) \frac{(1+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k})!}{\mathbf{k}!(3+\mathbf{d}\cdot\mathbf{k})!}. \end{aligned}$$

The $1 \cdot \mathbf{k}$ term can be split up into individual terms, and as above, we have

$$k_i \frac{(1+(\mathbf{d}+1)\mathbf{k})!}{\mathbf{k}!(3+\mathbf{d}\cdot\mathbf{k})!} = \frac{(1+(\mathbf{d}+1)\mathbf{k})!}{\mathbf{k}^i!(3+\mathbf{d}\cdot\mathbf{k})!} = \frac{1}{d_i+3}A(d_i+3;\mathbf{k}^i).$$

The $(3 + \mathbf{d} \cdot \mathbf{k})$ term yields

$$(3+\mathbf{d}\cdot\mathbf{k})\frac{(1+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k})!}{\mathbf{k}!(3+\mathbf{d}\cdot\mathbf{k})!} = \frac{(1+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k})!}{\mathbf{k}!(2+\mathbf{d}\cdot\mathbf{k})!}$$
$$= \frac{1}{2}\frac{2}{2+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k}}\binom{2+(\mathbf{d}+\mathbf{1})\cdot\mathbf{k}}{\mathbf{k}}$$
$$= \frac{1}{2}A(2;\mathbf{k}).$$

Thus we have

$$(4+\mathbf{d}\cdot\mathbf{k})\frac{1}{2+(\mathbf{d}+1)\cdot\mathbf{k}}\frac{(3+(\mathbf{d}+1)\cdot\mathbf{k})!}{\mathbf{k}!(4+\mathbf{d}\cdot\mathbf{k})!} = \frac{A(2;\mathbf{k})}{2} + \sum_{i=1}^{m}\frac{A(d_i+3;\mathbf{k}^i)}{d_i+3}.$$

We are left with the term

$$\begin{aligned} -(4+\mathbf{d}\cdot\mathbf{k})\frac{1}{3+(\mathbf{d}+1)\cdot\mathbf{k}}\frac{(3+(\mathbf{d}+1)\cdot\mathbf{k})!}{\mathbf{k}!(4+\mathbf{d}\cdot\mathbf{k})!} &= -\frac{1}{3}\frac{3}{3+(\mathbf{d}+1)\cdot\mathbf{k}}\binom{3+(\mathbf{d}+1)\cdot\mathbf{k}}{\mathbf{k}} \binom{3+(\mathbf{d}+1)\cdot\mathbf{k}}{\mathbf{k}} \\ &= -\frac{1}{3}A(3;\mathbf{k}). \end{aligned}$$

Combining the above, and recalling the coefficient of $\frac{3}{2}$, and summing over $\mathbf{k} \in (\mathbb{Z}^{\geq 0})^m$ we have the total contribution of the $\frac{3}{2} (4 + \mathbf{d} \cdot \mathbf{k})$ -terms:

(A.8)
$$\frac{3}{2} \left(\frac{1}{2} z^2 - \frac{1}{3} z^3 + \sum_{i=1}^m \frac{s_i z^{d_i+3}}{d_i+3} \right).$$

A.9.3. Combining terms. - To conclude, we combine equations (A.7) and (A.8) to obtain

$$G(\mathbf{s}) = \frac{3}{2} \left(\frac{1}{2} z^2 - \frac{1}{3} z^3 + \sum_{i=1}^m \frac{s_i z^{d_i+3}}{d_i+3} \right) - \sum_{i=1}^m \frac{d_i (d_i+4)}{2(d_i+2)} \left(\frac{s_i z^{d_i+3}}{d_i+3} - \frac{s_i z^{d_i+4}}{d_i+4} + \sum_{j=1}^m \frac{s_i s_j z^{4+d_i+d_j}}{4+d_i+d_j} \right).$$

4° SÉRIE – TOME 49 – 2016 – Nº 6

Collecting terms, we have

$$\begin{split} G(\mathbf{s}) &= \frac{3}{4}z^2 - \frac{1}{2}z^3 + \sum_{i=1}^m \frac{s_i z_i^{d_i+3}}{d_i+3} \left(\frac{3}{2} - \frac{d_i(d_i+4)}{2(d_i+2)}\right) \\ &+ \frac{1}{2} \left(\sum_{i=1}^m \frac{d_i s_i z^{d_i+4}}{d_i+2} - \sum_{i,j=1}^m \frac{d_i(d_i+4)}{d_i+2} \frac{s_i s_j z^{4+d_i+d_j}}{4+d_i+d_j}\right). \end{split}$$

Finally, using

$$\frac{3}{2} - \frac{d_i(d_i+4)}{2(d_i+2)} = -\frac{1}{2} \frac{(d_i+3)(d_i-2)}{d_i+2},$$

we get

$$\frac{s_i z_i^{d_i+3}}{d_i+3} \left(\frac{3}{2} - \frac{d_i(d_i+4)}{2(d_i+2)}\right) = -\frac{1}{2} \frac{d_i-2}{d_i+2} z^{d_i+3}.$$

Substituting this into our expression for G, we obtain as desired

$$G(\mathbf{s}) = \frac{3}{4}z^2 - \frac{1}{2}z^3 + \frac{1}{2}\left(\sum_{i=1}^m \frac{d_i s_i z^{d_i+4}}{d_i+2} - \sum_{i=1}^m \frac{(d_i-2)s_i z^{d_i+3}}{d_i+2} - \sum_{i,j=1}^m \frac{d_i (d_i+4)s_i s_j z^{4+d_i+d_j}}{(d_i+2)(4+d_i+d_j)}\right).$$

Appendix B

Counting pillowcase covers

In this appendix we describe the original approach to calculating the volume of the moduli space of Abelian and quadratic differentials suggested by H. Masur, M. Kontsevich, and the authors, and developed with success by A. Eskin and A. Okounkov, see [25, 26]. This approach was also used in [51] and [20]. The key idea is to translate the volume calculation into a counting problem for "integer points", which geometrically correspond to *square-tiled surfaces* for the moduli spaces of Abelian differentials and to *pillowcase covers* for the moduli spaces of quadratic differentials.

In §B.1 we show why volume calculation is equivalent to counting the lattice points. In §B.2 we recall the definition of the *pillowcase cover*, show that counting of lattice points is equivalent to the counting problem for pillowcase covers and prove Theorem 1.3.

B.1. Reduction of volume calculation to counting lattice points

The volume of a stratum $Q_1(d_1, \ldots, d_k)$ is defined by (4.5) as

 $\operatorname{Vol} \mathcal{Q}_1(d_1,\ldots,d_k) = \dim_{\mathbb{R}} \mathcal{Q}(d_1,\ldots,d_k) \cdot \mu(C(\mathcal{Q}_1(d_1,\ldots,d_k))),$

where $\mu(C(Q_1(d_1,\ldots,d_k)))$ is the total volume of the "cone" $C(Q_1(d_1,\ldots,d_k)) \subset Q(d_1,\ldots,d_k)$ measured by means of the volume element $d\mu$ on $Q(d_1,\ldots,d_k)$ defined in §4.1. The total volume of the cone $C(Q_1(d_1,\ldots,d_k))$ is the limit of the appropriately normalized Riemann sums.

The volume element $d\mu$ is defined as a linear volume element in cohomological coordinates, normalized by certain specific lattice. Choose a positive ε such that $1/\varepsilon$ is integer, and

1367

consider a sublattice of the initial lattice of index $(1/\varepsilon)^{\dim_{\mathbb{R}} \mathcal{Q}(d_1,...,d_k)}$ partitioning every side of the initial lattice into $1/\varepsilon$ pieces. The corresponding Riemann sums count the number of points of the sublattices which get inside the cone. Thus, by definition of the measure μ we get

$$\mu(C(\mathcal{Q}_1(d_1,\ldots,d_k)) = \lim_{\varepsilon \to 0} \varepsilon^{\dim_{\mathbb{R}}\mathcal{Q}(d_1,\ldots,d_k)} (\text{number of points of the } \varepsilon \text{-sublattice inside } C(\mathcal{Q}_1(d_1,\ldots,d_k)))$$

We assume that $1/\varepsilon$ is integer. Note that a flat surface S represents a point of the ε -lattice, if and only if the surface $(1/\varepsilon) \cdot S$ (in the sense of Definition (4.1)) represents a point of the integer lattice. Denoting by $C(Q_N(d_1, \ldots, d_k))$ the set of flat surfaces in the stratum $Q(d_1, \ldots, d_k)$ of area at most N/2, and taking into consideration that

$$\operatorname{area}((1/\varepsilon) \cdot S) = 1/\varepsilon^2 \cdot \operatorname{area}(S)$$

we can rewrite the above relation as

(B.1)
$$\mu(C(\mathcal{Q}_1(d_1,\ldots,d_k))) = \lim_{N \to +\infty} N^{-\dim_{\mathbb{C}} \mathcal{Q}(d_1,\ldots,d_k)}.$$

(number of lattice points inside the cone $C(Q_N(d_1, \ldots, d_k))$).

B.2. Lattice points, square-tiled surfaces, and pillowcase covers

Let $\Lambda \subset \mathbb{C}$ be a lattice, and let $\mathbb{T}^2 = \mathbb{C}/\Lambda$ be the associated torus. The quotient

$$\mathscr{P} := \mathbb{T}^2 / \pm$$

by the map $z \to -z$ is known as the *pillowcase orbifold*. It is a sphere with four $(\mathbb{Z}/2)$ -orbifold points (the corners of the pillowcase). The quadratic differential $(dz)^2$ on \mathbb{T}^2 descends to a quadratic differential on \mathscr{P} . Viewed as a quadratic differential on the Riemann sphere, $(dz)^2$ has simple poles at corner points. When the lattice Λ is the standard integer lattice $\mathbb{Z} \oplus i\mathbb{Z}$, the flat torus \mathbb{T}^2 is obtained by isometrically identifying the opposite sides of a unit square, and the pillowcase \mathscr{P} is obtained by isometrically identifying two squares with the side 1/2 by the boundary, see Figure 14.



FIGURE 14. Pillowcase cover (on the left) over the pillowcase orbifold (on the right). A general pillowcase cover it is not necessarily glued from two identical polygons.

Consider a connected ramified cover $\hat{\mathscr{P}}$ over \mathscr{P} of degree N having ramification points only over the corners of the pillowcase. Clearly, $\hat{\mathscr{P}}$ is tiled by 2N squares of the size $(1/2) \times (1/2)$ in such way that the squares do not superpose and the vertices are glued to the vertices. Coloring

1369

the two squares of the pillowcase \mathscr{P} one in black and the other in white, we get a chessboard coloring of the square tiling of the cover $\hat{\mathscr{P}}$: the white squares are always glued to the black ones and vice versa.

LEMMA B.1. – Let S be a flat surface in the stratum $Q(d_1, \ldots, d_k)$. The following properties are equivalent:

- 1. The surface S represents a lattice point in $Q(d_1, \ldots, d_k)$;
- 2. S is a cover over \mathcal{P} ramified only over the corners of the pillow;
- 3. *S* is tiled by black and white $(1/2) \times (1/2)$ squares respecting the chessboard coloring.

Proof. – We have just proved that (2) implies (3). To prove that (1) implies (2) we define the following map from S to \mathscr{P} . Fix a zero or a pole P_0 on S. For any $P \in S$ consider a path $\gamma(P)$ joining P_0 to P having no self-intersections and having no zeroes or poles inside. The restriction of the quadratic differential q to such $\gamma(P)$ admits a well-defined square root $\omega = \pm \sqrt{q}$, which is a holomorphic form on the interior of γ . Define

(B.2)
$$P \mapsto \left(\int_{\gamma(P)} \omega \mod \mathbb{Z} \oplus i\mathbb{Z} \right) / \pm 1$$

Of course, the path $\gamma(P)$ is not uniquely defined. However, since the flat surface S represents a lattice point (see the definition in §4.1), the difference of the integrals of ω over any two such paths $\gamma_1(P)$ and $\gamma_2(P)$ belongs to $\mathbb{Z} \oplus i\mathbb{Z}$, so taking the quotient over the integer lattice and over \pm we get a well-defined map. By definition of the pillowcase \mathscr{P} we have, $\mathscr{P} = (\mathbb{C}/(\mathbb{Z} \oplus i\mathbb{Z}))/\pm$. Thus, we have defined a map $S \to \mathscr{P}$. It follows from the definition of the map, that it is a ramified cover, and that all regular points of the flat surface S are regular points of the cover. Thus, all ramification points are located over the corners of the pillowcase.

A similar consideration shows that (3) implies (1).

Let $Sq_N(d_1, \ldots, d_k)$ be the number of surfaces in the stratum $Q(d_1, \ldots, d_k)$ tiled with at most N black and N white squares respecting the chessboard coloring. Lemma B.1 allows us to rewrite Formula (B.1) as follows:

$$\mu(C(\mathcal{Q}_1(d_1,\ldots,d_k))) = \lim_{N \to +\infty} N^{-\dim_{\mathbb{C}} \mathcal{Q}(d_1,\ldots,d_k)} \cdot \operatorname{Sq}_N(d_1,\ldots,d_k).$$

Taking into consideration (4.5) we get

(B.3)
$$\operatorname{Vol} Q_1(d_1, \dots, d_k) = 2 \dim_{\mathbb{C}} Q(d_1, \dots, d_k)$$

 $\cdot \lim_{N \to +\infty} N^{-\dim_{\mathbb{C}} Q(d_1, \dots, d_k)} \cdot \operatorname{Sq}_N(d_1, \dots, d_k).$

We now state and prove two lemmas which we use in the proof of Theorem 1.3.

LEMMA B.2. – For any η , ν as above the following asymptotic relation is valid:

(B.4)
$$\lim_{N \to +\infty} \frac{\sum_{d=1}^{N} \operatorname{Cov}_{4d}^{0}(\eta, \nu)}{\sum_{d=1}^{N} \operatorname{Cov}_{4d}^{0, \boxplus}(\eta, \nu)} = 2^{\ell(\eta)},$$

where $\ell(\eta)$ is the number of entries in η .

Proof. – Let *P* be a zero of even degree of a quadratic differential, and $U_{\varepsilon}(P)$ a multidisc of flat radius ε centered at *P*. Choosing ε sufficiently small, we can assume that $U_{\varepsilon}(P)$ is embedded into the ambient flat surface. (For example, for the flat surface $\hat{\mathscr{P}}$ induced from the standard $\frac{1}{2} \times \frac{1}{2}$ square pillowcase \mathscr{P} by means of the cover $\pi_{\mathbb{H}} : \hat{\mathscr{P}} \to \mathscr{P}$ one can choose any ε satisfying $0 < \varepsilon < \frac{1}{2}$.) Choose an orientation of the vertical direction in $U_{\varepsilon}(P)$. For any vector $\vec{v} \in \mathbb{R}^2$ such that $\|\vec{v}\| < \varepsilon$ there is a unique way to move the zero *P* in direction \vec{v} by the distance $\|\vec{v}\|$ via a local move inside U(P) keeping the flat metric outside of U(P) unchanged. The corresponding local surgery (called, depending on the author and the context *Schiffer*



FIGURE 15. Cartoon of a local move of a zero of even degree.

variation, or deformation along the kernel (or Rel) foliation, etc) is represented in Figure 15, where the separatrix rays ("prongs") adjacent to P are chosen to be parallel to the vector \vec{v} . This local deformation can be performed as follows. Make short slits along all prongs in direction \vec{v} and open them. The original conical point of a cone angle $\pi \cdot 2k$ gives rise to 2k marked points on the sides of the slits. Move the marked points along the sides of the slit by the distance $\|\vec{v}\|$ in direction \vec{v} and zip the slits back up to the new marked points. All new marked points get identified into a single conical singularity with the cone angle $\pi \cdot 2k$. Note that usually, the flat surfaces obtained after deformations along vectors \vec{v} and $-\vec{v}$ are generically non-isomorphic.

Consider now a pillowcase cover $\pi_{\mathbb{H}} : \hat{\mathcal{P}} \to \mathcal{P}$ as above. As the base sphere choose the standard pillowcase \mathcal{P} endowed with the quadratic differential $q_0 = (dz)^2$. It has four simple poles at the corners of the pillow and no other singularities. Pulling back $(dz)^2$ via $\pi_{\mathbb{H}}$ gives a quadratic differential on the covering \mathbb{CP}^1 with zeros and simple poles of degrees $\{\nu_i - 2\}$ and $\{2\eta_j - 2\}$ and with no other singularities. Thus, by construction the pillowcase cover $\hat{\mathcal{P}} := (\mathbb{CP}^1, \pi^*q_0)$ belongs to $\mathcal{Q}(\eta, \nu)$.

Move the zero P_1 of degree $2\eta_1 - 2$ in direction \vec{v} as above. The deformed flat surface inherits a structure of a ramified cover over the pillowcase orbifold \mathscr{P} . The corresponding cover can be defined by a formal construction as Equation (B.2), or can be seen in plain terms as follows. We have deformed our flat structure only inside a neighborhood $U(P_1)$, so we let the projection of the complement of $U(P_1)$ to the pillowcase orbifold \mathscr{P} unchanged. The neighborhood $U(P_1)$ is glued from even number 2k of half-disks as in Figure 15; we define the projection of each half-disk to the pillowcase orbifold \mathscr{P} unchanged. The definition matches on the common boundaries of the half-discs. By construction, the deformed cover $\pi': \hat{\mathscr{P}}' \to \mathscr{P}$, has the ramification profile $(2\eta_2, \ldots, 2\eta_{\ell(\eta)}, \nu, 2^{2d-|\eta|+\eta_1-|\nu|/2})$ over $0 \in \mathscr{P}$, profile $(\eta_1, 1^{4d-\eta_1})$ over the projection of the deformed zero P'_1 , and profile (2^{2d}) over the other three corners of \mathscr{P} . The cover π' is unramified elsewhere. Consider now the same pillowcase cover $\pi_{\mathbb{H}} : \hat{\mathscr{P}} \to \mathscr{P}$ as above and move the zero P_1 of degree $2\eta_1 - 2$ of the initial quadratic differential q in direction $-\vec{v}$. Clearly we get a pillowcase cover $\pi'' : \hat{\mathscr{P}}'' \to \mathscr{P}$ with exactly the same profile as $\pi' : \hat{\mathscr{P}}' \to \mathscr{P}$. Moreover, since the zero P_1 of q on the original cover $\hat{\mathscr{P}}$ was projected to a conical singularity of the pillowcase orbifold \mathscr{P} with the cone angle π , moving from the corresponding corner of the pillow in directions \vec{v} and $-\vec{v}$ we get to the same point of the pillowcase orbifold \mathscr{P} . In other words, the zero P'_1 of the deformed quadratic differential q' on $\hat{\mathscr{P}}'$ obtained by moving the zero P_1 of q'' on $\hat{\mathscr{P}}''$ obtained by moving the zero P_1 of q'' on $\hat{\mathscr{P}}''$ obtained by moving the zero P_1 of q in direction $-\vec{v}$. The number of covers $\hat{\mathscr{P}}$ for which the resulting covers $\hat{\mathscr{P}}'$ and $\hat{\mathscr{P}}''$ are isomorphic has asymptotics of lower order in N than $\sum_{d=1}^{N} \operatorname{Cov}^0_{4d}(\eta, \nu)$.

Moving all even-order zeroes $P_1, \ldots, P_{\ell(\eta)}$ in directions of pairwise-distinct vectors $\pm \vec{v}_1, \ldots, \pm \vec{v}_{\ell(\eta)}$ we establish a $2^{\ell(\eta)}$ -to-one correspondence (up to a term of lower order asymptotics in N) between covers π and π_{\boxplus} .

Quadratic differentials induced from dz^2 on the standard pillowcase orbifold via the pillowcase covers (1.5) constructed in §1.2 have the following structure. All zeroes of odd degrees of such differentials are projected to the same corner of the pillow, while all zeroes of even degrees are projected to pairwise distinct non-corner points. Quadratic differentials induced from dz^2 on the standard pillowcase orbifold via the pillowcase covers $\pi_{\mathbb{H}} : \hat{\mathcal{P}} \to \mathcal{P}$ as above have slightly different structure. Namely, *all* their zeroes (no matter of odd or even degree) project to the same corner of the pillowcase orbifold, so they are really *square-tiled*. Moreover, since all preimages under $\pi_{\mathbb{H}}$ of the three remaining corners of the pillowcase orbifold are regular points of the flat metric, the resulting flat surface $\hat{\mathcal{P}}$ can be tiled with 1×1 squares (compared to $\frac{1}{2} \times \frac{1}{2}$ squares of the pillowcase orbifold). Our next lemma proves, that (in genus zero) this new square tiling with larger squares admits a chessboard coloring.

LEMMA B.3. – Any pillowcase cover $\pi_{\mathbb{H}} : \hat{\mathcal{P}} \to \mathcal{P}$ of genus zero with a ramification profile as above decomposes into two consecutive covers

$$\pi: \hat{\mathscr{P}} \to \mathscr{P}_4 \to \mathscr{P},$$

where $\mathscr{P}_4 \to \mathscr{P}$ is a cover of order 4 of a pillowcase orbifold of size 1×1 over the standard pillowcase of size $\frac{1}{2} \times \frac{1}{2}$.

REMARK B.4. – Without the condition that the genus of $\hat{\mathscr{P}}$ as above is equal to zero the assertion of Lemma B.3 is no longer true in general.

Proof of Lemma B.3. – Consider the decomposition of $\hat{\mathcal{P}}$ into maximal horizontal cylinders. We associate to this decomposition a finite graph. The edges of the graph are in one-to-one correspondence with the cylinders. The vertices of the graph are in one-to-one correspondence with connected components of singular horizontal layers. Two edges have common vertex if the corresponding maximal cylinders are adjacent to the same connected component of the critical horizontal layer.

Note, that for square-tiled surfaces of genus zero the resulting graph is, actually, a tree which we denote by T. To prove the lemma we first prove that perimeters of all horizontal

cylinders are even integer numbers. The proof is an induction in the number of horizontal cylinders.

The base of induction corresponds to the case when $\hat{\mathscr{P}}$ has a single horizontal cylinder. Then $\hat{\mathscr{P}}$ has only two singular layers, one on each side of the cylinder. Consider one of the horizontal singular layers as a graph of horizontal saddle connections. Since $\hat{\mathscr{P}}$ is a sphere, the corresponding graph is, actually, a tree. By construction each saddle connection has integer length. The waist curve of the cylinder follows each saddle connection twice, so its perimeter is twice the sum of lengths of all saddle connections in the layer, and hence the perimeter is an even number.

When the number of horizontal cylinders is greater than one, analogous consideration shows that perimeters of all cylinders represented by the extremity edges (leaves) of the tree T have even perimeters. Chopping one of these cylinders out from the initial flat surface $\hat{\mathcal{P}}$ and isometrically identifying the parts of the boundary in the natural way we get a new pillowcase cover $\tilde{\mathcal{P}}$ satisfying the same properties as before. By induction all its horizontal cylinders have even perimeters.

Having proved that the perimeters of all horizontal cylinders are even integers we apply the induction in the number of cylinders one more time proving now that the tiling of $\hat{\mathscr{P}}$ with 1×1 squares admits chessboard coloring.

Now everything is ready to prove Theorem 1.3.

Proof of Theorem 1.3. – By Lemma B.3 a pillowcase cover $\pi_{\mathbb{H}} : \hat{\mathscr{P}} \to \mathscr{P}$ of genus zero and of degree 4d with ramification data (η, ν) as above uniquely defines a square-tiled pillowcase cover of degree d in $Q(\eta, \nu)$.

Reciprocally, consider an arbitrary square-tiled surface S as in Lemma B.1 above in the stratum $\mathcal{Q}(\eta, \nu)$, and let d be the degree of the corresponding cover over \mathscr{P} . Subdividing each square into four; considering the underlying pillowcase as \mathscr{P}_4 and postcomposing the initial cover $S \to \mathscr{P}_4$ with the cover $\mathscr{P}_4 \to \mathscr{P}$ we get a pillowcase cover with singularity pattern (η, ν) . This implies that

$$\sum_{d=1}^{N} \operatorname{Cov}_{4d}^{0,\boxplus}(\eta,\nu) = \operatorname{Sq}_{N}(\nu_{1}-2,\nu_{2}-2,\ldots,2\eta_{1}-2,2\eta_{2}-2,\ldots).$$

Applying Equation (B.4) from Lemma B.2, taking into consideration that

$$\dim_{\mathbb{C}} \mathcal{Q}(\nu_1 - 2, \nu_2 - 2, \dots, 2\eta_1 - 2, 2\eta_2 - 2, \dots) = \ell(\nu) + \ell(\eta) - 2$$

and applying Equation (B.3) we complete the proof of Theorem 1.3.

Appendix C Equidistribution of Circle Translates by Jon Chaika

We use a variation of an argument of G. A. Margulis to obtain equidistribution of circles from exponential mixing of the Teichmüller geodesic flow on Q_1 . The strategy is similar in spirit to [17, Section 3.6].

4° SÉRIE – TOME 49 – 2016 – Nº 6

C.1. Notation

As in §1.3, let $\mathcal{B} = \mathcal{B}(k_1, \ldots, k_n)$ be the space of directional billiard tables, that is, billiard tables with interior angles $(\frac{\pi}{2}k_1, \ldots, \frac{\pi}{2}k_n)$ and a distinguished direction. \mathcal{B} has the natural the measure $\mu_{\mathcal{B}}$ which is the product measure of the Lebesgue measure arising from the side lengths and the angular measure $d\phi$.

Given $\Pi \in \mathcal{B}$, let q_{Π} be the meromorphic quadratic differential given by gluing together two copies of Π . Recall that $\Pi \mapsto q_{\Pi}$ from \mathcal{B} to $\mathcal{Q} = \mathcal{Q}(k_1-2, \ldots, k_n-2)$ is a local embedding where $\mathcal{Q}(d_1, \ldots, d_n)$ is the stratum of quadratic differentials with zeros of order d_1, \ldots, d_n . Using this map, we may view \mathcal{B} as a subset of \mathcal{Q} . Let $\mathcal{Q}_1 \subset \mathcal{Q}$ denote the subset of surfaces of flat area 1/2 (see Convention 4.1), and let $\mu_1 = \mu_{\mathcal{Q}_1}$ denote the Masur-Veech measure on \mathcal{Q}_1 . Let \mathcal{B}_1 denote the intersection of $\mathcal{B} \subset \mathcal{Q}$ with \mathcal{Q}_1 . We abuse notation by denoting the restriction of the measure $\mu_{\mathcal{B}}$ to \mathcal{B}_1 again by $\mu_{\mathcal{B}}$. Let $\tilde{\mathcal{B}}_1$ denote the subset of \mathcal{B}_1 where the direction of the billiard flow is parallel to one of the sides, and let $\mu_{\tilde{\mathcal{B}}}$ be the restriction of the measure to $\tilde{\mathcal{B}}_1$. We recall notation for important one-parameter subgroups of $SL(2, \mathbb{R})$:

$$g_t = \begin{pmatrix} e^t & 0\\ 0 & e^{-t} \end{pmatrix}$$
 and $r_{\theta} = \begin{pmatrix} \cos \theta & \sin \theta\\ -\sin \theta & \cos \theta \end{pmatrix}$.

C.2. Equidistribution

We define the function space

$$\mathcal{L}_{c}^{0} := \left\{ f \in C_{c}(\mathcal{Q}_{1}) : f \text{ is 1-Lipschitz }, \int_{\mathcal{Q}_{1}} f d\mu_{1} = 0 \right\}$$

with respect to the Euclidean metric induced by local coordinates on Q_1 (see, for example [3, §5] for a formal definition of this distance). We denote \mathcal{L}_c the space without the mean 0 condition.

THEOREM C.1. – Let
$$f \in \mathcal{L}_c^0$$
. Then for $\mu_{\mathscr{B}}$ -almost every right angled billiard q_{Π} we have

$$\lim_{T \to \infty} \frac{1}{2\pi} \int_0^{2\pi} f(g_T r_{\theta} q_{\Pi}) \, d\theta = \frac{1}{\mu_1(\mathcal{Q}_1)} \int_{\mathcal{Q}_1} f \, d\mu_1 = 0.$$

C.3. Small arcs and strategy

The strategy to prove Theorem C.1 is to break the integral over the circle into (exponentially) small arcs so that the limit converges as desired. Let $\mathcal{M}_{\varepsilon} \subset \mathcal{Q}_1$ be the ε -thick part of the stratum, that is, the set of $q \in \mathcal{Q}_1$ so that all saddle connections on q have length at least ε .

PROPOSITION C.2. – Let $f \in \mathcal{L}_c^0$ and $\delta > 0$. Define

$$S_{\delta} := \left\{ \theta \in [0, 2\pi) \setminus \left(B\left(0, \delta\right) \cup B\left(\frac{\pi}{2}, \delta\right) \cup B\left(\pi, \delta\right) \cup B\left(\frac{3\pi}{2}, \delta\right) \right) \right\}.$$

 S_{δ} avoids neighborhoods of directions parallel to the sides. There exists an exponentially decaying function $v : \mathbb{R}^+ \to [0, \pi)$ such that for any $\varepsilon > 0$

$$\lim_{N \to \infty} \frac{1}{2v(\varepsilon N)} \int_{-v(\varepsilon N)}^{v(\varepsilon N)} f(g_{\varepsilon N} r_{\theta + \phi} q_{\Pi}) d\phi = \frac{1}{\mu_1(\mathcal{Q}_1)} \int_{\mathcal{Q}_1} f d\mu_1$$

ANNALES SCIENTIFIQUES DE L'ÉCOLE NORMALE SUPÉRIEURE

(37)

for $\mu_{\mathscr{B}}$ -almost every $q_{\Pi} \in \tilde{\mathscr{B}}_1 \cap \mathscr{M}_{\delta}$, Lebesgue almost every $\theta \in S_{\delta}$.

We prove Theorem C.1 assuming Proposition C.2 in §C.7. To prove Proposition C.2, we estimate the L^2 -norms of the functions

$$F_N(q) = \frac{1}{2v(N)} \int_{-v(N)}^{v(N)} f(g_N r_\theta q) d\theta,$$

on $\mathcal{M}_{\delta} \cap r_{\theta} \tilde{\mathcal{B}}_1$, the δ -thick part of the rotated billiard subvariety. To estimate

$$\int_{r_{\theta}\tilde{\mathscr{B}}_{1}\cap\mathscr{M}_{\delta}}(F_{N}(q))^{2}\,d\mu_{r_{\theta}\mathscr{B}} = \int_{r_{\theta}\tilde{\mathscr{B}}_{1}\cap\mathscr{M}_{\delta}}\frac{1}{4v(N)^{2}}\int_{-v(N)}^{v(N)}\int_{-v(N)}^{v(N)}(g_{N}r_{x}q)f(g_{N}r_{y}q)\,dx\,dy\,d\mu_{r_{\theta}\mathscr{B}},$$

we separate the domain of integration into two pieces: one where x and y are very close, and another when they are sufficiently separated. Heuristically, for x and y sufficiently separated, the translates $g_N r_x q$ and $g_N r_y q$ move away from each other exponentially in N, and thus become uncorrelated due to exponential mixing of the Teichmüller flow. This is made precise in Proposition C.10. For x and y sufficiently close, we estimate trivially by the measure of the set.

C.4. Exponential recurrence, mixing, and contraction

To implement the above strategy, we need three crucial technical results on Teichmüller geodesic flow.

C.4.1. *Exponential Recurrence.* – Athreya [1] showed that most (in an exponential sense) trajectories spend at least half their life in the thick part of a stratum. Precisely, let

$$G_L(\varepsilon) = \left\{ q \in \mathcal{Q}_1 : |\{0 \le t < T : g_t q \in \mathcal{M}_\varepsilon\}| > \frac{T}{2} \text{ for all } T > L \right\}$$

denote the set of $q \in Q_1$ so that the g_t -trajectory of q eventually (after time L) spends at least half its life in $\mathcal{M}_{\varepsilon}$.

THEOREM C.3 ([1, Theorem 2.3]). – For all small enough $\varepsilon > 0$, there exists $C, \xi > 0$ such that for all L > 0

$$\mu_1(G_L(\varepsilon)) > (1 - Ce^{-\xi L})\mu_1(\mathcal{Q}_1).$$

C.4.2. Exponential Mixing. – Avila-Resende [5], building on work of Avila-Gouezel-Yoccoz [4], showed that the Teichmüller geodesic flow is exponentially mixing. Let d be the Riemannian distance on $SL(2, \mathbb{R})$ induced by the Killing form. For functions f, g on Q_1 , and $M \in SL(2, \mathbb{R})$, define the M-correlation

$$\mathscr{C}(f,g,M) = \left| \int_{\mathscr{Q}_1} f(Mq)g(q)d\mu_1 - \int_{\mathscr{Q}_1} fd\mu_1 \int_{\mathscr{Q}_1} gd\mu_1 \right|.$$

THEOREM C.4. – ([5], [4, Theorem 2.14]) There exists constants C, λ so that if h_1, h_2 are Lipshitz and compactly supported then there exists C_K that depend only on the smallest systole of a surface in the compact support such that for any $M \in SL(2, \mathbb{R})$,

 $\mathcal{C}(h_1, h_2, M) \le C \left(C_K + \|h_1\|_{\infty} + \|h_1\|_{Lip} \right) \left(C_K + \|h_2\|_{\infty} + \|h_2\|_{Lip} \right) e^{-\lambda d(M, id)}.$

C.4.3. *Exponential Contraction.* – Eskin-Mirzakhani-Rafi [24], following Forni [28], proved an important result on the hyperbolicity of the Teichmüller flow. For a subset $A \subset \mathbb{R}^n$ we use |A| to denote its Lebesgue measure.

THEOREM C.5 ([24, Lemma 8.3]). – Given a fixed compact part $\mathcal{M}_{\frac{e}{2}}$ there exists c, C > 0 such that if q and q' differ only along a stable manifold for g_t (that is, if they share the same horizontal foliation) then

$$d_S(q_tq, q_tq') < \tilde{C}d_S(q, q')e^{-c} \left| \left\{ t > 0: g_tq \in \mathcal{M}_{\frac{\varepsilon}{2}} \right\} \right|.$$

d_S denotes Hodge distance along the stable manifold. See [24, Section 8.2].

C.5. Mixing on open sets

In this section we state and prove our first key Proposition C.6, which will play a crucial role in the proof of Proposition C.2. Proposition C.6 uses exponential mixing (Theorem C.4) in a crucial fashion.

Given an open set $U \subset \mathcal{M}_{\delta}$, let $\partial_{\varepsilon} U$ denote the ε -neighborhood of the boundary ∂U . We say U is *polynomially regular* with regularity polynomial P if there is a polynomial P so that

$$\mu_1(\partial_{\varepsilon} U) \le P(\varepsilon)$$

PROPOSITION C.6. – Let $\delta > 0$ and let $U \subset \mathcal{M}_{\delta}$ be polynomially regular. Let $f, h \in \mathcal{L}_{c}^{0}$. Then there exist $\hat{\lambda}, D > 0$ and $\ell_{0} < 1$ so that for all $0 < \ell < \ell_{0}$

$$\left|\int_U f(g_t q) h(g_{(1+\ell)t} q) d\mu_1(q)\right| \le D e^{-\hat{\lambda} \ell t}.$$

Moreover, the constants only depend on δ , $||f||_{\infty}$, $||h||_{\infty}$ and the regularity polynomial of U.

The first step in the proof is the following effective equidistribution lemma for translates of polynomially regular sets.

LEMMA C.7. – Let $\delta > 0$, $U \subset \mathcal{M}_{\delta}$ be polynomially regular and $f \in \mathcal{L}_{0}^{c}$. There exist $E, \lambda_{f} > 0$ so that

$$\left|\frac{1}{\mu_1(U)}\int_U f(g_t q)d\mu_1(q)\right| \le Ee^{-\lambda_f t}.$$

The constants depend only on δ , f and the regularity polynomial of U.

Proof. – Let $U_r = \{q \in U : B(q, r) \subset U\}$ and

$$h_{\varepsilon}(q) = \chi_U(q) \left(1 - \frac{1}{\varepsilon} d(q, U_{\varepsilon}) \right).$$

Notice that h_{ε} is ε^{-1} -Lipshitz. We will obtain the lemma by applying exponential mixing (Theorem C.4) to the functions f and h_{ε} .

$$\left|\int_{U} f(g_t q) d\mu_1\right| \leq \left|\int_{\mathcal{Q}_1} h_{\varepsilon}(q) f(g_t q) d\mu_1\right| + \|\chi_U - h_{\varepsilon}\|_{\infty} \mu_1 \left\{q : \chi_U(q) \neq h_{\varepsilon}(q)\right\} \|f\|_{\infty}.$$

By Theorem C.4 we have that there exists C_3 (subsuming the various constants)

$$\left| \int_{\mathcal{Q}_1} h_{\varepsilon}(q) f(g_t q) d\mu_1 \right| \le C_3 \frac{1}{\varepsilon} e^{-\lambda t}$$

By our assumption on U there exists C, d (essentially the leading coefficient and degree of the regularity polynomial) so that

$$\|\chi_U - h_{\varepsilon}\|_{\infty} \mu_1 \{q : \chi_U(q) \neq h_{\varepsilon}(q)\} \|f\|_{\infty} \le C\varepsilon^d.$$

Letting $\varepsilon = e^{-\frac{\lambda}{4}t}$ we obtain

$$\int_{U} f(g_t q) d\mu_1 \le C_{\varepsilon, f} e^{\frac{\lambda}{4}t} e^{-\lambda t} + C e^{-\frac{\lambda d}{4}t}$$

To complete the lemma, let $\lambda_f = \min\left\{\frac{\lambda}{2}, \frac{d\lambda}{4}\right\}$.

Applying this to small balls, we obtain

COROLLARY C.8. – Let $\varepsilon > 0$, $q_0 \in \mathcal{M}_{\varepsilon+e^{-k}}$ and $f \in \mathcal{L}^0_c$. Then there exists $C_{k,\varepsilon,f} > 0$ so that for all k > 0,

$$\left|\frac{1}{\mu_1(B(q_0, e^{-k})))} \int_{B(q_0, e^{-k})} f(g_t q) d\mu_1(q)\right| \le C_{k,\varepsilon,f} e^{-\lambda_f t}.$$

Moreover, $C_{k,\varepsilon,f}$ can be chosen to be $C_{\varepsilon,f}e^{kL_{\varepsilon,f}}$ where $L_{\varepsilon,f}$ depends only on ε, f .

Applying this corollary to $f = \chi_U - \mu_1(U)$ for a polynomially regular set, we obtain

COROLLARY C.9. – Let $\delta > 0$. Let $U \subset \mathcal{M}_{\delta}$ be polynomially regular. There exist k_2, D_2, λ_2 so that for any $q_0 \in \mathcal{M}_{\delta}$ we have, for all r > 0

$$\left|\frac{1}{\mu_1(B(q_0, e^{-r}))} \int_{B(q_0, e^{-r})} \chi_U(g_{k_2r}q) d\mu_1(q) - \mu_1(U)\right| < D_2 e^{-\lambda_2 r}.$$

The number k_2 can be chosen to be either positive or negative and the corollary holds for all large enough (in absolute value) k_2 . The constants can be chosen to only depend on δ and the regularity polynomial of U.

Proof. - Let

$$H_{r,q_0}(q) = \left(e^{2r} - 1\right) d\left(q, B(q_0, e^{-r})\right) \chi_{B(q_0, e^{-r} + e^{-2r})}(q).$$

By the regularity of U and using period coordinates on the stratum we have that there exist C'_1 , d_1 and C'_2 , d_2 so that

$$\left| \int_{B(q_0, e^{-r})} \chi_U(g_{kr}q) d\mu_1 - \int_{\mathcal{Q}_1} h_{\varepsilon}(q) H_{r, q_0}(q) d\mu_1 \right| < C_1' \varepsilon^{d_1} + C_2' e^{-2rd_2}.$$

By Theorem C.4, since

$$||H_{r,q_0}||_{Lip} \le e^{2r} \text{ and } ||h_{\varepsilon}||_{Lip} \le \frac{1}{\varepsilon}$$

we have that there exists C_3 (subsuming the various constants) so that

$$\int_{\mathcal{Q}_1} h_{\varepsilon}(g_{kr}q) H_{q_0,r}(q) d\mu_1 - \int H_{q,r} d\mu_1 \int h_{\varepsilon} d\mu_1 \bigg| < C_3 \frac{1}{\varepsilon} e^{2r} e^{-\lambda k}.$$

Combining these two estimates the corollary follows.

4° SÉRIE - TOME 49 - 2016 - Nº 6

1377

Note that if $f \in \mathcal{L}_c^0$ then $f \circ g_t$ is Ce^{3t} -Lipshitz. We use this observation to prove Proposition C.6 by splitting U into balls of size e^{-4t} where $f \circ g_t$ is basically constant. Then we apply Lemma C.8 to these balls. This gives us the required independence.

Proof of Proposition C.6. – We want to estimate

$$\left|\int_{U} f(g_t q) h(g_{t+s} q) d\mu_1(q)\right| = \left|\int_{\mathcal{Q}_1} f(q) h(g_s q) \chi_U(g_{-t} q) d\mu_1(q)\right|.$$

By doing an extra integration over small balls, we rewrite this as

$$\left| \int_{\mathcal{Q}_1} \frac{1}{\mu_1(B(q_0, e^{-4s}))} \int_{B(q_0, e^{-4s})} f(q) h(g_s q) \chi_U(g_{-t} q) d\mu_1(q) d\mu_1(q_0) \right|.$$

Since $f, h \in \mathcal{L}^0_c$ we can estimate their values in small balls by values at the center points, allowing us to bound from above the previous integral by

$$\left| \int_{\mathcal{Q}_1} \frac{1}{\mu_1(B(q_0, e^{-4s}))} \int_{B(q_0, e^{-4s})} \left(f(q_0) h(g_s q_0) + O(e^{-4s}) \|f\|_{Lip} \|h \circ g_s\|_{Lip} \right) \chi_U(g_{-t}q) d\mu_1(q) d\mu_1(q_0) \right|.$$

Integrating the error term out, we derive the further estimate

$$\left| \int_{\mathcal{Q}_1} \frac{1}{\mu_1(B(q_0, e^{-4s}))} f(q_0) h(g_s q_0) \int_{B(q_0, e^{-4s})} \chi_U(g_{-t}q) d\mu_1(q) d\mu_1(q_0) \right| + O\left(e^{-\frac{3s}{4}}\right) d\mu_1(q) d\mu_1(q_0) d\mu$$

By Corollary C.9 if $s < \frac{t}{k_2}$ then for any q_0 in the support of f (which is assumed to be compact)

$$\left|\frac{1}{\mu_1(B(q_0, e^{-4s}))} \int_{B(q_0, e^{-4s})} \chi_U(g_{-t}q) d\mu_1(q) - \mu_1(U)\right| < D_2 e^{-\lambda_2 s}.$$

So we obtain

(C.1)
$$\left| \int_{\mathcal{Q}_1} \frac{1}{\mu_1(B(q_0, e^{-4s}))} f(q_0) h(g_s q_0) \int_{B(q_0, e^{-4s})} \chi_U(g_{-t}q) d\mu_1(q) d\mu_1(q_0) \right|$$
$$= \left| \mu(U) \int_{\mathcal{Q}_1} f(q_0) h(g_s q_0) d\mu_1(q_0) \right| + O\left(D_2 e^{-\lambda_2 s}\right) \|f\|_{\infty} \|h\|_{\infty}.$$

Applying Theorem C.4 Proposition C.6 follows.

In this subsection, we state our other key Lemma Proposition C.10, which estimates the correlation of $f \in \mathcal{L}_c^0$ with a translate of f along a thickened g_t -translate of the billiard manifold $\tilde{\mathcal{B}}_1$. Fix $\varepsilon > 0$ so that Theorem C.3 holds.

PROPOSITION C.10. – If $f \in \mathcal{L}^0_c$, $\delta, a > 0$ and $\theta \notin \{0, \frac{\pi}{2}, \pi, 3\frac{\pi}{2}\}$ then there exist constants C'_1, λ' and $C_{2,\theta} < 1$ such that for any $M \in SL(2, \mathbb{R})$ with $d(M, Id) \leq e^{tC_{2,\theta}}$ we have

$$\left| \int_{-a}^{a} \int_{r_{\theta} \tilde{\mathcal{B}}_{1} \cap \mathcal{M}_{\delta}} f(Mg_{t+\ell}q) f(g_{t+\ell}q) \, d\mu_{r_{\theta} \tilde{\mathcal{B}}_{1}}(q) d\ell \right| < C_{\theta} e^{-\lambda_{\theta}' d(M, Id)}$$

Here, and below, $\mu_{r_{\theta}\mathcal{B}_{1}}$ denotes $(r_{\theta})_{*}\mu_{\tilde{\mathcal{B}}_{1}}$. $C_{\theta}, C_{2,\theta}$ and λ'_{θ} depend on f, δ , a and θ . Moreover fixing f, a, δ the dependence on θ is continuous.

The proof of Proposition C.10 relies on all of the technical ingredients from the previous sections: exponential recurrence (Theorem C.3) exponential mixing (Theorem C.4) and exponential contraction (Theorem C.5) as well as Proposition C.6. Our key lemma is:

LEMMA C.11. – Let $\delta > 0$ and $U \subset \mathcal{M}_{\delta}$ be polynomially regular. Let $f, h \in \mathcal{L}_{c}^{0}$. Then there exist constants \hat{C}, C_{2}, λ such that for any $M \in SL(2, \mathbb{R})$ with $||M|| < C_{2}t$ we have

$$\left| \int_{U} f(g_t q) h(Mg_t q) \, d\mu_1(q) \right| < \hat{C} e^{-\lambda d(M, Id)}.$$

As in Proposition C.6, \hat{C} and λ depend on f, h, δ and the regularity polynomial of U.

Proof. – There exist $\theta, \phi \in S^1, s \in \mathbb{R}$ with $e^s = ||M||$ so that $M = r_{\theta}g_s r_{\phi}$.

$$\begin{split} \int_{U} f(g_t q) h(Mg_t q) \, d\mu_1(q) &= \int_{Q} f(q) h(r_\theta g_s r_\phi q) \chi_U(g_{-t} q) d\mu_1(q) \\ &= \int_{\mathcal{Q}_1} (f \circ r_{-\phi}(q)) (h \circ r_\theta(g_s q)) \chi_U(g_{-t} r_{-\phi} q) d\mu_1(q). \end{split}$$

We now follow the approach of the proof of Proposition C.6 and for convenience introduce $h_{\theta} = h \circ r_{\theta}, f_{\phi} = f \circ r_{-\phi} \in \mathcal{L}_{c}^{0}$.

$$\begin{aligned} & \left| \int_{Q_1} \frac{1}{\mu_1(B(q, e^{-4s}))} \int_{B(q, e^{-4s})} f_{\phi}(\omega) h_{\theta}(g_s \omega) \chi_U(g_{-t} r_{-\phi} \omega) d\mu_1(\omega) d\mu_1(q) \right| \\ &= \left| \int_{Q_1} \frac{1}{\mu_1(B(q, e^{-4s}))} f_1(q) h_1(g_s q) \int_{B(q, e^{-4s})} \chi_U(g_{-t} r_{-\phi} \omega) d\mu_1(\omega) d\mu_1(q) \right| + O\left(e^{-\frac{3s}{4}}\right). \end{aligned}$$

Now

$$\begin{aligned} \left| \frac{1}{\mu_1(B(q, e^{-4s}))} \int_{B(q, e^{-4s})} \chi_U(g_{-t}r_{-\phi}\omega) d\mu_1(\omega) - \mu_1(U) \right| \\ &= \left| \frac{1}{\mu_1(B(q, e^{-4s}))} \int_{r_{-\phi}B(q, e^{-4s})} \chi_U(g_{-t}\omega) d\mu_1(\omega) - \mu_1(U) \right| < D_2 e^{-\lambda_2 s} \end{aligned}$$

since $r_{-\phi}B(q, e^{-4s})$ has the same regularity polynomial as $B(q, e^{-4s})$. This implies the lemma.

Proof of Proposition C.10. – Let $\theta \notin \{0, \frac{\pi}{2}, \pi, \frac{3\pi}{2}\}$. Intersect $r_{\theta}\tilde{\mathcal{B}}_1$ with $\mathcal{M}_{\varepsilon}$ and flow it by a small interval $\{g_{\ell}, |\ell| < a\}$, and consider

$$\bigcup_{|\ell| < a} g_{\ell} r_{\theta} \tilde{\mathcal{B}}_1 \cap \mathcal{M}_{\varepsilon}$$

Thicken it by c' > 0 along the stable manifold for g_t , and call the resulting set V. We pick a and c' small enough so that the intersection of V with the support of f has a local product structure (as stable \times unstable \times flow) with respect to the Teichmüller flow g_t . By the continuity of the trigonometric functions in the proof of Proposition 3.3 a and c' can be chosen to depend continuously on θ . Let Φ denote the local projection from V to the a-thickened and θ -rotated billiard subvariety

$$\bigcup_{|\ell| < a} g_{\ell} r_{\theta} \tilde{\mathcal{B}}_1.$$

By Proposition 3.3,

$$\mu_1(V) > 0$$

By Corollary C.11, if $||M||_{op} < \hat{C}_{2,\theta}$ we have that

$$\left| \int_{V \cap G_t} f(Mg_t q) f(g_t q) \, d\mu_1 \right| \leq \hat{C}_{\theta} e^{-\lambda_{\theta} d(M, Id)} + \mu_1(G_t^c).$$

By the continuity of r_{θ} and the construction of V the polynomial that bounds the decay of an ε neighborhood of the boundary of V can be chosen to depend continuously on θ . So \hat{C}_{θ} and λ_{θ} depend continuously on θ . By exponential recurrence (Theorem C.3) $\mu_1(G_t^c)$ decays exponentially in t and so there exists C''_{θ} , λ''_{θ} such that

$$\left|\int_{V\cap G_t} f(Mg_t q) f(g_t q) \, d\mu_1\right| \le C_{\theta}'' e^{-\lambda_{\theta}'' d(M, Id)}$$

Now for each $q \in V$ there exists $q' \in r_{\theta} \tilde{\mathcal{B}}_1$ on the same stable manifold which is distance at most c' away. It follows from the exponential contraction of g_t (Theorem C.5) that for t large enough and $q' \in G_t$ then

$$d(g_t q, g_t q') < Cc' e^{-\frac{c}{2}t}.$$

By our assumption that $f \in \mathcal{L}^0_c$ it follows that

$$|f(g_tq) - f(g_tq')| < \tilde{C}c'e^{-\frac{c}{2}t}$$

and

$$|f(Mg_tq) - f(Mg_tq')| \le ||M||_{\operatorname{op}} \tilde{C}c'e^{-\frac{c}{2}t},$$

where $\|\cdot\|_{\text{op}}$ denotes the operator norm of $\text{SL}(2, \mathbb{R})$ acting linearly on \mathbb{R}^2 . By our assumption on V and the fact that f is 1-Lipschitz it follows that

$$\begin{aligned} \left| \int_{V \cap G_t} f(Mg_t q) f(g_t q) d\mu_1(q) \right| \\ &\geq \left| \int_{V \cap G_t} f\left(Mg_t \Phi(q) + e_q\right) \left(f(g_t \Phi(q) + e'_q) d\mu_1(q) \right| \right. \\ &\geq \left| \int_{[0,c']^k} \int_{-a}^a \int_{r_\theta \tilde{\mathcal{B}}_1 \cap \mathcal{M}_{\frac{\epsilon}{2}}} \left(f(Mg_{\ell+t}q') + e_{q',\ell,s} \right) \left(f(g_{t+\ell}q') + e'_{q',\ell,s} \right) d\mu_{r_\theta \tilde{\mathcal{B}}}(q') d\ell d\lambda^k(s) \right| + \zeta e^{-\xi t}, \end{aligned}$$

ANNALES SCIENTIFIQUES DE L'ÉCOLE NORMALE SUPÉRIEURE

1379

where

$$|e'_q|, |e'_{q',s}| < \tilde{C}c'e^{-\frac{c}{2}t}$$
 and $|e_q|, |e_{q',s}| < \|M\|_{\mathrm{op}}\tilde{C}c'e^{-\frac{c}{2}t}$.

In the last inequality of the integral estimate, $\zeta e^{-\xi t}$ comes from the exponential recurrence result Theorem C.3. This establishes the proposition.

C.7. Proof of Theorem C.1

We prove our main Theorem C.1 assuming our key tool Proposition C.2. By Proposition C.2, for every $\delta > 0$ there exists a T_0 such that for any $T > T_0, T \in \varepsilon \mathbb{N}$ and set S with

$$\mu_{\mathcal{B}}(S) \ge \mu_{\mathcal{B}}(\mathcal{B}_1 \cap \mathcal{M}_\delta) - \delta$$

we have that for each $q \in S$ a subset G_q of S_δ with

$$\lambda(G_q) \ge 2\pi - 8\delta - \delta$$

such that for each $\theta \in G_q$ we have

$$\left|\frac{1}{2v(T)}\int_{-v(T)}^{v(T)}f(g_Tr_{\theta+\phi}q)d\phi\right|<\delta.$$

It follows that for $q \in S$ and $T > T_0$ we have

$$\left|\frac{1}{2\pi}\int_{0}^{2\pi}f(g_{T}r_{\phi}q)\,d\phi\right| = \left|\frac{1}{2\pi}\int_{0}^{2\pi}\frac{1}{2v(T)}\int_{-v(T)}^{v(T)}f(g_{T}r_{\theta+\phi}q)d\phi d\theta\right|.$$

We break this integral into two pieces, over G_q and its complement. We have

$$\left|\frac{1}{2\pi}\int_{G_q}\frac{1}{2v(T)}\int_{-v(T)}^{v(T)}f(g_Tr_{\theta+\phi}q)d\phi d\theta\right|\leq \delta$$

and

$$\left| \int_{G_q^c} \frac{1}{2v(T)} \int_{-v(T)}^{v(T)} f(g_T r_{\theta+\phi} q) d\phi d\theta \right| \le 9\delta \|f\|_{\infty}.$$

So we have

$$\left|\frac{1}{2\pi}\int_0^{2\pi} f(g_T r_\phi q) \, d\phi\right| \le \delta + 9\delta \|f\|_\infty \le 10\delta \max\left(1, \|f\|_\infty\right).$$

Because f is 1-Lipschitz we have that if

$$\limsup_{T \to \infty} \left| \frac{1}{2\pi} \int_0^{2\pi} f(g_T r_{\phi} q) d\phi \right| \le \varepsilon + \limsup_{N \in \mathbb{N}\varepsilon} \left| \frac{1}{2\pi} \int_0^{2\pi} f(g_N r_{\phi} q) d\phi \right|.$$

Since δ and ε are arbitrary Theorem C.1 follows.

4° SÉRIE - TOME 49 - 2016 - Nº 6

1380
C.7.1. *Proof of Proposition C.2.* – To prove Proposition C.2, we estimate the L^2 -norms of the integrals

$$F_N(q) = \frac{1}{2v(N)} \int_{-v(N)}^{v(N)} f(g_N r_\theta q) d\theta$$

where we define v(N) below. We have

$$\int_{r_{\theta}\tilde{\mathcal{B}}_{1}\cap\mathcal{M}_{\delta}} (F_{N}(q))^{2} d\mu_{r_{\theta}\tilde{\mathcal{B}}} = \int_{r_{\theta}\tilde{\mathcal{B}}_{1}\cap\mathcal{M}_{\delta}} \frac{1}{4v(N)^{2}} \int_{-v(N)}^{v(N)} \int_{-v(N)}^{v(N)} (g_{N}r_{x}q) f(g_{N}r_{y}q) dx dy d\mu_{r_{\theta}\mathcal{B}}.$$

Changing the order of integration, we obtain

$$\int_{-v(N)}^{v(N)} \int_{g_N r_x(r_\theta \tilde{\mathcal{B}}_1 \cap \mathcal{M}_\delta)} \frac{1}{4v(N)^2} \int_{-v(N)}^{v(N)} f(g_N r_{x-y} g_{-N} q) f(q) \, dy \, d((g_N)_* \mu_{r_{x+\theta} \tilde{\mathcal{B}}}) \, dx.$$

To ease notation, $\mu_{\mathcal{B}}$ will denote $(g_N)_*\mu_{r_{x+\theta}\tilde{\mathcal{B}}}$ for the remainder of this proof. We note that

$$r_{\theta} = h_{-\tan\theta}^{\tau} g_{\log\cos\theta} h_{\tan\theta}$$

where

$$h_s = \begin{pmatrix} 1 & s \\ 0 & 1 \end{pmatrix}, h_s^{\tau} = \begin{pmatrix} 1 & 0 \\ s & 1 \end{pmatrix},$$

so the previous expression is

$$\int_{-v(N)}^{v(N)} \int_{g_N r_{x+\theta}(\tilde{\mathcal{B}}_1 \cap \mathcal{M}_{\delta})} \frac{1}{4v(N)^2} \int_{-v(N)}^{v(N)} f(h_{e^{-2N}\tan(x-y)}g_{\log(\cos(x-y))}h_{e^{2N}\tan(x-y)}q)f(q) \, dy \, d\mu_{\mathcal{B}} \, dx.$$

Now consider $C_{2,\theta}$ from Proposition C.10. Recall that we defined

$$S_{\delta} = [0, 2\pi) \setminus \left(B\left(0, \frac{\delta}{2}\right) \cup B\left(\frac{\pi}{2}, \frac{\delta}{2}\right) \cup B\left(\pi, \frac{\delta}{2}\right) \cup B\left(\frac{3\pi}{2}, \frac{\delta}{2}\right) \right),$$

and

$$C_2 = \min_{\theta \in S_{\delta}} C_{2,\theta}.$$

Observe that this is defined and greater than zero because $C_{2,\theta}$ is continuous in θ and S_{δ} is compact. Let

$$v(N) = \min\left\{e^{\left(-2 + \frac{C_2}{3}\right)N}, \frac{\delta}{2}\right\}.$$

Then

$$\left| h_{e^{-2N} \tan(x-y)}^{\tau} g_{\log(\cos(x-y))} \right| < e^{-N} \text{ for } x, y \in [-v(N), v(N)].$$

Since f is 1-Lipschitz we can dominate the integral by

$$e^{-N} + \int_{-v(N)}^{v(N)} \int_{g_N r_{x+\theta}(\tilde{\mathscr{B}}_1 \cap \mathscr{M}_{\delta})} \frac{1}{4v(N)^2} \int_{-v(N)}^{v(N)} f(h_{e^{2N} \tan(x-y)}q) f(q) \, dy \, d\mu_{\mathscr{B}} \, dx.$$

We break the domain of integration into pieces, and we estimate the integral on each separately. Let

$$\Delta_N := \left\{ x, y \in \left[-v(N), v(N) \right], |x - y| \le e^{\left(-2 + \frac{C_2}{6} \right)N} \right\}$$

be a small neighborhood of the diagonal in $[-v(N), v(N)]^2$. The first piece P_1 is when $(x, y) \notin \Delta_N$:

$$P_1 := \{ (x, y, q) : (x, y) \notin \Delta_N, q \in g_N \left(r_{x+\theta} \mathcal{B}_1 \cap \mathcal{M}_\delta \right) \}.$$

ANNALES SCIENTIFIQUES DE L'ÉCOLE NORMALE SUPÉRIEURE

1381

We add an integral over time so we can estimate the integral over P_1 using Proposition C.10, yielding

$$\begin{split} \frac{1}{4v(N)^2} \left(\int_{-a}^{a} \int_{P_1} f(h_{e^{2N} \tan(x-y)} g_{\ell} q) f(q) \, dy \, d\mu_{\mathcal{B}} \, dx d\ell \right) \\ &= \frac{1}{4v(N)^2} \int_{(x,y)\notin\Delta_N} \int_{-a}^{a} \int_{r_{\theta}\tilde{\mathcal{B}}_1\cap\mathcal{M}_{\delta}} f(h_{e^{2N} \tan(x-y)} g_{N+\ell} q) f(g_{N+\ell} q) d\mu_{\mathcal{B}} \, d\ell \, dx \, dy \\ &\leq \frac{1}{4v(N)^2} \int_{(x,y)\notin\Delta_N} C_1 e^{-\lambda' \frac{C_2}{6}N} dx \, dy \leq C_1 e^{-\lambda' \frac{C_2}{6}N} dx \, dy \\ \end{split}$$

To see this is justified, first observe that the domain of integration is appropriate because for all large enough N, $\theta + x \in S_{\delta}$. Second, the size the matrices is appropriate because by our choice of v(N) we have

$$e^{2N}2v(N) + 1 < e^{C_2N}$$

for all N sufficiently large and and so

$$\|h_{e^{2N}\tan(x-y)}\| < 2e^{C_2N} \text{ for all } x, y \in [-v(N), v(N)]$$

since tan is 2-Lipschitz on $\left[-\frac{\pi}{4}, \frac{\pi}{4}\right]$. Moreover, since

$$|\tan(x-y)| \ge \frac{1}{2}|x-y|$$
 and $x, y \notin \Delta_N$

we have

$$||h_{e^{2N}\tan(x-y)}|| > \frac{1}{2}e^{\frac{C_2}{6}N}.$$

Our second piece, P_2 , is when x and y are close:

$$P_2 := \left\{ (x, y, q) : (x, y) \in \Delta_N, q \in g_N \left(r_{x+\theta} \tilde{\mathcal{B}}_1 \cap \mathcal{M}_\delta \right) \right\}.$$

We have, via naive (measure and $\|\cdot\|_{\infty}$) estimates

$$\begin{split} \frac{1}{4v(N)^2} \left(\int_{P_2} f(h_{e^{2N}\tan(x-y)}q)f(q) \, d\mu_{\mathcal{B}} \, dy \, dx \right) &\leq \frac{\|f\|_{\infty}^2 |\Delta_N|}{4v(N)^2} \\ &\leq e^{-\frac{C_2}{6}N} \|f\|_{\infty}^2. \end{split}$$

Combining the estimates on the integrals over P_1 and P_2 we obtain

$$\int_{r_{\theta}\tilde{\mathscr{B}}_{1}\cap\mathcal{M}_{\delta}} (F_{N}(q))^{2} d\mu_{r_{\theta}\mathscr{B}} \leq C_{1}' e^{-\lambda' \frac{C_{2}}{6}N} + e^{-\frac{C_{2}}{6}N} \|f\|_{\infty}^{2} + e^{-N}$$

So there exists $\delta > 0$ such that for all large enough N.

$$\int_{-a}^{a} \int_{r_{\theta} \tilde{\mathcal{G}}_{1} \cap \mathcal{M}_{\delta}} \left(F_{N}(g_{\ell}q) \right)^{2} d\mu_{r_{\theta} \tilde{\mathcal{G}}}(q) d\ell \leq e^{-\delta N}.$$

Let

$$m_N(q,a) = \min_{\ell \in [-a,a]} |F_N(g_\ell q)|.$$

We have, for any $\eta > 0$, that

$$\mu_{\mathscr{B}}\left\{q\in r_{\theta}\tilde{\mathscr{B}}_{1}\cap \mathscr{M}_{\delta}: m_{N}(q,a)>\eta\right\}\leq \frac{1}{2a\eta^{2}}e^{-\delta N}.$$

4° SÉRIE - TOME 49 - 2016 - Nº 6

Indeed,

$$2a\eta^{2}\mu_{\mathscr{B}}\left\{q\in r_{\theta}\tilde{\mathscr{B}}_{1}\cap\mathscr{M}_{\delta}: m_{N}(q,a)>\eta\right\}\leq \int_{-a}^{a}\int_{r_{\theta}\tilde{\mathscr{B}}_{1}\cap\mathscr{M}_{\delta}}\left(F_{N}(g_{\ell}q)\right)^{2}d\mu_{r_{\theta}\mathscr{B}}(q)d\ell.$$

Since for any $\eta > 0$, we have that

$$\sum_{N=1}^{\infty} \frac{1}{\eta^2} e^{-\delta N} < \infty,$$

the easy half of the Borel-Cantelli lemma implies that for $\mu_{\mathcal{B}}$ -almost every $q \in \tilde{\mathcal{B}}_1 \cap \mathcal{M}_{\delta}$, the set

$$\{N \ge 1 : m_N(q, a) > \eta\}$$

is finite. Because F is 1-Lipshitz, for any such q we have that there exists N_0 so that

$$|F_N(q)| < 2a + \eta$$
 for all $N > N_0$.

Since η and a are arbitrary and our estimates hold for all $\theta \in S_{\delta}$ the proposition follows. \Box

BIBLIOGRAPHY

- J. S. ATHREYA, Quantitative recurrence and large deviations for Teichmüller geodesic flow, *Geom. Dedicata* 119 (2006), 121–140.
- [2] J. S. ATHREYA, A. ESKIN, A. ZORICH, Counting generalized Jenkins-Strebel differentials, *Geom. Dedicata* 170 (2014), 195–217.
- [3] A. AVILA, S. GOUËZEL, Small eigenvalues of the Laplacian for algebraic measures in moduli space, and mixing properties of the Teichmüller flow, *Ann. of Math.* 178 (2013), 385–442.
- [4] A. AVILA, S. GOUËZEL, J.-C. YOCCOZ, Exponential mixing for the Teichmüller flow, *Publ. Math. IHÉS* 104 (2006), 143–211.
- [5] A. AVILA, M. J. RESENDE, Exponential mixing for the Teichmüller flow in the space of quadratic differentials, *Comment. Math. Helv.* 87 (2012), 589–638.
- [6] M. BAINBRIDGE, Euler characteristics of Teichmüller curves in genus two, *Geom. Topol.* 11 (2007), 1887–2073.
- [7] M. BAINBRIDGE, Billiards in L-shaped tables with barriers, *Geom. Funct. Anal.* 20 (2010), 299–356.
- [8] C. BOISSY, Configurations of saddle connections of quadratic differentials on CP¹ and on hyperelliptic Riemann surfaces, *Comment. Math. Helv.* 84 (2009), 757–791.
- [9] K. CALTA, Veech surfaces and complete periodicity in genus two, J. Amer. Math. Soc. 17 (2004), 871–908.
- [10] K. CALTA, K. WORTMAN, On unipotent flows in $\mathscr{H}(1,1)$, Ergodic Theory Dynam. Systems **30** (2010), 379–398.
- [11] M. COHEN, B. WEISS, Parking garages with optimal dynamics, *Geom. Dedicata* 161 (2012), 157–167.
- [12] V. DELECROIX, E. GOUJARD, P. ZOGRAF, A. ZORICH, Integer square-tiled surfaces and interval exchanges of fixed combinatorial type: equidistribution, counting, volumes, in progress.

1383

- [13] V. DELECROIX, P. HUBERT, S. LELIÈVRE, Diffusion for the periodic wind-tree model, Ann. Sci. Éc. Norm. Supér. 47 (2014), 1085–1110.
- [14] V. DELECROIX, A. ZORICH, Cries and whispers in windtree forests, preprint arXiv:1502.06405.
- [15] A. DOUADY, J. HUBBARD, On the density of Strebel differentials, *Invent. math.* 30 (1975), 175–179.
- [16] A. ESKIN, M. KONTSEVICH, A. ZORICH, Sum of Lyapunov exponents of the Hodge bundle with respect to the Teichmüller geodesic flow, *Publ. Math. IHÉS* 120 (2014), 207–333.
- [17] A. ESKIN, G. MARGULIS, S. MOZES, Upper bounds and asymptotics in a quantitative version of the Oppenheim conjecture, Ann. of Math. 147 (1998), 93–141.
- [18] A. ESKIN, J. MARKLOF, D. WITTE MORRIS, Unipotent flows on the space of branched covers of Veech surfaces, *Ergodic Theory Dynam. Systems* 26 (2006), 129–162.
- [19] A. ESKIN, H. MASUR, Asymptotic formulas on flat surfaces, *Ergodic Theory Dynam*. Systems 21 (2001), 443–478.
- [20] A. ESKIN, H. MASUR, M. SCHMOLL, Billiards in rectangles with barriers, *Duke Math. J.* 118 (2003), 427–463.
- [21] A. ESKIN, H. MASUR, A. ZORICH, Moduli spaces of abelian differentials: the principal boundary, counting problems, and the Siegel-Veech constants, *Publ. Math. IHÉS* 97 (2003), 61–179.
- [22] A. ESKIN, M. MIRZAKHANI, Invariant and stationary measures for the SL(2, ℝ) action on moduli space, preprint arXiv:1302.3320.
- [23] A. ESKIN, M. MIRZAKHANI, A. MOHAMMADI, Isolation, equidistribution, and orbit closures for the $SL(2, \mathbb{R})$ -action on moduli space, preprint arXiv:1305.3015, to appear in *Annals of Math*.
- [24] A. ESKIN, M. MIRZAKHANI, K. RAFI, Counting closed geodesics in strata, preprint arXiv:1206.5574, to appear in *Invent. math*.
- [25] A. ESKIN, A. OKOUNKOV, Asymptotics of numbers of branched coverings of a torus and volumes of moduli spaces of holomorphic differentials, *Invent. math.* 145 (2001), 59–103.
- [26] A. ESKIN, A. OKOUNKOV, Pillowcases and quasimodular forms, in Algebraic geometry and number theory, Progr. Math. 253, Birkhäuser, 2006, 1–25.
- [27] A. ESKIN, A. OKOUNKOV, R. PANDHARIPANDE, The theta characteristic of a branched covering, Adv. Math. 217 (2008), 873–888.
- [28] G. FORNI, Deviation of ergodic averages for area-preserving flows on surfaces of higher genus, Ann. of Math. 155 (2002), 1–103.
- [29] E. GOUJARD, Volumes of strata of moduli spaces of quadratic differentials: getting explicit values, preprint arXiv:1501.01611.
- [30] E. GUTKIN, C. JUDGE, Affine mappings of translation surfaces: geometry and arithmetic, *Duke Math. J.* 103 (2000), 191–213.
- [31] S. KERCKHOFF, H. MASUR, J. SMILLIE, Ergodicity of billiard flows and quadratic differentials, Ann. of Math. 124 (1986), 293–311.

4° SÉRIE - TOME 49 - 2016 - Nº 6

1385

- [32] M. KONTSEVICH, Intersection theory on the moduli space of curves and the matrix Airy function, Comm. Math. Phys. 147 (1992), 1–23.
- [33] H. MASUR, Interval exchange transformations and measured foliations, *Ann. of Math.* 115 (1982), 169–200.
- [34] H. MASUR, Closed trajectories for quadratic differentials with an application to billiards, *Duke Math. J.* 53 (1986), 307–314.
- [35] H. MASUR, Lower bounds for the number of saddle connections and closed trajectories of a quadratic differential, in *Holomorphic functions and moduli, Vol. I (Berkeley, CA, 1986)*, Math. Sci. Res. Inst. Publ. 10, Springer, New York, 1988, 215–228.
- [36] H. MASUR, The growth rate of trajectories of a quadratic differential, *Ergodic Theory Dynam. Systems* 10 (1990), 151–176.
- [37] H. MASUR, J. SMILLIE, Hausdorff dimension of sets of nonergodic measured foliations, Ann. of Math. 134 (1991), 455–543.
- [38] H. MASUR, S. TABACHNIKOV, Rational billiards and flat structures, in *Handbook of dynamical systems, Vol. 1A*, North-Holland, Amsterdam, 2002, 1015–1089.
- [39] H. MASUR, A. ZORICH, Multiple saddle connections on flat surfaces and the principal boundary of the moduli spaces of quadratic differentials, *Geom. Funct. Anal.* 18 (2008), 919–987.
- [40] C. T. MCMULLEN, Billiards and Teichmüller curves on Hilbert modular surfaces, J. Amer. Math. Soc. 16 (2003), 857–885.
- [41] C. T. MCMULLEN, Dynamics of $SL_2(\mathbb{R})$ over moduli space in genus two, *Ann. of Math.* **165** (2007), 397–456.
- [42] S. G. MOHANTY, Some convolutions with multinomial coefficients and related probability distributions, SIAM Rev. 8 (1966), 501–509.
- [43] A. NEVO, Pointwise ergodic theorems for radial averages on simple Lie groups. I, Duke Math. J. 76 (1994), 113–140.
- [44] D.-M. NGUYEN, Volumes of the sets of translation surfaces with small saddle connections, preprint arXiv:1211.7314.
- [45] R. RÍOS-ZERTUCHE, The pillowcase distribution and near-involutions, *Electron. J. Probab.* 19 (2014), no. 116, 22.
- [46] R. E. SCHWARTZ, Obtuse triangular billiards. II. One hundred degrees worth of periodic trajectories, *Experiment. Math.* 18 (2009), 137–171.
- [47] W. A. VEECH, Gauss measures for transformations on the space of interval exchange maps, Ann. of Math. 115 (1982), 201–242.
- [48] W. A. VEECH, Teichmüller curves in moduli space, Eisenstein series and an application to triangular billiards, *Invent. math.* 97 (1989), 553–583.
- [49] W. A. VEECH, Siegel measures, Ann. of Math. 148 (1998), 895–944.
- [50] Y. VOROBETS, Periodic geodesics on generic translation surfaces, in Algebraic and topological dynamics, Contemp. Math. 385, Amer. Math. Soc., Providence, RI, 2005, 205–258.
- [51] A. ZORICH, Square tiled surfaces and Teichmüller volumes of the moduli spaces of abelian differentials, in *Rigidity in dynamics and geometry (Cambridge, 2000)*, Springer, Berlin, 2002, 459–471.

[52] A. ZORICH, Flat surfaces, in *Frontiers in number theory, physics, and geometry. I*, Springer, Berlin, 2006, 437–583.

(Manuscrit reçu le 7 décembre 2013; accepté, après révision, le 18 janvier 2016.)

Jayadev S. ATHREYA Department of Mathematics, University of Washington, Seattle, WA 98195 USA E-mail: jathreya@uw.edu

Alex ESKIN Department of Mathematics, University of Chicago, Chicago, Illinois 60637, USA E-mail: eskin@math.uchicago.edu

Anton ZORICH Institut Universitaire de France Institut de Mathématiques de Jussieu Paris Rive Gauche, UMR CNRS 7586, Bâtiment Sophie Germain, Case 7012, 75205 PARIS Cedex 13, France E-mail: anton.zorich@imj-prg.fr

Jon CHAIKA Department of Mathematics University of Utah Salt Lake City, Utah 84112, USA E-mail: chaika@math.utah.edu